UNITED STATES SECURITIES AND EXCHANGE COMMISSION

WASHINGTON, D.C. 20549

FORM 10-Q

QUARTERLY REPORT PURSUANT TO SECTION 13 OR 15(d) OF THE SECURITIES EXCHANGE ACT OF 1934

For the quarterly period ended September 30, 2014 Commission File Number 001-31539



(Exact name of registrant as specified in its charter)

Delaware

(State or other jurisdiction of incorporation or organization)

41-0518430 (I.R.S. Employer Identification No.)

1775 Sherman Street, Suite 1200, Denver, Colorado

(Address of principal executive offices)

Large accelerated filer ☑

80203 (Zip Code)

Accelerated filer □

(303) 861-8140

(Registrant's telephone number, including area code)

Indicate by check mark whether the registrant (1) has filed all reports required to be filed by Section 13 or 15(d) of the Securities Exchange Act of 1934 during the preceding 12 months (or for such shorter period that the registrant was required to file such reports), and (2) has been subject to such filing requirements for the past 90 days. Yes \boxtimes No \square

Indicate by check mark whether the registrant has submitted electronically and posted on its corporate Web site, if any, every Interactive Data File required to be submitted and posted pursuant to Rule 405 of Regulation S-T (§232.405 of this chapter) during the preceding 12 months (or for such shorter period that the registrant was required to submit and post such files). Yes ☑ No □

Indicate by check mark whether the registrant is a large accelerated filer, an accelerated filer, a non-accelerated filer, or a smaller reporting company. See the definitions of "large accelerated filer," "accelerated filer" and "smaller reporting company" in Rule 12b-2 of the Exchange Act.

Non-accelerated filer ☐

(Do not check if a smaller reporting company)

Smaller reporting company ☐

Indicate by check mark whether the registrant is a shell company (as defined in Rule 12b-2 of the Exchange Act). Yes 🗆 No 🗹

Indicate the number of shares outstanding of each of the issuer's classes of common stock, as of the latest practicable date.

As of October 22, 2014, the registrant had 67,405,401 shares of common stock, \$0.01 par value, outstanding.

1

$\frac{\text{SM ENERGY COMPANY}}{\text{TABLE OF CONTENTS}}$

Part I.	FINANCIA	AL INFORMATION	<u>PAGE</u>
	Item 1.	Financial Statements (Unaudited)	
		Condensed Consolidated Balance Sheets September 30, 2014, and December 31, 2013	<u>3</u>
		Condensed Consolidated Statements of Operations Three and Nine Months Ended September 30, 2014, and 2013	4
		Condensed Consolidated Statements of Comprehensive Income Three and Nine Months Ended September 30, 2014, and 2013	<u>5</u>
		Condensed Consolidated Statements of Cash Flows Nine Months Ended September 30, 2014, and 2013	<u>6</u>
		Notes to Condensed Consolidated Financial Statements	<u>8</u>
	Item 2.	Management's Discussion and Analysis of Financial Condition and Results of Operations	<u>25</u>
	Item 3.	Quantitative and Qualitative Disclosures About Market Risk (included within the content of Item 2)	45
	Item 4.	Controls and Procedures	<u>45</u>
Part II.	OTHER IN	FORMATION .	
	Item 1.	Legal Proceedings	<u>46</u>
	Item 1A.	Risk Factors	<u>46</u>
	Item 2.	Unregistered Sales of Equity Securities and Use of Proceeds	<u>47</u>
	Item 6.	<u>Exhibits</u>	<u>48</u>
		2	

PART I. FINANCIAL INFORMATION

ITEM 1. FINANCIAL STATEMENTS

SM ENERGY COMPANY AND SUBSIDIARIES CONDENSED CONSOLIDATED BALANCE SHEETS (UNAUDITED) (in thousands, except share amounts)

	September 30, 2014	December 31, 2013
ASSETS		
Current assets:		
Cash and cash equivalents	\$ 269	\$ 282,248
Accounts receivable	304,175	318,371
Derivative asset	41,295	21,559
Deferred income taxes	9,084	10,749
Prepaid expenses and other	16,782	14,574
Total current assets	371,605	647,501
Property and equipment (successful efforts method):		
Proved oil and gas properties	6,789,927	5,637,462
Less - accumulated depletion, depreciation, and amortization	(3,045,006)	(2,583,698)
Unproved oil and gas properties	536,100	271,100
Wells in progress	637,584	279,654
Oil and gas properties held for sale net of accumulated depletion, depreciation and amortization of \$22,482 and \$7,390, respectively	19,883	19,072
Other property and equipment, net of accumulated depreciation of \$35,627 and \$28,775, respectively	276,953	236,202
Total property and equipment, net	5,215,441	3,859,792
Noncurrent assets:		
Derivative asset	22,510	30,951
Restricted cash	_	96,713
Other noncurrent assets	51,190	70,208
Total other noncurrent assets	73,700	197,872
Total Assets	\$ 5,660,746	\$ 4,705,165
LIABILITIES AND STOCKHOLDERS' EQUITY		
Current liabilities:		
Accounts payable and accrued expenses	\$ 664,925	\$ 606,751
Derivative liability	4,649	26,380
Other current liabilities	-,,,,,,	6,000
Total current liabilities	669,574	639,131
Noncurrent liabilities:		
Revolving credit facility		
	390,000	_
Senior Notes (note 5)	1,600,000	1,600,000
Asset retirement obligation	123,905	115,659
Asset retirement obligation associated with oil and gas properties held for sale	452	3,033
Net Profits Plan liability	41,705	56,985
Deferred income taxes	846,698	650,125
Derivative liability	8,243	4,640
Other noncurrent liabilities	28,329	28,771
Total noncurrent liabilities	3,039,332	2,459,213
Commitments and contingencies (note 6)		
Stockholders' equity:		
Common stock, \$0.01 par value - authorized: 200,000,000 shares; issued: 67,393,867 and 67,078,853 shares outstanding, respectively; net of treasury shares: 67,393,867 and 67,056,441, respectively	674	671
Additional paid-in capital	273,783	257,720
	2/3,/63	
Treasury stock, at cost: zero and 22,412 shares, respectively		(823)
Retained earnings	1,682,273	1,354,669
Accumulated other comprehensive loss	(4,890)	(5,416)
Total stockholders' equity	1,951,840	1,606,821
Total Liabilities and Stockholders' Equity	\$ 5,660,746	\$ 4,705,165

SM ENERGY COMPANY AND SUBSIDIARIES CONDENSED CONSOLIDATED STATEMENTS OF OPERATIONS (UNAUDITED) (in thousands, except per share amounts)

		For the Three Months Ended September 30,			For the Nine Months Ended September 30,			
		2014		2013		2014		2013
Operating revenues:								
Oil, gas, and NGL production revenue	\$	617,207	\$	601,787	\$	1,894,977	\$	1,605,882
Other operating revenues		1,579		11,320		31,509		50,765
Total operating revenues		618,786		613,107		1,926,486		1,656,647
Operating expenses:								
Oil, gas, and NGL production expense		178,390		158,921		519,697		434,291
Depletion, depreciation, amortization, and asset retirement obligation liability accretion		183,259		195,792		548,255		620,232
Exploration		34,556		16,280		80,161		52,335
Impairment of proved properties		_		5,935		_		61,706
Abandonment and impairment of unproved properties		15,522		3,818		18,487		8,459
General and administrative		41,696		33,920		114,862		101,574
Change in Net Profits Plan liability		(6,399)		940		(15,280)		(6,423)
Derivative (gain) loss		(190,661)		39,933		33,470		(14,685)
Other operating expenses		5,444		20,084		19,505		71,192
Total operating expenses		261,807		475,623		1,319,157		1,328,681
	'							
Income from operations		356,979		137,484		607,329		327,966
Non-operating income (expense):								
Interest expense		(22,621)		(24,488)		(70,851)		(65,170)
Other, net		(672)		28		(2,493)		64
Income before income taxes		333,686		113,024		533,985		262,860
Income tax expense		(124,748)		(42,334)		(199,660)		(98,921)
								_
Net income	\$	208,938	\$	70,690	\$	334,325	\$	163,939
	-							
Basic weighted-average common shares outstanding		67,379		66,943		67,169	_	66,486
Diluted weighted-average common shares outstanding		68,430		68,253		68,258		67,969
	-							
Basic net income per common share	\$	3.10	\$	1.06	\$	4.98	\$	2.47
Diluted net income per common share	\$	3.05	\$	1.04	\$	4.90	\$	2.41
·								
Dividends per common share	\$	0.05	\$	0.05	\$	0.10	\$	0.10

SM ENERGY COMPANY AND SUBSIDIARIES CONDENSED CONSOLIDATED STATEMENTS OF COMPREHENSIVE INCOME (UNAUDITED) (in thousands)

	For the Three Months Ended September 30,			For the Nine Months Ended Septe 30,				
		2014		2013		2014		2013
Net income	\$	208,938	\$	70,690	\$	334,325	\$	163,939
Other comprehensive income (loss), net of tax:								
Reclassification to earnings (1)		_		308		_		1,115
Pension liability adjustment		196		_		526		(3)
Total other comprehensive income, net of tax		196		308		526	· · ·	1,112
Total comprehensive income	\$	209,134	\$	70,998	\$	334,851	\$	165,051

⁽¹⁾ Reclassification from accumulated other comprehensive loss ("AOCL") related to de-designated hedges. As of December 31, 2013, all commodity derivative contracts that had been designated as cash flow hedges were settled and reclassified into earnings from AOCL.

SM ENERGY COMPANY AND SUBSIDIARIES CONDENSED CONSOLIDATED STATEMENTS OF CASH FLOWS (UNAUDITED) (in thousands)

		ns Ended September
	2014	2013
Cash flows from operating activities:		
Net income	\$ 334,325	\$ 163,939
Adjustments to reconcile net income to net cash provided by operating activities:		
Depletion, depreciation, amortization, and asset retirement obligation liability accretion	548,255	620,232
Exploratory dry hole expense	22,844	5,878
Impairment of proved properties	_	61,706
Abandonment and impairment of unproved properties	18,487	8,459
Stock-based compensation expense	24,568	25,495
Change in Net Profits Plan liability	(15,280)	(6,423)
Derivative (gain) loss	33,470	(14,685)
Derivative cash settlement gain (loss)	(62,894)	12,715
Amortization of deferred financing costs	4,433	3,914
Deferred income taxes	198,180	98,619
Plugging and abandonment	(6,193)	(7,453)
Other, net	(2,986)	3,439
Changes in current assets and liabilities:		
Accounts receivable	6,476	(45,209)
Prepaid expenses and other	234	(2,461)
Accounts payable and accrued expenses	(28,797)	72,704
Net cash provided by operating activities	1,075,122	1,000,869
Cash flows from investing activities:		
Net proceeds from sale of oil and gas properties	41,868	20,498
Capital expenditures	(1,317,862)	(1,121,355)
Acquisition of proved and unproved oil and gas properties	(459,277)	(62,007)
Other, net	(714)	(3,509)
Net cash used in investing activities	(1,735,985)	(1,166,373)
Cash flows from financing activities:		
Proceeds from credit facility	536,500	976,500
Repayment of credit facility	(146,500)	(1,288,500)
Deferred financing costs related to credit facility	_	(3,444)
Net proceeds from 2024 Notes	_	490,274
Proceeds from sale of common stock	2,898	4.450
Dividends paid	(3,353)	(3,314)
Net share settlement from issuance of stock awards	(10,576)	(16,203)
Other, net	(85)	(9)
Net cash provided by financing activities	378,884	159,754
Net change in cash and cash equivalents	(281,979)	(5,750)
Cash and cash equivalents at beginning of period	282,248	5,926
	\$ 269	\$ 176
Cash and cash equivalents at end of period	\$ 209	φ 1/0

SM ENERGY COMPANY AND SUBSIDIARIES CONDENSED CONSOLIDATED STATEMENTS OF CASH FLOWS (UNAUDITED) (Continued)

Supplemental schedule of additional cash flow information and non-cash investing and financing activities:

	Fo	For the Nine Months Ended September 30,				
		2014	2013			
		(in tho	usands)	_		
Cash paid for interest, net of capitalized interest	\$	79,119	\$	59,841		
Net cash paid (refunded) for income taxes	\$	1,979	\$	(259)		

Dividends of approximately \$3.4 million were declared by the Company's Board of Directors, but not paid, as of September 30, 2014. Dividends of approximately \$3.3 million were declared by the Company's Board of Directors, but not paid, as of September 30, 2013.

As of September 30, 2014, and 2013, \$404.8 million and \$238.7 million, respectively, of accrued capital expenditures were included in accounts payable and accrued expenses in the Company's condensed consolidated balance sheets. These oil and gas property additions are reflected in cash used in investing activities in the periods during which the payables are settled.

During the second quarter of 2014, the Company exchanged properties in its Rocky Mountain region with a fair value of \$6.2 million. During the third quarter of 2013, the Company exchanged properties in its Rocky Mountain region with a fair value of \$25.0 million. The cash consideration exchanged at the respective closings for agreed upon adjustments is reflected in the acquisition of proved and unproved oil and gas properties line item in the condensed consolidated statements of cash flows.

SM ENERGY COMPANY AND SUBSIDIARIES NOTES TO CONDENSED CONSOLIDATED FINANCIAL STATEMENTS (UNAUDITED)

Note 1 - The Company and Business

SM Energy Company ("SM Energy" or the "Company") is an independent energy company engaged in the acquisition, exploration, development, and production of crude oil and condensate, natural gas, and natural gas liquids (also respectively referred to as "oil," "gas," and "NGLs" throughout this report) in onshore North America, with a current focus on oil and liquids-rich resource plays.

Note 2 - Basis of Presentation, Significant Accounting Policies, and Recently Issued Accounting Standards

Basis of Presentation

The accompanying unaudited condensed consolidated financial statements of SM Energy have been prepared in accordance with accounting principles generally accepted in the United States ("GAAP") for interim financial information and the instructions to Form 10-Q and Regulation S-X. They do not include all information and notes required by GAAP for complete financial statements. However, except as disclosed herein, there has been no material change in the information disclosed in the notes to consolidated financial statements included in SM Energy's Annual Report on Form 10-K for the year ended December 31, 2013 (the "2013 Form 10-K"). In the opinion of management, all adjustments, consisting of normal recurring accruals considered necessary for a fair presentation of interim financial information, have been included. Operating results for the periods presented are not necessarily indicative of expected results for the full year. In connection with the preparation of its unaudited condensed consolidated financial statements, the Company evaluated events subsequent to the balance sheet date of September 30, 2014, through the filing date of this report.

Significant Accounting Policies

The significant accounting policies followed by the Company are set forth in Note 1 to the Company's consolidated financial statements in its2013 Form 10-K, and are supplemented by the notes to the unaudited condensed consolidated financial statements in this report. These unaudited condensed consolidated financial statements should be read in conjunction with the consolidated financial statements and notes included in the 2013 Form 10-K.

Recently Issued Accounting Standards

In April 2014, the Financial Accounting Standards Board ("FASB") issued new authoritative accounting guidance related to the recognition and presentation of discontinued operations in the financial statements. This guidance intends to reduce the frequency of disposals reported as discontinued operations by focusing on strategic shifts that have, or will have, a major effect on an entity's operations and financial results. This guidance is to be applied prospectively and is effective for annual periods, and interim periods within those annual periods, beginning after December 15, 2014. Early application is permitted. The Company is currently evaluating the provisions of this guidance and assessing its impact on the Company's financial statements and disclosures.

In May 2014, the FASB issued new authoritative accounting guidance related to the recognition of revenue from contracts with customers. This guidance is to be applied using a retrospective method or a modified retrospective method, as outlined in the guidance, and is effective for annual periods, and interim periods within those annual periods, beginning after December 15, 2016. Early application is not permitted. The Company is currently evaluating the provisions of this guidance and assessing its impact on the Company's financial statements and disclosures.

In June 2014, the FASB issued new authoritative accounting guidance related to the recognition of share-based compensation when an award provides that a performance target can be achieved after the requisite service period. This guidance may be applied either prospectively or retrospectively and is effective for annual periods, and interim periods within those annual periods, beginning after December 15, 2015. Early application is permitted. The Company is currently evaluating the provisions of this guidance and assessing its impact on the Company's financial statements and disclosures.

In August 2014, the FASB issued new authoritative guidance that requires management to evaluate whether there are conditions or events that raise substantial doubt about an entity's ability to continue as a going concern within one year after the date that the entity's financial statements are issued, or within one year after the date that the entity's financial statements are available to be issued, and to provide disclosures when certain criteria are met. This guidance is effective for the annual period ending after December 15, 2016, and for annual periods and interim periods thereafter. Early application is permitted. The Company is currently evaluating the provisions of this guidance and assessing its impact on the Company's financial statements and disclosures.

There are no other new significant accounting standards applicable to the Company that have been issued but not yet adopted by the Company as of September 30, 2014, and through the filing date of this report.

Note 3 - Acquisitions, Divestitures, and Assets Held for Sale

Gooseneck Prospect Acquisition

On September 24, 2014, the Company acquired approximately 61,000 net acres of proved and unproved oil and gas properties in the Company's Gooseneck prospect area in North Dakota, along with other related equipment, contracts, records and other assets (collectively the "Assets"). Total consideration paid by the Company was \$325.2 million in cash and the effective date for the acquisition was July 1, 2014.

It was determined that this acquisition met the definition of a business combination under ASC 805, *Business Combinations*, as it included proved properties. The Company allocated the preliminary adjusted purchase price to the acquired assets and liabilities based on fair values as of the acquisition date of September 24, 2014, as summarized in the following table. This acquisition is subject to normal post-closing adjustments, which are expected to be completed in early 2015. Refer to *Note 11 - Fair Value Measurements* for additional discussion on the valuation techniques used in determining the fair value of acquired properties.

Purchase Price	As of September 24, 2014 (in thousands)			
Cash consideration	\$	325,230		
Fair value of assets acquired:				
Proved oil and gas properties	\$	201,842		
Unproved oil and gas properties		125,562		
Total fair value of oil and gas properties acquired	'	327,404		
Working capital		(61)		
Asset retirement obligation		(2,113)		
Total fair value of net assets acquired	\$	325,230		

Rocky Mountain Acquisitions

In addition to the Gooseneck prospect acquisition discussed above, the Company acquired other proved and unproved properties in its Rocky Mountain region during the nine months ended September 30, 2014, from multiple sellers for approximately \$134.1 million in total cash consideration plus approximately 7,000 net acres of non-core assets in the Company's Rocky Mountain region. These acquisitions are subject to normal post-closing adjustments, which are expected to be completed in late 2014 and early 2015.

Subsequent to September 30, 2014, the Company closed an acquisition of proved and unproved properties in its Gooseneck prospect area for total cash consideration of approximately \$84.8 million. This acquisition is subject to normal post-closing adjustments, which are expected to be completed in early 2015.

Divestitures

During the second quarter of 2014, the Company divested certain non-strategic assets in the Williston Basin located in its Rocky Mountain region. Total cash proceeds received at closing (referred throughout this report as "divestiture proceeds") were \$50.2 million and the estimated net gain is \$26.8 million. This divestiture is subject to normal post-closing adjustments, which are expected to be completed during the fourth quarter of 2014.

Assets Held for Sale

Assets are classified as held for sale when the Company commits to a plan to sell the assets and there is reasonable certainty the sale will take place within year. Upon classification as held for sale, long-lived assets are no longer depreciated or depleted, and a measurement for impairment is performed to identify and expense any excess of carrying value over fair value less estimated costs to sell. Subsequent decreases to the estimated fair value less the costs to sell impact the measurement of assets held for sale.

As of September 30, 2014, the accompanying condensed consolidated balance sheets ("accompanying balance sheets") present \$19.9 million of oil and gas properties held for sale, net of accumulated depletion, depreciation, and amortization expense. A corresponding asset retirement obligation liability of \$0.5 million is separately presented. Assets held for sale are recorded at the lesser of their respective carrying value or fair value less estimated costs to sell. Certain assets classified as held for sale as of September 30, 2014, were written down to fair value less estimated costs to sell, which was recorded as a loss on divestiture activity and is included within the other operating revenues line item in the accompanying condensed consolidated statements of operations ("accompanying statements of operations").

The Company determined that these planned asset sales do not qualify for discontinued operations accounting under financial statement presentation authoritative guidance.

Note 4 - Income Taxes

Income tax expense for the three and nine months ended September 30, 2014, and 2013, differs from the amounts that would be provided by applying the statutory United States federal income tax rate to income before income taxes primarily due to the effect of state income taxes, changes in valuation allowances, percentage depletion, research and development ("R&D") credits, and other permanent differences. The quarterly rate can also be impacted by the proportional effects of forecasted net income as of each period end presented.

The provision for income taxes consists of the following:

	For	For the Three Months Ended September 30,			For	ed September			
		2014 2013		2013	2014			2013	
		(in thousands)							
Current portion of income tax (benefit) expense:									
Federal	\$	_	\$	_	\$	_	\$	_	
State		479		(46)		1,480		302	
Deferred portion of income tax expense		124,269		42,380		198,180		98,619	
Total income tax expense	\$	124,748	\$	42,334	\$	199,660	\$	98,921	
		37.4%		37.5%		37.4%		37.6%	
									

A change in the Company's effective tax rate between reported periods will generally reflect differences in its estimated highest marginal state tax rate due to changes in the composition of income from Company activities among various state tax jurisdictions. Cumulative effects of state rate changes are reflected in the period legislation is enacted.

The Company and its subsidiaries file federal income tax returns and various state income tax returns. With certain exceptions, the Company is no longer subject to United States federal or state income tax examinations by tax authorities for years before 2007. Federal tax law allowing for the calculation of an R&D credit was enacted in 2013, which allowed the credit for the 2012 and 2013 tax years. However, the Company has not yet commissioned a study to calculate the credit for these tax years. The table above excludes the impact for any credit that could be claimed for the 2013 tax year. The Internal Revenue Service ("IRS") initiated an audit in the first quarter of 2012 related to R&D tax credits claimed by the Company for the 2007 through 2010 tax years. On April 23, 2013, the IRS issued a Notice of Proposed Adjustment disallowing \$4.6 million of R&D tax credits claimed for open tax years during the audit period. During the quarter ended September 30, 2014, the Company successfully reached an agreement with the IRS Appeals Office, which allowed the Company to claim a portion of these R&D credits and resulted in no significant adjustment.

On September 13, 2013, the United States Department of the Treasury issued the final and re-proposed tangible property regulations effective for tax years beginning January 1, 2014. The Company has determined it is materially compliant with the requirements of these regulations.

Note 5 - Long-term Debt

Revolving Credit Facility

The Company's Fifth Amended and Restated Credit Agreement provides a maximum loan amount of \$2.5 billion, current aggregate lender commitments of \$1.3 billion, and a maturity date of April 12, 2018. The borrowing base is subject to regular semi-annual redeterminations. On October 6, 2014, the lending group redetermined the Company's borrowing base under the credit facility and increased it from \$2.2 billion to \$2.4 billion, with no change in the aggregate lender commitments of \$1.3 billion. The borrowing base redetermination process under the credit facility considers the value of the Company's proved oil and gas properties, as determined by the lender group. The next scheduled redetermination date is April 1, 2015. Borrowings under the facility are secured by at least 75 percent of the Company's proved oil and gas properties.

The Company must comply with certain financial and non-financial covenants under the terms of its credit facility agreement, including limitations on the payment of dividends to \$50.0 million per year. The Company was in compliance with all covenants under the credit facility as of September 30, 2014, and through the filing date of this report.

The following table presents the outstanding balance, total amount of letters of credit, and available borrowing capacity under the Company's credit facility as of October 22, 2014, September 30, 2014, and December 31, 2013:

	As	As of October 22, 2014 As of September 30, 2014			As of December 31, 2013		
				(in thousands)			
Credit facility balance	\$	558,500	\$	390,000	\$	_	
Letters of credit (1)	\$	808	\$	808	\$	808	
Available borrowing capacity	\$	740,692	\$	909,192	\$	1,299,192	

⁽¹⁾ Letters of credit reduce the available borrowing capacity under the credit facility on a dollar-for-dollar basis.

Senior Notes

The Senior Notes line on the accompanying balance sheets represents the outstanding principal amount of the 6.625% Senior Notes due 2019 (the "2019 Notes"), the 6.50% Senior Notes due 2021 (the "2021 Notes"), the 6.50% Senior Notes due 2023 (the "2023 Notes"), and the 5.0% Senior Notes due 2024 (the "2024 Notes" and collectively with the 2019 Notes, 2021 Notes, and 2023 Notes, the "Senior Notes"), as shown in the table below:

	As of September 30, 2014		As	of December 31, 2013		
		(in thousands)				
2019 Notes	\$	350,000	\$	350,000		
2021 Notes		350,000		350,000		
2023 Notes		400,000		400,000		
2024 Notes		500,000		500,000		
Total Senior Notes	\$	1,600,000	\$	1,600,000		

The Senior Notes are unsecured senior obligations and rank equal in right of payment with all of the Company's existing and any future unsecured senior debt, and are senior in right of payment to any future subordinated debt. There are no subsidiary guarantors of the Senior Notes. The Company is subject to certain covenants under the respective indentures governing the Senior Notes that limit the Company's ability to incur additional indebtedness, issue preferred stock, and make restricted payments, including dividends; provided, however, that the first \$6.5 million of dividends paid each year are not restricted by these covenants. The Company does not expect these restrictions to limit its ability to continue paying dividends at its current rate for the foreseeable future if declared by the Company's Board of Directors. The Company was in compliance with all covenants under its Senior Notes as of September 30, 2014, and through the filing date of this report.

2024 Notes

On May 20, 2013, the Company issued \$500.0 million in aggregate principal amount of 2024 Notes. The 2024 Notes were issued at par and mature on January 15, 2024. Please refer to *Note 5 - Long-term Debt* in the Company's 2013 Form 10-K for additional discussion of the terms of these notes.

On May 20, 2013, the Company entered into a registration rights agreement that provided holders of the 2024 Notes certain registration rights under the Securities Act of 1933, as amended (the "Securities Act"). The Company closed its offer to exchange its 2024 Notes for notes registered under the Securities Act on June 25, 2014.

Note 6 - Commitments and Contingencies

Commitments

During the first nine months of 2014, the Company entered into drilling rig contracts with total expected commitments of \$110.5 million and varying terms extending through 2016. As of September 30, 2014, the remaining expected drilling commitments totaled \$72.3 million. Early termination of the remaining contracts would result in penalties totaling \$42.5 million as of September 30, 2014. Subsequent to September 30, 2014, the Company entered into additional drilling rig contracts with future expected commitments totaling \$24.5 million, and varying terms extending through 2016. Early termination of these contracts would result in penalties totaling \$14.4 million.

During the third quarter of 2014, as part of the Gooseneck prospect acquisition described in *Note 3 - Acquisitions, Divestitures, and Assets Held for Sale* the Company assumed a share of the seller's rights and obligations under a gas purchase agreement whereby the Company is subject to certain through-put commitments extending through 2028. The Company may be required to make periodic deficiency payments for any shortfalls in delivering the minimum applicable annual volume commitment. In the event that no product is delivered in accordance with this agreement, the undiscounted aggregate deficiency payments totaled approximately \$24.4 million as of September 30, 2014. Subsequent to September 30, 2014, the Company closed an acquisition resulting in an increase it its obligations under this gas purchase agreement. As of the filing date of this report, in the event that no product is delivered in accordance with this agreement, the undiscounted aggregate deficiency payments total approximately \$42.6 million. The Company expects to deliver at least the minimum applicable annual volumes under this agreement.

Contingencies

The Company is subject to litigation and claims arising in the ordinary course of business. The Company accrues for such items when a liability is both probable and the amount can be reasonably estimated. In the opinion of management, the results of such pending litigation and claims will not have a material effect on the results of operations, the financial position, or the cash flows of the Company.

On April 16, 2014, the Company settled its previously disclosed litigation against Endeavour Operating Corporation ("Endeavour"). The Company, its working interest partners, and Endeavour agreed to mutually release all claims and dismiss the lawsuit in exchange for certain cash payments and other consideration paid to the Company and its working interest partners by Endeavour. The Company recorded a \$10.7 million gain in the other operating revenues line item in the accompanying statements of operations in the second quarter of 2014 relating to this settlement.

On January 27, 2011, Chieftain Royalty Company ("Chieftain") filed a Class Action Petition against the Company in the District Court of Beaver County, Oklahoma, claiming damages related to royalty payments on all of the Oklahoma oil and gas wells operated by the Company and its predecessors. These claims include breach of contract, breach of fiduciary duty, fraud, unjust enrichment, tortious breach of contract, conspiracy, and conversion, based generally on asserted improper deduction of post-production costs. The Company removed this lawsuit to the United States District Court for the Western District of Oklahoma on February 22, 2011. The Company responded to the petition and denied the allegations. The district court did not rule on Chieftain's motion to certify the putative class, and stayed all proceedings until the United States Court of Appeals for the Tenth Circuit issued its rulings on class certification in two similar royalty class action lawsuits. On July 9, 2013, the Tenth Circuit issued its opinions, reversing the trial courts' grant of class certification and remanding the matters to the trial courts for those cases. The district court presiding over the Company's case subsequently lifted its stay, and the Company now expects Chieftain to file a new motion for class certification in the first half of 2016.

This case involves complex legal issues and uncertainties; a potentially large class of plaintiffs, and a large number of related producing properties, lease agreements and wells; and an alleged class period commencing in 1988 and spanning the entire producing life of the wells. Because the proceedings are in the early stages, with discovery yet to be completed, the Company is unable to estimate what impact, if any, the action will have on its financial condition, results of operations, or cash flows. The Company is still evaluating the claims, but believes that it has properly paid royalties under Oklahoma law and has and will continue to vigorously defend this case. On December 30, 2013, the Company sold a substantial portion of the assets that were subject to this matter and the buyer assumed any such liabilities related to such properties.

Note 7 - Compensation Plans

Cash Bonus Plan

During the first nine months of 2014 and 2013, the Company paid \$41.7 million and \$16.0 million, respectively, for cash bonuses earned during the 2013 and 2012 performance years, respectively. The general and administrative ("G&A") expense and exploration expense line items in the accompanying statements of operations include \$6.2 million and \$5.8 million of accrued cash bonus plan expense for the three months endedSeptember 30, 2014, and 2013, respectively, and \$19.0 million and \$16.7 million of accrued cash bonus plan expense for the nine months ended September 30, 2014, and 2013, respectively, related to the respective performance years.

Non-qualified Deferred Compensation Plan

In January 2014, the Company established a non-qualified deferred compensation ("NQDC") plan intended to provide plan participants with the ability to plan for income tax events and the opportunity to receive a benefit for matching contributions in excess of Internal Revenue Code ("IRC") limits applicable to the Company's 401(k) plan. The NQDC plan is designed to allow employee participants to defer a portion of base salary and cash bonuses paid pursuant to the Company's cash bonus plan and director participants to defer a portion of the cash retainer paid to directors. Each year, participating employees may elect to defer (i) between 0% and 50% of their base salary and (ii) between 0% and 100% of the cash bonus paid pursuant to the cash bonus plan, and participating directors may elect to defer between0% and 100% of their cash retainer. The NQDC plan requires the Company to make contributions for each eligible employee equal to 100% of the deferred amount for such employee, limited to 6% of such employee's base salary and cash bonus. Each eligible employee's interest in contributions made by the Company will vest 40% after the second year of such employee's service to the Company, and 20% per year thereafter. A participant's account will be distributed based upon the participant's payment election made at the time of deferral. A participant may elect to have distributions made in lump sum or in

annual installments ranging for a period from 1 to 10 years. Participants in the NQDC plan are currently limited to the Company's officers and directors.

Restricted Stock Units Under the Equity Incentive Compensation Plan

The Company grants restricted stock units ("RSUs") as part of its equity compensation program. Each RSU represents a right to one share of the Company's common stock to be delivered upon settlement of the award at the end of the specified vesting period. Expense associated with RSUs is recognized as G&A expense and exploration expense over the vesting period of the award.

Total expense recorded for RSUs for the three months endedSeptember 30, 2014, and 2013, was \$4.8 million and \$3.7 million, respectively, and \$10.5 million and \$10.0 million for the nine months endedSeptember 30, 2014, and 2013, respectively. As of September 30, 2014, there was \$26.7 million of total unrecognized compensation expense related to unvested RSU awards, which is being amortized through 2017.

A summary of the status and activity of non-vested RSUs for the nine-month period endedSeptember 30, 2014, is presented in the following table:

	RSUs	V	Veighted-Average Grant-Date Fair Value
Non-vested at beginning of year	580,431	\$	57.05
Granted	233,601	\$	84.11
Vested	(250,601)	\$	58.09
Forfeited	(32,736)	\$	59.07
Non-vested at end of quarter	530,695	\$	68.34

The fair value of the RSUs granted during the first nine months of 2014 was \$19.6 million. These RSUs will vest $1/3^{rd}$ on each of the next three anniversary dates of the grant. During the first nine months of 2014, the Company settled 250,601 RSUs that related to awards granted in previous years. The Company and the majority of grant participants mutually agreed to net share settle the awards to cover income and payroll tax withholdings as provided for in the plan document and award agreements. As a result, the Company issued 169,835 net shares of common stock. The remaining 80,766 shares were withheld to satisfy income and payroll tax withholding obligations that occurred upon delivery of the shares underlying those RSUs.

Performance Share Units Under the Equity Incentive Compensation Plan

The Company grants performance share units ("PSUs") as part of its equity compensation program. The number of shares of the Company's common stock issued to settle PSUs ranges from 0% to 200% of the number of PSUs awarded and is determined based on the Company's performance over a three-year measurement period. The performance criteria for the PSUs are based on a combination of the Company's annualized total shareholder return ("TSR") for the measurement period and the relative performance of the Company's TSR compared with the annualized TSRs of a group of peer companies for the measurement period. Expense associated with PSUs is recognized as G&A expense and exploration expense over the vesting period of the award.

Total expense recorded for PSUs for the three months endedSeptember 30, 2014, and 2013, was \$4.8 million and \$3.5 million, respectively, and \$11.6 million and \$13.2 million for the nine months endedSeptember 30, 2014, and 2013, respectively. As of September 30, 2014, there was \$24.7 million of total unrecognized compensation expense related to unvested PSU awards, which is being amortized through 2017.

A summary of the status and activity of non-vested PSUs for thenine-month period ended September 30, 2014, is presented in the following table:

	PSUs (1)	Weighted-Average Grant-Date Fair Value
Non-vested at beginning of year	572,469	\$ 66.07
Granted	202,404	\$ 94.66
Vested	(115,784)	\$ 59.30
Forfeited	(129,491)	\$ 87.21
Non-vested at end of quarter	529,598	\$ 73.31

⁽¹⁾ The number of awards assumes a one multiplier. The final number of shares of common stock issued may vary depending on the three-year performance multiplier, which ranges from zero to two.

The fair value of the PSUs granted during the firstnine months of 2014 was \$19.2 million. These PSUs will vest on the third anniversary of the date of the grant. During the first nine months of 2014, the Company settled PSUs that were granted in 2011, which earned a0.55 times multiplier, by issuing a net 85,121 shares of the Company's common stock in accordance with the terms of the PSU awards. The Company and the majority of grant participants mutually agreed to net share settle the awards to cover income and payroll tax withholdings as provided for in the plan document and award agreements. As a result, the Company withheld 45,042 shares to satisfy income and payroll tax withholding obligations that occurred upon delivery of the shares underlying those PSUs.

Stock Option Grants Under the Equity Incentive Compensation Plan

A summary of activity associated with the Company's Stock Option Plan for thenine months ended September 30, 2014, is presented in the following table:

	Shares	Weighted- Average Exercise Price	Aggregate Intrinsic Value (in thousands)
Outstanding, at beginning of year	39,088	\$ 20.87	\$ 2,433
Exercised	(19,544)	\$ 20.87	\$ 1,237
Forfeited	_	\$ _	\$ _
Outstanding, at end of quarter	19,544	\$ 20.87	\$ 1,117
Vested and exercisable, at end of quarter	19,544	\$ 20.87	\$ 1,117

As of September 30, 2014, there was no unrecognized compensation expense related to stock option awards.

Director Shares

During the nine months ended September 30, 2014 and 2013, the Company issued 27,677 and 28,169 shares, respectively, of its common stock to its non-employee directors, under the Company's Equity Incentive Compensation Plan. The Company recorded \$196,000 of compensation expense related to these awards for the three months ended September 30, 2014, and no compensation expense related to these awards for the three months ended September 30, 2013. The Company recorded \$1.4 million and \$1.4 million of compensation expense related to these awards for thenine months ended September 30, 2014, and 2013, respectively.

All shares of common stock issued to the Company's non-employee directors are earned over the one-year service period following the date of grant, unless five years of service has been provided by the director, in which case that director's shares vest immediately.

Under the Company's Employee Stock Purchase Plan ("ESPP"), eligible employees may purchase shares of the Company's common stock through payroll deductions of up to 15 percent of eligible compensation, without accruing in excess of \$25,000 in value from purchases for each calendar year. The purchase price of the stock is 85 percent of the lower of the fair market value of the stock on the first or last day of the purchase period, and shares issued under the ESPP have no restriction period. The ESPP is intended to qualify under Section 423 of the IRC. The Company had 1.2 million shares available for issuance under the ESPP as of September 30, 2014. The Company issued 35,249 and 44,437 shares under the ESPP during the first nine months of 2014 and 2013, respectively. The fair value of ESPP grants is measured at the date of grant using the Black-Scholes option-pricing model.

Net Profits Interest Bonus Plan

Cash payments made or accrued under the Company's Net Profits Interest Bonus Plan ("Net Profits Plan") that have been recorded as either G&A expense or exploration expense are presented in the table below:

	For the Three Months Ended September 30,					he Nine Months	Ende	ed September 30,
	2014			2013		2014	2013	
				(in thou	ısands)			
General and administrative expense	\$	2,487	\$	4,302	\$	7,451	\$	11,531
Exploration expense		162		329		644		1,026
Total	\$	2,649	\$	4,631	\$	8,095	\$	12,557

Additionally, the Company accrued or made cash payments under the Net Profits Plan of \$8.3 million and \$2.6 million for the nine months ended September 30, 2014, and 2013, respectively, as a result of divestiture proceeds. These cash payments are accounted for as a reduction in gain on divestiture activity included within the other operating revenues line in the accompanying statements of operations.

Note 8 - Pension Benefits

Pension Plans

The Company has a non-contributory pension plan covering substantially all employees who meet age and service requirements (the "Qualified Pension Plan"). The Company also has a supplemental non-contributory pension plan covering certain management employees (the "Nonqualified Pension Plan" and together with the Qualified Pension Plan, the "Pension Plans").

Components of Net Periodic Benefit Cost for the Pension Plans

The following table presents the components of the net periodic benefit cost for the Pension Plans:

	For the Three Months Ended September 30,				Fo	or the Nine Month 3	s Ended September		
	2014			2013		2014	2013		
				(in tho	usan	ds)			
Service cost	\$	1,584	\$	1,572	\$	4,752	\$ 4,718		
Interest cost		548		407		1,643	1,220		
Expected return on plan assets that reduces periodic pension costs		(494)		(384)		(1,483)	(1,153)		
Amortization of prior service costs		4		4		13	13		
Amortization of net actuarial loss		172		306		516	917		
Net periodic benefit cost	\$	1,814	\$	1,905	\$	5,441	\$ 5,715		

Prior service costs are amortized on a straight-line basis over the average remaining service period of active participants. Gains and losses in excess of 10 percent of the greater of the benefit obligation and the market-related value of assets are amortized over the average remaining service period of active participants.

The Company contributed \$5.3 million to the Pension Plans during the nine month period ended September 30, 2014.

Note 9 - Earnings per Share

Basic net income per common share is calculated by dividing net income available to common stockholders by the basic weighted-average common shares outstanding for the respective period. The Company's earnings per share calculations reflect the impact of any repurchases of shares of common stock made by the Company.

Diluted net income per common share is calculated by dividing adjusted net income by the diluted weighted-average common shares outstanding, which includes the effect of potentially dilutive securities. Potentially dilutive securities for this calculation consist of in-the-money outstanding stock options, unvested RSUs, and contingent PSUs. The treasury stock method is used to measure the dilutive impact of unvested RSUs, contingent PSUs, and in-the-money stock options.

PSUs represent the right to receive, upon settlement of the PSUs after completion of thethree-year performance period, a number of shares of the Company's common stock that may range from 0% to 200% of the number of PSUs granted on the award date. The number of potentially dilutive shares related to PSUs is based on the number of shares, if any, that would be issuable at the end of the respective reporting period, assuming that date was the end of the contingency period applicable to such PSUs. For additional discussion on PSUs, please refer to Note 7 - Compensation Plans under the heading Performance Share Units Under the Equity Incentive Compensation Plan.

The following table sets forth the calculations of basic and diluted earnings per share:

	For the Three Months Ended September 30,				For tl	he Nine Months	Ende	d September 30,
		2014		2013	2014			2013
			(in tl	ousands, excep	per sh	are amounts)		
Net income	\$	208,938	\$	70,690	\$	334,325	\$	163,939
Basic weighted-average common shares outstanding		67,379		66,943		67,169		66,486
Add: dilutive effect of stock options, unvested RSUs, and contingent PSUs		1,051		1,310		1,089		1,483
Diluted weighted-average common shares outstanding		68,430		68,253		68,258		67,969
Basic net income per common share	\$	3.10	\$	1.06	\$	4.98	\$	2.47
Diluted net income per common share	\$	3.05	\$	1.04	\$	4.90	\$	2.41

Note 10 - Derivative Financial Instruments

Summary of Oil, Gas, and NGL Derivative Contracts in Place

The Company has entered into various commodity derivative contracts to mitigate a portion of its exposure to potentially adverse market changes in commodity prices and the associated impact on cash flows. All contracts are entered into for other-than-trading purposes. The Company's derivative contracts include swap and costless collar arrangements for oil, gas, and NGLs.

As of September 30, 2014, the Company had commodity derivative contracts outstanding through the second quarter of 2018 for a total of 17.6 million Bbls of oil production, 164.1 million MMBtu of gas production, and 1.8 million Bbls of NGL production.

In a typical commodity swap agreement, if the agreed upon published third-party index price ("index price") is lower than the swap fixed price, the Company receives the difference between the index price and the agreed upon swap fixed price. If the index price is higher than the swap fixed price, the Company pays the difference. For collar agreements, the Company receives the difference between an index price and the floor price if the index price is below the floor price. The Company pays the difference between the ceiling price and the index price if the index price is above the ceiling price. No amounts are paid or received if the index price is between the floor and ceiling prices.

The following tables summarize the approximate volumes and average contract prices of contracts the Company had in place as of September 30, 2014:

Oil Contracts

Oil Swaps

Contract Period	NYMEX WTI Volumes	Weighted-Average Contract Price			
	(Bbls)		(per Bbl)		
Fourth quarter 2014	2,016,000	\$	96.01		
2015	5,741,000	\$	91.14		
2016	5,570,000	\$	88.01		
All oil swaps	13,327,000				

Oil Collars

Contract Period	NYMEX WTI Volumes		Weighted- Average Floor Price	Weighted- Average Ceiling Price
	(Bbls)		(per Bbl)	(per Bbl)
Fourth quarter 2014	923,000	\$	85.00	\$ 102.63
2015	3,366,000	\$	85.00	\$ 94.25
All oil collars	4,289,000			

Gas Contracts

Gas Swaps

Contract Period	Volumes	Weighted-Average Contract Price			
	(MMBtu)		(per MMBtu)		
Fourth quarter 2014	22,014,000	\$	4.02		
2015	57,943,000	\$	4.04		
2016	37,472,000	\$	4.17		
2017	23,430,000	\$	4.21		
2018	10,200,000	\$	4.31		
All gas swaps*	151,059,000				

^{*}Gas swaps are comprised of IF El Paso Permian (4%), IF HSC (83%), IF NGPL TXOK (2%), IF NNG Ventura (2%), IF Enable East (8%), and IF CIG (1%).

Gas Collars

Contract Period	Volumes	Weighted- Average Floor Price	loor Average Ceil	
	(MMBtu)	(per MMBtu)		(per MMBtu)
2015	13,002,000	\$ 3.98	\$	4.30
All gas collars*	13,002,000			

^{*}Gas collars are comprised of IF El Paso Permian (4%), IF HSC (80%), IF NNG Ventura (8%), and IF Enable East (8%).

NGL Swaps

Contract Period	Volumes	'	Weighted-Average Contract Price
	(Bbls)		(per Bbl)
Fourth quarter 2014	980,000	\$	61.69
2015	781,000	\$	55.42
All NGL swaps*	1,761,000		

^{*}NGL swaps are comprised of Oil Price Information System ("OPIS") Mont Belvieu LDH Propane (67%) and OPIS Mont Belvieu NON-LDH Natural Gasoline (63%).

Derivative Assets and Liabilities Fair Value

The Company's commodity derivatives are measured at fair value and are included in the accompanying balance sheets as derivative assets and liabilities. The fair value of the commodity derivative contracts was a net asset of \$50.9 million and net asset of \$21.5 million as of September 30, 2014, and December 31, 2013, respectively.

The following tables detail the fair value of derivatives recorded in the accompanying balance sheets, by category:

As of September 30, 2014	
	Derivat

	115 01 50 ptermoet 50, 201.						
	Derivative Assets			Derivative Lia	abilities		
	Balance Sheet Classification	Fa	ir Value	Balance Sheet Classification	Fa	ir Value	
			(in thou	isands)			
Commodity contracts	Current assets	\$	41,295	Current liabilities	\$	4,649	
Commodity contracts	Noncurrent assets		22,510	Noncurrent liabilities		8,243	
Derivatives not designated as hedging instruments		\$	63,805		\$	12,892	

As of December 31, 2013

	Derivative	Derivative Assets			Derivative Liabilities				
	Balance Sheet Classification	Fair Value		Balance Sheet Classification	F	air Value			
			(in tho	isands)					
Commodity contracts	Current assets	\$	21,559	Current liabilities	\$	26,380			
Commodity contracts	Noncurrent assets		30,951	Noncurrent liabilities		4,640			
Derivatives not designated as hedging instruments		\$	52,510		\$	31,020			

Offsetting of Derivative Assets and Liabilities

As of September 30, 2014, and December 31, 2013, all derivative instruments held by the Company were subject to enforceable master netting arrangements by various financial institutions. In general, the terms of the Company's agreements provide for offsetting of amounts payable or receivable between it and the counterparty, at the election of both parties, for settlements that occur on the same date and in the same currency. The Company's agreements also provide that in the event of an early termination, the counterparties have the right to offset amounts owed or owing under that and any other agreement with the same counterparty. The Company's accounting policy is to not offset these positions in its accompanying balance sheets.

The following table provides a reconciliation between the gross assets and liabilities reflected on the accompanying balance sheets and the potential effects of master netting arrangements on the fair value of the Company's derivative contracts:

	Derivative Assets					Derivative Liabilities			
		A	s of			As	of		
Offsetting of Derivative Assets and Liabilities		eptember 30, 2014	Decei	mber 31, 2013	Se	ptember 30, 2014	Dece	mber 31, 2013	
		(in thousands)							
Gross amounts presented in the accompanying balance sheets	\$	63,805	\$	52,510	\$	(12,892)	\$	(31,020)	
Amounts not offset in the accompanying balance sheets		(12,744)		(30,652)		12,744		30,652	
Net amounts	\$	51,061	\$	21,858	\$	(148)	\$	(368)	

The following table summarizes the components of the derivative (gain) loss presented in the accompanying statements of operations:

	For the Three Months Ended September 30,				For the Nine Months Ended Septemb			
		2014		2013		2014		2013
				(in tho	usands)			
Derivative cash settlement (gain) loss:								
Oil contracts	\$	517	\$	13,538		27,435	\$	13,786
Gas contracts		1,687		(11,019)		28,563		(18,752)
NGL contracts		(1,930)		(1,231)		6,896		(7,749)
Total derivative cash settlement (gain) loss ⁽¹⁾		274		1,288		62,894		(12,715)
Derivative (gain) loss:								
Oil contracts		(141,429)		30,488		(42,802)		8,233
Gas contracts		(43,039)		(2,264)		17,700		(12,462)
NGL contracts		(6,467)		10,421		(4,322)		2,259
Total derivative (gain) loss (2)	\$	(190,661)	\$	39,933	\$	33,470	\$	(14,685)

⁽¹⁾ Total derivative cash settlement (gain) loss is reported in the derivative cash settlement gain (loss) line item on the condensed consolidated statements of cash flows within net cash provided by operating activities.

Credit Related Contingent Features

As of September 30, 2014, and through the filing date of this report, all of the Company's derivative counterparties were members of the Company's credit facility lender group. The Company's obligations under its derivative contracts are secured by liens on at least 75 percent of the Company's proved oil and gas properties.

⁽²⁾ Total derivative (gain) loss is reported in the derivative (gain) loss line item on the condensed consolidated statements of cash flows within net cash provided by operating activities.

Note 11 - Fair Value Measurements

The Company follows fair value measurement authoritative accounting guidance for all assets and liabilities measured at fair value. That authoritative accounting guidance defines fair value as the price that would be received to sell an asset or paid to transfer a liability (an exit price) in an orderly transaction between market participants at the measurement date. Market or observable inputs are the preferred sources of values, followed by assumptions based on hypothetical transactions in the absence of market inputs. The fair value hierarchy for grouping these assets and liabilities is based on the significance level of the following inputs:

- Level 1 quoted prices in active markets for identical assets or liabilities
- Level 2 quoted prices in active markets for similar assets or liabilities, quoted prices for identical or similar instruments in markets that are not active, and model-derived valuations whose inputs are observable or whose significant value drivers are observable
- Level 3 significant inputs to the valuation model are unobservable

The following is a listing of the Company's assets and liabilities that are measured at fair value and their classification within the fair value hierarchy as of September 30, 2014:

		Level 1	Level 2	Level 3
	<u> </u>		(in thousands)	
Assets:				
Derivatives (1)	\$	_	\$ 63,805	\$ _
Proved oil and gas properties (2)	\$	_	\$ _	\$ 9,443
Unproved oil and gas properties (2)	\$	_	\$ _	\$ 43,252
Oil and gas properties held for sale ⁽²⁾	\$	_	\$ _	\$ 17,952
<u>Liabilities:</u>				
Derivatives (1)	\$	_	\$ 12,892	\$ _
Net Profits Plan (1)	\$	_	\$ _	\$ 41,705
Asset retirement obligation (2)	\$	_	\$ _	\$ 923
Asset retirement obligation associated with oil and gas properties held for sale(2)	\$	_	\$ _	\$ 452

⁽¹⁾ This represents a financial asset or liability that is measured at fair value on a recurring basis.

The following is a listing of the Company's assets and liabilities that are measured at fair value and their classification within the fair value hierarchy as of December 31, 2013:

	<u></u>	Level 1		Level 2	Level 3	
				(in thousands)		
Assets:						
Derivatives (1)	\$	_	\$	52,510	\$	_
Proved oil and gas properties (2)	\$	_	\$	_	\$	62,178
Unproved oil and gas properties (2)	\$	_	\$	_	\$	3,280
Oil and gas properties held for sale(2)	\$	_	\$	_	\$	650
<u>Liabilities:</u>						
Derivatives (1)	\$	_	\$	31,020	\$	_
Net Profits Plan (1)	\$	_	\$	_	\$	56,985

⁽¹⁾ This represents a financial asset or liability that is measured at fair value on a recurring basis.

⁽²⁾ This represents a non-financial asset or liability that is measured at fair value on a nonrecurring basis.

⁽²⁾ This represents a non-financial asset or liability that is measured at fair value on a nonrecurring basis.

Both financial and non-financial assets and liabilities are categorized within the above fair value hierarchy based on the lowest level of input that is significant to the fair value measurement. The following is a description of the valuation methodologies used by the Company as well as the general classification of such instruments pursuant to the above fair value hierarchy.

Derivatives

The Company uses Level 2 inputs to measure the fair value of oil, gas, and NGL commodity derivatives. Fair values are based upon interpolated data. The Company derives internal valuation estimates taking into consideration the counterparties' credit ratings, the Company's credit rating, and the time value of money. These valuations are then compared to the respective counterparties' mark-to-market statements. These factors result in an estimated exit-price that management believes provides a reasonable and consistent methodology for valuing derivative instruments. The derivative instruments utilized by the Company are not considered by management to be complex, structured, or illiquid. The oil, gas, and NGL commodity derivative markets are highly active.

Generally, market quotes assume that all counterparties have near zero, or low, default rates and have equal credit quality. However, an adjustment may be necessary to reflect the credit quality of a specific counterparty to determine the fair value of the instrument. The Company monitors the credit ratings of its counterparties and may require counterparties to post collateral if their ratings deteriorate. In some instances, the Company will attempt to novate the trade to a more stable counterparty.

Valuation adjustments are necessary to reflect the effect of the Company's credit quality on the fair value of any liability position with a counterparty. This adjustment takes into account any credit enhancements, such as collateral margin that the Company may have posted with a counterparty, as well as any letters of credit between the parties. The methodology to determine this adjustment is consistent with how the Company evaluates counterparty credit risk, taking into account the Company's credit rating, current credit facility margins, and any change in such margins since the last measurement date. All of the Company's derivative counterparties are members of the Company's credit facility lender group.

The methods described above may result in a fair value estimate that may not be indicative of net realizable value or may not be reflective of future fair values and cash flows. While the Company believes that the valuation methods utilized are appropriate and consistent with authoritative accounting guidance and with other marketplace participants, the Company recognizes that third parties may use different methodologies or assumptions to determine the fair value of certain financial instruments that could result in a different estimate of fair value at the reporting date.

Refer to Note 10 - Derivative Financial Instruments for more information regarding the Company's derivative instruments.

Net Profits Plan

The Net Profits Plan is a standalone liability for which there is no available market price, principal market, or market participants. The inputs available for this instrument are unobservable and are therefore classified as Level 3 inputs. The Company employs the income approach, which converts expected future cash flow amounts to a single present value amount. This technique uses the estimate of future cash payments, expectations of possible variations in the amount and/or timing of cash flows, the risk premium, and nonperformance risk to calculate the fair value. There is a direct correlation between realized oil, gas, and NGL commodity prices driving net cash flows and the Net Profits Plan liability. Generally, higher commodity prices result in a larger Net Profits Plan liability and lower commodity prices result in a smaller Net Profits Plan liability.

The Company records the estimated fair value of the long-term liability for estimated future payments under the Net Profits Plan based on the discounted value of estimated future payments associated with each individual pool. The calculation of this liability is a significant management estimate. A discount rate of 12 percent is used to calculate this liability and is intended to represent the Company's best estimate of the present value of expected future payments under the Net Profits Plan.

The Company's estimate of its liability is highly dependent on commodity prices, cost assumptions, discount rates, and overall market conditions. The Company regularly assesses the current market environment. The Net Profits Plan liability is determined using price assumptions of five one-year strip prices with the fifth year's pricing then carried out indefinitely. The average price is adjusted for realized price differentials and to include the effects of the forecasted production covered by derivatives contracts in the relevant periods. The non-cash expense associated with this significant management estimate is highly volatile from period to period due to fluctuations that occur in the oil, gas, and NGL commodity markets.

If the commodity prices used in the calculation changed by five percent, the liability recorded at September 30, 2014, would differ by approximately \$4 million. A one percent increase or decrease in the discount rate would result in a change of approximately \$2 million. Actual cash payments to be made to participants in future periods are dependent on realized actual production, realized commodity prices, and costs associated with the properties in each individual pool of the Net Profits Plan. Consequently, actual cash payments are inherently different from the amounts estimated.

No published market quotes exist on which to base the Company's estimate of fair value of its Net Profits Plan liability. As such, the recorded fair value is based entirely on management estimates that are described within this footnote. While some inputs to the Company's calculation of fair value on the Net Profits Plan's future payments are from published sources, others, such as the discount rate and the expected future cash flows, are derived from the Company's own calculations and estimates.

The following table reflects the activity for the Company's Net Profits Plan liability measured at fair value using Level 3 inputs:

		ne Months Ended ober 30, 2014
	(in t	housands)
Beginning balance	\$	56,985
Net increase in liability (1)		1,156
Net settlements (1) (2)		(16,436)
Transfers in (out) of Level 3		_
Ending balance	\$	41,705

⁽¹⁾ Net changes in the Company's Net Profits Plan liability are shown in the Change in Net Profits Plan liability line item of the accompanying statements of operations.

Long-term Debt

The following table reflects the fair value of the Senior Notes measured using Level 1 inputs based on quoted secondary market trading prices. The Senior Notes were not presented at fair value on the accompanying balance sheets as of September 30, 2014, or December 31, 2013, as they are recorded at historical value.

	As of Sept	ember 30, 2014	As of December 31, 2013
		(in thousa	ands)
2019 Notes	\$	364,000 \$	374,290
2021 Notes	\$	371,875 \$	373,625
2023 Notes	\$	415,500 \$	422,000
2024 Notes	\$	492,500 \$	475,315

As of September 30, 2014, the Company had \$390.0 million of floating-rate debt outstanding. The carrying value of the Company's credit facility approximates its fair value, as the applicable interest rates are floating and based on prevailing market rates.

⁽²⁾ Settlements represent cash payments made or accrued under the Net Profits Plan. The Company accrued or made cash payments under the Net Profits Plan of \$8.3 million as a result of divestitures during the nine months ended September 30, 2014.

Proved and Unproved Oil and Gas Properties

Proved oil and gas property costs are evaluated for impairment and reduced to fair value when there is an indication that the carrying costs may not be recoverable. The Company uses Level 3 inputs and the income valuation technique, which converts future amounts to a single present value amount, to measure the fair value of proved properties through an application of discount rates and price forecasts selected by the Company's management. The calculation of the discount rate is based on the best information available and was estimated to be 12 percent as of September 30, 2014, and December 31, 2013. The Company believes that the discount rate is representative of current market conditions and takes into account estimates of future cash payments, expectations of possible variations in the amount and/or timing of cash flows, the risk premium, and nonperformance risk. The prices for oil and gas are forecasted based on New York Mercantile Exchange ("NYMEX") strip pricing, adjusted for basis differentials, for the first five years, after which a flat terminal price is used for each commodity stream. The prices for NGLs are forecasted using OPIS Mont Belvieu pricing, for as long as the market is actively trading, after which a flat terminal price is used. Future operating costs are also adjusted as deemed appropriate for these estimates. Proved properties classified as held for sale are valued using a market approach, based on an estimated selling price, as evidenced by the most current bid prices received from third parties. If an estimated selling price is not available, the Company utilizes the income valuation technique discussed above.

Unproved oil and gas property costs are evaluated for impairment and reduced to fair value when there is an indication that the carrying costs may not be recoverable. To measure the fair value of unproved properties, the Company uses a market approach, which takes into account the following significant assumptions: future development plans, risk weighted potential resource recovery, and estimated reserve values. Unproved properties classified as held for sale are valued using a market approach, based on an estimated selling price, as evidenced by the most current bid prices received from third parties. If an estimated selling price is not available, the Company estimates acreage value based on the price received for similar acreage in recent transactions by the Company or other market participants in the principal market.

The fair value measurements of assets acquired and liabilities assumed are measured on a nonrecurring basis on the acquisition date using an income valuation technique based on inputs that are not observable in the market and therefore represent Level 3 inputs. Significant inputs to the valuation of acquired oil and gas properties include estimates of: (i) reserves; (ii) production rates; (iii) future operating and development costs; (iv) future commodity prices, including price differentials; (v) future cash flows; and (vi) a market participant-based weighted average cost of capital rate. These inputs require significant judgments and estimates by the Company's management at the time of the valuation and are subject to change. Refer to *Note 3 - Acquisitions, Divestiture, and Assets Held for Sale* for additional information on the fair value of assets acquired during the nine months ended September 30, 2014.

Asset Retirement Obligations

The Company utilizes the income valuation technique to determine the fair value of the asset retirement obligation liability at the point of inception by applying a credit-adjusted risk-free rate, which takes into account the Company's credit risk, the time value of money, and the current economic state, to the undiscounted expected abandonment cash flows. Given the unobservable nature of the inputs, the initial measurement of the asset retirement obligation liability is deemed to use Level 3 inputs.

ITEM 2. MANAGEMENT'S DISCUSSION AND ANALYSIS OF FINANCIAL CONDITION AND RESULTS OF OPERATIONS

This discussion and analysis contains forward-looking statements. Refer to Cautionary Information about Forward-Looking Statements at the end of this item for an explanation of these types of statements.

Overview of the Company, Highlights, and Outlook

General Overview

We are an independent energy company engaged in the acquisition, exploration, development, and production of oil, gas, and NGLs in onshore North America. Our assets include leading positions in the Eagle Ford shale and Bakken/Three Forks resource plays, oil-focused plays in the Powder River Basin and Permian Basin, and a position in an emerging play in East Texas. We have built a portfolio of onshore properties in the contiguous United States primarily through early entry into existing and emerging resource plays. This portfolio is comprised of properties with established production and reserves, prospective drilling opportunities, and unconventional resource prospects. We believe our strategy provides for stable and predictable production and reserves growth. By entering these plays early, we believe we can capture larger resource potential at a lower cost.

Our principal business strategy is to focus on the early capture of resource plays in order to create and then enhance value for our stockholders while maintaining a strong balance sheet. We strive to leverage industry-leading exploration and leasehold acquisition teams to quickly acquire and test new resource play concepts at a reasonable cost. Once we have identified potential value through these efforts, our goal is to develop such potential through top-tier operational and project execution, and as appropriate, high-grade our portfolio by selectively divesting assets. We regularly examine our portfolio for opportunities to improve the quality of our asset base in order to optimize our returns and preserve our financial strength.

In the third quarter of 2014, we had the following financial and operational results:

- Average net daily production for the three months endedSeptember 30, 2014, was 43.5 MBbls of oil, 386.5 MMcf of gas, and 34.6 MBbls of NGLs, for a
 quarterly equivalent daily production rate of 142.5 MBOE, compared with 138.8 MBOE for the same period in 2013. Please see additional discussion
 below under *Production Results*.
- Net income for the three months ended September 30, 2014, was \$208.9 million, or \$3.05 per diluted share, compared to net income for the three months ended September 30, 2013, of \$70.7 million, or \$1.04 per diluted share. The increase in net income in the current period is largely driven by an increase in the fair value of commodity derivative contracts. Please refer to the Comparison of Financial Results and Trends Between the Three Months Ended September 30, 2014, and 2013 below for additional discussion regarding the components of net income.
- Costs incurred for oil and gas property acquisitions and exploration and development activities for the three months endedSeptember 30, 2014, totaled \$1.0 billion. The majority of our drilling and completion costs incurred during this period were in our Eagle Ford shale and Bakken/Three Forks programs. Additionally, we acquired approximately \$367.6 million of proved and unproved properties in our Gooseneck prospect area and the Powder River Basin during the third quarter of 2014. Total costs incurred for the same period in 2013 were \$438.1 million. Please refer to *Overview of Liquidity and Capital Resources* below for additional discussion on how we expect to fund our capital program.
- Adjusted EBITDAX, a non-GAAP financial measure, for the three months endedSeptember 30, 2014, was \$406.2 million, compared to \$410.4 million for the same period in 2013. Please refer to Non-GAAP Financial Measures below for additional discussion, including our definition of adjusted EBITDAX and reconciliations of our GAAP net income and net cash provided by operating activities to adjusted EBITDAX.

Our financial condition and the results of our operations are significantly affected by the prices we receive for our oil, gas, and NGL production, which can fluctuate dramatically. We sell the majority of our gas under contracts using first-of-the-month index pricing, which means gas produced in a given month is sold at the first-of-the-month price regardless of the spot price on the day the gas is produced. For assets where high BTU gas is sold at the wellhead, we also receive additional value for the higher energy content contained in the gas stream. Our NGL production is generally sold using contracts paying us a monthly average of the posted OPIS daily settlement prices, adjusted for processing, transportation, and location differentials. Our oil and condensate are sold using contracts paying us various industry posted prices, most commonly NYMEX West Texas Intermediate ("WTI"). We are paid the average of the daily settlement price for the respective posted prices for the period in which the product is sold, adjusted for quality, transportation, American Petroleum Institute ("API") gravity, and location differentials. Substantially all of our oil production in our South Texas & Gulf Coast region is condensate. When we refer to realized oil, gas, and NGL prices below, the disclosed price represents the average price for the respective period, before the effects of derivative cash settlements, unless otherwise indicated.

The following table summarizes commodity price data, as well as the effects of derivative cash settlements as further discussed under the caption Derivative Activity below, for the second and third quarters of 2014, as well as the third quarter of 2013:

	For the Three Months Ended					
	Septem	September 30, 2014		June 30, 2014		tember 30, 2013
Crude Oil (per Bbl):				_		
Average daily NYMEX price	\$	97.60	\$	103.06	\$	105.82
Realized price, before the effects of derivative cash settlements	\$	86.56	\$	91.78	\$	96.44
Effects of derivative cash settlements	\$	(0.12)	\$	(5.18)	\$	(3.66)
Natural Gas:						
Average daily NYMEX price (per MMBtu)	\$	3.94	\$	4.59	\$	3.55
Realized price, before the effects of derivatives cash settlements (per Mcf)	\$	4.49	\$	4.87	\$	3.81
Effects of derivative cash settlements (per Mcf)	\$	(0.05)	\$	(0.36)	\$	0.29
Natural Gas Liquids (per Bbl):						
Average daily OPIS price	\$	39.37	\$	41.21	\$	40.23
Realized price, before the effects of derivative cash settlements	\$	34.86	\$	35.61	\$	34.01
Effects of derivative cash settlements	\$	0.61	\$	(0.02)	\$	0.49

Note: Average OPIS prices per barrel of NGL, historical or strip, are based on a product mix of 37% Ethane, 32% Propane, 6% Isobutane, 11% Normal Butane, and 14% Natural Gasoline for all periods presented. This product mix represents the industry standard composite barrel and does not necessarily represent our product mix for NGL production. Realized prices reflect our actual product mix.

While quoted NYMEX oil and gas and OPIS NGL prices are generally used as a basis for comparison within our industry, the prices we receive are affected by quality, energy content, location, and transportation differentials for these products.

We expect future prices for oil, gas, and NGLs to be volatile. In addition to supply and demand fundamentals, as a global commodity, the price of oil will continue to be impacted by real or perceived geopolitical risks in oil producing regions of the world, particularly the Middle East. The relative strength of the U.S. dollar compared to other currencies also could affect the price of oil. Concerns about lower forecasted levels of global economic growth combined with ample supply, particularly from producing countries in the Organization of Petroleum Exporting Countries ("OPEC"), have negatively impacted oil prices in recent weeks. The supply of NGLs in the United States is expected to continue to grow in the near term as a result of the number of industry participants targeting projects that produce these products. If demand does not keep pace with anticipated growth in NGL supply, prices could be negatively impacted. The prices of several NGL products correlate to the price of oil and accordingly are likely to directionally follow that market. Gas prices have been under sustained downward pressure due to high levels of supply in recent years, particularly in the Northeast United States. Longer term, we anticipate natural gas prices will remain near current levels. Changes to existing laws and regulations pertaining to the ability to export oil, gas, and NGLs also have the potential to impact the prices for these commodities. The following table summarizes 12-month strip prices for NYMEX WTI oil, NYMEX Henry Hub gas, and OPIS NGLs (same product mix as discussed under the table above) as of October 22, 2014, and September 30, 2014:

	As of O	ctober 22, 2014	As of	September 30, 2014
NYMEX WTI oil (per Bbl)	\$	79.31	\$	88.68
NYMEX Henry Hub gas (per MMBtu)	\$	3.68	\$	4.01
OPIS NGLs (per Bbl)	\$	32.44	\$	37.75

Derivative Activity

We use financial derivative instruments as part of our financial risk management program. We have a financial risk management policy governing our use of derivatives. The amount of our production covered by derivatives is driven by the amount of debt on our balance sheet and the level of capital commitments and long-term obligations we have in place. With our current derivative contracts, we believe we have established a base cash flow stream for our future operations and have partially reduced our exposure to volatility in commodity prices. Our use of costless collars for a portion of our derivatives allows us to participate in some of the upward movements in oil, gas, and NGL prices while also setting a price floor for a portion of our production. Please refer to *Note 10 - Derivative Financial Instruments* in Part I, Item 1 of this report for additional information regarding our oil, gas, and NGL derivatives.

The Dodd-Frank Wall Street Reform and Consumer Protection Act ("Dodd-Frank Act") included provisions requiring over-the-counter derivative transactions to be cleared through clearinghouses and traded on exchanges. On July 10, 2012, the Commodity Futures Trading Commission ("CFTC") and the SEC adopted final joint rules under Title VII of the Dodd-Frank Act, which define certain terms that determine what types of transactions will be subject to regulation under the Dodd-Frank Act swap rules. The issuance of these final rules also triggers compliance dates for a number of other final Dodd-Frank Act rules, including new rules proposed by the CFTC governing margin requirements for uncleared swaps entered into by non-bank swap entities, and new rules proposed by U.S. banking regulators regarding margin requirements for uncleared swaps entered into by bank swap entities. The ultimate effect of these new rules on our business and any additional regulations is currently uncertain. Under CFTC rules we believe our derivative activity qualifies for the non-financial, commercial end-user exception, which exempts derivatives intended to hedge or mitigate commercial risk entered into by entities predominantly engaged in non-financial activity from the mandatory swap clearing requirement. However, we are not certain whether the provisions of the final rules and regulations will exempt us from the requirements to post margin in connection with commodity price risk management activities. Final rules and regulations on major provisions of the legislation, such as new margin requirements, are to be established through regulatory rulemaking. Although we cannot predict the ultimate outcome of these rulemakings, new rules and regulations in this area may result in increased costs and cash collateral requirements for the types of derivative instruments we use to manage our financial risks related to volatility in oil, gas, and NGL commodity prices.

Third Quarter 2014 Highlights and Outlook for the Remainder of 2014

Operational Activities. During the third quarter of 2014, we operated between four and five drilling rigs supported by two frac spreads in our operated Eagle Ford shale program in South Texas. We primarily focused on pad drilling in the northern portion of our acreage position, where there is a higher liquids contribution to our product mix. This year, our development program has shifted to utilizing longer laterals and completions with higher sand loading. Early results from these enhanced completions suggest significantly improved well performance. In the third quarter, production declined due to required shut-ins of producing wells during offset well completions. We believe we have secured the requisite services, such as gas pipeline takeaway capacity and drilling and completion services, to support our current development plans.

In our outside operated Eagle Ford shale program, the operator began the third quarter of 2014 running nine drilling rigs and dropped two rigs during the quarter. The operator expects to run seven rigs for the remainder of 2014. During the second quarter of 2014, the remainder of our carry under our Acquisition and Development Agreement with Mitsui E&P Texas LP ("Mitsui"), an indirect subsidiary of Mitsui & Co., Ltd. (the "Acquisition and Development Agreement"), was expended. Accordingly, we accrued our full share of drilling and completion costs for this program in the third quarter of 2014.

We have an ongoing exploration program to acquire leasehold and test concepts in new plays. In 2014, we are evaluating an emerging new venture play in East Texas. We are currently constructing a gathering system to allow for longer-term production tests on wells that we have drilled and completed. We expect this system to be completed in the fourth quarter of 2014.

In our Bakken/Three Forks program, we operated three drilling rigs during thethird quarter of 2014 focusing on infill drilling of our Raven/Bear Den and Gooseneck prospects in the North Dakota portion of the Williston Basin. We are monitoring the results of various well and completion designs and down-spacing tests of both our operated and non-operated properties and testing the Bakken interval on our Gooseneck and Stateline acreage. During the third quarter of 2014, we completed our acquisition of additional Gooseneck assets and paid \$325.2 million at closing, net of normal closing adjustments. We expect to increase rig activity in our Bakken/Three Forks program in the fourth quarter of 2014.

In our Powder River Basin program in Wyoming, we have been accelerating activity and expanding our acreage position through acquisitions from multiple sellers during the nine months ending September 30, 2014. We operated three drilling rigs during the third quarter, and we plan to add a fourth operated drilling rig in the fourth quarter of 2014.

In our Permian program, we operated two drilling rigs during thethird quarter of 2014 focused on horizontal testing and development of the Wolfcamp B interval in our Sweetie Peck prospect. In our Buffalo prospect in Gaines and Dawson Counties, Texas, we are monitoring the results of a recently completed Wolfcamp D test well. In the fourth quarter of 2014, we plan to test the Lower Spraberry in both our Sweetie Peck and Buffalo prospects.

Please refer to Overview of Liquidity and Capital Resources below for additional discussion regarding how we intend to fund our 2014 capital program.

Production Results. The table below provides a regional breakdown of our production for the third quarter of 2014:

	South Texas & Gulf Coast	Rocky Mountain	Permian	Mid-Continent	Total (1)
Oil (MMBbl)	1.7	1.8	0.5	_	4.0
Gas (Bcf)	28.0	1.7	1.1	4.7	35.6
NGLs (MMBbl)	3.2	_	_	_	3.2
Equivalent (MMBOE)	9.5	2.1	0.7	0.8	13.1
Avg. daily equivalents (MBOE/d)	103.6	22.8	7.3	8.8	142.5
Relative percentage	73%	16%	5%	6%	100%

⁽¹⁾ Totals may not add due to rounding.

Production increased for the three months ended September 30, 2014, compared to the same period in 2013, driven primarily by the continued development of our operated and non-operated Eagle Ford shale programs in our South Texas & Gulf Coast region. Please refer to Comparison of Financial Results and Trends Between the Three Months Ended September 30, 2014, and 2013 below for additional discussion on production.

Rocky Mountain Acquisitions. In the third quarter of 2014, we closed multiple transactions to acquire proved and unproved properties in our Gooseneck prospect area and the Powder River Basin for a total of approximately \$360.7 million. These acquisitions are subject to normal post-closing adjustments, which are expected to be completed during late 2014 and early 2015.

Subsequent Events. Subsequent to September 30, 2014, as a result of our regularly scheduled semi-annual borrowing base redetermination, the lending group increased our borrowing base under our credit facility from \$2.2 billion to \$2.4 billion with no change in the aggregate lender commitment amount of \$1.3 billion. Additionally, we acquired proved and unproved properties in our Gooseneck prospect area for total cash consideration of \$84.8 million. This acquisition is subject to normal post-closing adjustments.

Production Results. The table below provides a regional breakdown of our production for the firstnine months of 2014:

	South Texas & Gulf Coast	Rocky Mountain	Permian	Mid- Continent	Total (1)
Oil (MMBbl)	5.0	5.1	1.4	_	11.6
Gas (Bcf)	86.7	4.7	3.2	14.4	109.1
NGLs (MMBbl)	9.1	_	_	0.1	9.2
Equivalent (MMBOE)	28.5	6.0	2.0	2.5	39.0
Avg. daily equivalents (MBOE/d)	104.5	21.8	7.2	9.2	142.7
Relative percentage	73%	15%	5%	7%	100%

⁽¹⁾ Totals may not add due to rounding.

Please refer to Third Quarter 2014 Highlights and Outlook for the Remainder of 2014 above and Comparison of Financial Results and Trends Between the Nine Months Ended September 30, 2014, and 2013 below for additional discussion on production.

Costs Incurred in Oil and Gas Producing Activities. For the nine months ended September 30, 2014, we incurred \$2.0 billion in costs related to oil and gas property acquisitions and exploration and development activities, including both capitalized and expensed amounts. This amount includes a combined \$472.6 million of proved and unproved property acquisitions in the Gooseneck prospect area and the Powder River Basin. The majority of drilling and completion costs incurred during this period were in our Eagle Ford shale and Bakken/Three Forks programs. Please refer to Overview of Liquidity and Capital Resources below for additional discussion on how we expect to fund our capital program.

Rocky Mountain Divestiture. In the second quarter of 2014, we completed the divestiture of certain non-strategic assets in the Williston Basin for \$50.2 million of total divestiture proceeds. The estimated net gain on this divestiture is \$26.8 million. This divestiture is subject to normal post-closing adjustments, which are expected to be completed during the fourth quarter of 2014.

Financial Results of Operations and Additional Comparative Data

The tables below provide information regarding selected production and financial information. A detailed discussion follows.

	For the Three Months Ended							
	S	eptember 30,		June 30,		March 31,	December 31,	
		2014		2014		2014		2013
	(in millions, except for production data)							
Production (MMBOE)		13.1		13.4		12.5		13.2
Oil, gas, and NGL production revenue	\$	617.2	\$	654.7	\$	623.1	\$	593.7
Lease operating expense	\$	66.5	\$	62.8	\$	57.0	\$	61.1
Transportation costs	\$	81.5	\$	83.0	\$	79.2	\$	75.0
Production taxes	\$	30.4	\$	31.8	\$	27.5	\$	26.7
Depletion, depreciation, amortization, and asset retirement obligation liability accretion	\$	183.3	\$	187.8	\$	177.2	\$	202.6
Exploration	\$	34.6	\$	24.3	\$	21.3	\$	21.8
General and administrative	\$	41.7	\$	38.1	\$	35.1	\$	48.0
Net income	\$	208.9	\$	59.8	\$	65.6	\$	7.0

Selected Performance Metrics:

	For the Three Months Ended							
		September 30,		June 30,		March 31,		December 31,
		2014		2014		2014		2013
Average net daily production equivalent (MBOE/d)		142.5		147.0		138.6		143.8
Lease operating expense (per BOE)	\$	5.07	\$	4.69	\$	4.58	\$	4.62
Transportation costs (per BOE)	\$	6.22	\$	6.20	\$	6.35	\$	5.67
Production taxes as a percent of oil, gas, and NGL production revenue		4.9 %		4.9 %		4.4%		4.5 %
Depletion, depreciation, amortization, and asset retirement obligation liability accretion (per BOE)	\$	13.97	\$	14.03	\$	14.21	\$	15.31
General and administrative (per BOE)	\$	3.18	\$	2.85	\$	2.81	\$	3.63

Note: Amounts may not recalculate due to rounding.

A three-month and nine-month overview of selected production and financial information, including trends:

		For the Three I Ended Septem				Amount Change	Percent Change	Fo	For the Nine Months Ended September 30,			(Amount Change	Percent Change
		2014		2013	Between Between Periods Periods	Between Periods		2014		2013	Between Periods	Between Periods		
Net production volumes (1)													•	
Oil (MMBbl)		4.0		3.8		0.2	5 %		11.6		10.2		1.4	13 %
Gas (Bcf)		35.6		38.5		(2.9)	(8)%		109.1		109.9		(0.8)	(1)%
NGLs (MMBbl)		3.2		2.5		0.7	26 %		9.2		6.6		2.6	40 %
Equivalent (MMBOE)		13.1		12.8		0.3	3 %		39.0		35.1		3.9	11 %
Average net daily production (1)														
Oil (MBbl per day)		43.5		41.6		1.9	5 %		42.3		37.3		5.0	13 %
Gas (MMcf per day)		386.5		418.1		(31.6)	(8)%		399.5		402.4		(2.9)	(1)%
NGLs (MBbl per day)		34.6		27.5		7.1	26 %		33.8		24.2		9.6	40 %
Equivalent (MBOE per day)		142.5		138.8		3.7	3 %		142.7		128.6		14.1	11 %
Oil, gas, & NGL production revenue (in millions)														
Oil production revenue	\$	346.5	\$	368.9	\$	(22.4)	(6)%	\$	1,029.1	\$	946.6	\$	82.5	9 %
Gas production revenue		159.6		146.7		12.9	9 %		530.1		429.3		100.8	23 %
NGL production revenue		111.1		86.2		24.9	29 %		335.8		230.0		105.8	46 %
Total	\$	617.2	\$	601.8	\$	15.4	3 %	\$	1,895.0	\$	1,605.9	\$	289.1	18 %
Oil, gas, & NGL production expense (in millions)	_		_					_		_				
Lease operating expense	\$	66.5	\$	61.0	\$	5.5	9 %	\$	186.3	\$	171.9	\$	14.4	8 %
Transportation costs	•	81.5		68.8	-	12.7	18 %		243.7	*	183.2	*	60.5	33 %
Production taxes		30.4		29.1		1.3	4 %		89.7		79.2		10.5	13 %
Total	\$	178.4	\$	158.9	\$	19.5	12 %	\$	519.7	\$	434.3	\$	85.4	20 %
Realized price	_				Ť		/ •	_		÷		_		, ,
Oil (per Bbl)	\$	86.56	\$	96.44	\$	(9.88)	(10)%	\$	89.08	\$	92.93	\$	(3.85)	(4)%
Gas (per Mcf)	\$	4.49	\$	3.81	\$	0.68	18 %	\$	4.86	\$	3.91	\$	0.95	24 %
NGLs (per Bbl)	\$	34.86	\$	34.01	\$	0.85	2 %	\$	36.34	\$	34.77	\$	1.57	5 %
Per BOE	\$	47.06	\$	47.13	\$	(0.07)	— %	\$	48.63	\$	45.74	\$	2.89	6 %
Per BOE Data (1)	Ψ	47.00	Ψ	47.13	Ψ	(0.07)	70	Ψ	40.03	Ψ	43.74	Ψ	2.07	0 70
Production costs:														
Lease operating expense	\$	5.07	\$	4.77	\$	0.30	6 %	\$	4.78	\$	4.89	\$	(0.11)	(2)%
Transportation costs	\$	6.22	\$	5.38	\$	0.84	16 %	\$	6.25	\$	5.22	\$	1.03	20 %
Production taxes	\$	2.32	\$	2.29	\$	0.03	1 %	\$	2.30	\$	2.26	\$	0.04	2 %
General and administrative	\$	3.18	\$	2.66	\$	0.52	20 %	\$	2.95	\$	2.89	\$	0.06	2 %
Depletion, depreciation, amortization, and asset	φ	5.10	φ	2.00	φ	0.32	20 /0	Φ	2.93	ψ	2.09	φ	0.00	2 /0
retirement obligation liability accretion	\$	13.97	\$	15.33	\$	(1.36)	(9)%	\$	14.07	\$	17.67	\$	(3.60)	(20)%
Derivative cash settlement (2)	\$	(0.02)	\$	(0.14)	\$	0.12	(86)%	\$	(1.61)	\$	0.31	\$	(1.92)	(619)%
Earnings per share information														
Basic net income per common share	\$	3.10	\$	1.06	\$	2.04	192 %	\$	4.98	\$	2.47	\$	2.51	102 %
Diluted net income per common share	\$	3.05	\$	1.04	\$	2.04	192 %	\$	4.90	\$	2.41	\$	2.49	102 %
Basic weighted-average common shares outstanding (in thousands)	Ψ	67,379	Ψ	66,943	Ψ	436	1 %	Ψ	67,169	Ψ	66,486	Ψ	683	1 %
Diluted weighted-average common shares outstanding (in thousands)	1	68,430		68,253		177	— %		68,258		67,969		289	— %

⁽¹⁾ Amount and percentage changes may not recalculate due to rounding.
(2) We discontinued hedge accounting on January 1, 2011. As a result, fair values at December 31, 2010, were frozen in AOCL and were reclassified into earnings as the original derivative transactions settled, the last of which settled in the third quarter of 2013. For the three and nine months ended September 30, 2013, derivative cash settlements are included within the other operating revenues and derivative (gain) loss line items in the accompanying statements of operations. All derivative cash settlements for the three and nine months ended September 30, 2014, are included within the derivative (gain) loss line item only.

We present per BOE information because we use this information to evaluate our performance relative to our peers and to identify and measure trends we believe may require additional analysis. Average daily reported production for the three and nine months ended September 30, 2014, increased three percent and 11 percent, respectively, compared with the same periods in 2013, driven primarily by the development of our Eagle Ford shale assets. Please refer to Comparison of Financial Results and Trends Between the Three Months Ended September 30, 2014, and 2013 and Comparison of Financial Results and Trends Between the Nine Months Ended September 30, 2014, and 2013 below for additional discussion on changes in our production mix in2014.

Changes in production volumes, revenues, and costs reflect the cyclical and highly volatile nature of our industry. Our realized price on a per BOE basis for the three months ended September 30, 2014, was slightly lower than the same period in 2013. Our realized price on a per BOE basis increased six percent for the nine months ended September 30, 2014, compared to the same period in 2013, primarily due to higher natural gas prices.

Lease operating expense ("LOE") on a per BOE basis for the three months endedSeptember 30, 2014, increased six percent compared to the same period in 2013, whereas for the nine months ended September 30, 2014, LOE on a per BOE basis decreased two percent compared to the same period in 2013. Our LOE is comprised of recurring LOE, workover expense, and ad valorem tax expense. Absolute LOE increased nine percent and eight percent for the three and nine months ended September 30, 2014, respectively, compared to the same periods in 2013, driven primarily by increased recurring LOE and workover activity in our Rocky Mountain region and higher ad valorem tax expense in our South Texas & Gulf Coast region. Production volumes were lower in the third quarter of 2014 as compared to the second quarter of 2014 due to well shut-ins in the Eagle Ford for offset completions and weather-related delays in the Williston Basin. As such, volumes did not increase at the same rate as in previous quarters. This, along with increased workover activity in the Rocky Mountain region resulted in an increase in LOE on a per BOE basis for the three months ended September 30, 2014, production grew at a faster rate than absolute LOE resulting in a slight decrease in LOE on a per BOE basis for the nine months ended September 30, 2014. We experience volatility in our LOE as a result of seasonality in workover expense and the impact industry activity has on service provider costs. Overall, we expect absolute LOE to increase as a result of continued development and the recent acquisitions of properties in our Rocky Mountain region during the third quarter of 2014. We expect Company-wide production to increase at a faster rate than absolute LOE resulting in a slight decrease in LOE per BOE for the remainder of 2014.

Transportation costs on a per BOE basis for the three and nine months ended September 30, 2014, increased 16 percent and 20 percent, respectively, compared to the same periods in 2013. Our Eagle Ford shale program has meaningfully higher transportation expense per unit of production compared to our other regions. Ongoing development of the Eagle Ford shale has resulted in production from these assets becoming a larger portion of our total production, thereby increasing company-wide transportation expense per BOE over time. Included in transportation costs in our Eagle Ford shale program are charges for compression and fuel, which can fluctuate with the price of natural gas. We anticipate that we will recognize fluctuations in our per unit Eagle Ford shale transportation rate over time; however, we also anticipate company-wide transportation costs on a per BOE basis will increase as Eagle Ford shale production continues to grow and constitutes a larger portion of our total production mix.

Production taxes on a per BOE basis for the three and nine months ended September 30, 2014, increased one percent and two percent, respectively, compared to the same periods in 2013. We generally expect absolute production tax expense to trend with oil, gas, and NGL production revenue. Product mix, the location of production, and incentives to encourage oil and gas development can all impact or change the amount of production tax we recognize.

G&A expense on a per BOE basis for the three and nine months ended September 30, 2014, increased 20 percent and two percent, respectively, compared to the same periods in 2013. Absolute G&A expense increased between these two periods due to an increase in employee headcount. A portion of our G&A expense is linked to our profitability and cash flow, which are driven in large part by the realized commodity prices that we receive for our production. The Net Profits Plan and a portion of our short-term incentive compensation correlate with net cash flows and therefore are subject to variability. Additionally, because production decreased from the second quarter of 2014 to the third quarter of 2014, G&A expense on a per BOE basis increased. Generally, we expect G&A expense on a per BOE basis to decrease as we anticipate production going forward will increase at a faster rate than absolute G&A expense.

Depletion, depreciation, amortization, and asset retirement obligation liability accretion ("DD&A") expense on a per BOE basis for the three anchine months ended September 30, 2014, decreased nine percent and 20 percent, respectively, compared to the same periods in 2013. Our DD&A rate can fluctuate as a result of impairments, planned and closed divestitures, and changes in the mix of our production and the underlying proved reserve volumes. Our DD&A rate has decreased as assets with lower finding and development costs have become a larger portion of our total production mix. Our finding and development costs have benefited from a general decrease in well costs and an increase in recoveries per well, as well as from our outside operated Eagle Ford shale program, where we have added reserves with minimal associated costs due to our carry with Mitsui under our Acquisition and Development Agreement. This carry was exhausted during the second quarter of 2014.

Please refer to Comparison of Financial Results and Trends Between the Three Months Ended September 30, 2014, and 2013 and Comparison of Financial Results and Trends Between the Nine Months Ended September 30, 2014, and 2013 below for additional discussion on operating expenses.

Please refer to Note 9 - Earnings per Share in Part I, Item 1 of this report for additional discussion on the types of shares included in our basic and diluted net income per common share calculations.

Comparison of Financial Results and Trends Between the Three Months Ended September 30, 2014, and 2013

Oil, gas, and NGL production revenue and costs. The following table presents the regional changes in our production, and oil, gas, and NGL production revenue and costs between the three months ended September 30, 2014, and 2013:

	Average Net Daily Production Added (Lost)	Oil, Gas, & NGL Production Revenue Added (Lost)	Production Costs Increase (Decrease)		
	(MBOE/d)	(in millions)	(in millions)		
South Texas & Gulf Coast	14.6	\$ 65.5	\$ 20.7		
Rocky Mountain	1.6	(5.2)	5.7		
Permian	0.4	(4.3)	(2.8)		
Mid-Continent	(12.9)	(40.6)	(4.1)		
Total	3.7	\$ 15.4	\$ 19.5		

In our South Texas & Gulf Coast region, average net daily production increased 6 percent between the two periods as a result of drilling activity in our Eagle Ford shale program. This significant production growth in our Eagle Ford shale program exceeded the production decrease in our Mid-Continent region resulting from our Anadarko Basin asset divestiture in December 2013.

A three percent increase in equivalent production with relatively consistent realized prices on a per BOE basis resulted in athree percent increase in oil, gas, and NGL production revenue between the two periods. Please refer to *A three-month and nine-month overview of selected production and financial information, including trends* above for realized prices received before the effects of derivative cash settlements for the three months ended September 30, 2014, and 2013. Based on current levels of activity, we expect production volumes to continue to increase. We expect our realized prices to trend with commodity prices.

Other operating revenues. Gains and losses on divestiture activity are recorded to other operating revenues. We recorded a loss on divestiture activity of \$5.4 million and \$6.2 million for the three months ended September 30, 2014, and 2013, respectively, due largely to the write-down to fair value of certain assets held for sale. Other operating revenues also includes marketed gas system revenues, which decreased \$10.5 million between the three months ended September 30, 2014, and 2013. The decrease occurred as a result of our Anadarko Basin divestiture in December 2013, which reduced marketed gas volumes and the overall significance of marketed gas system revenues and expenses.

Oil, gas, and NGL production expense. Total production costs increased 12 percent for the three months ended September 30, 2014, compared with the same period of 2013, as a result of a three percent increase in net equivalent production volumes as well as an overall increase in transportation costs in our South Texas & Gulf Coast region. Please refer to A three-month and nine-month overview of selected production and financial information, including trendsabove for discussion of production costs on a per BOE basis.

Depletion, depreciation, amortization, and asset retirement obligation liability accretion. DD&A expense decreased six percent for the three-month period ended September 30, 2014, compared with the same period in 2013. Please refer to A three-month and nine-month overview of selected production and financial information, including trends above for discussion of DD&A on a per BOE basis, as our lower DD&A rate for the three months ended September 30, 2014, has resulted in decreased absolute DD&A expense compared to the same period in 2013.

Exploration. The components of exploration expense are summarized as follows:

	For th	the Three Months Ended September 30,			
		2014 2013			
		(in n	nillions)		
Geological and geophysical expenses	\$	1.4	\$	0.9	
Exploratory dry hole expense		16.3		_	
Overhead and other expenses		16.9		15.4	
Total	\$	34.6	\$	16.3	

Exploration expense for the three months ended September 30, 2014, increased 112 percent compared to the same period in 2013 as a result of an exploratory dry hole during the three months ended September 30, 2014. An exploratory project resulting in non-commercial quantities of oil, gas, or NGLs is deemed an exploratory dry hole and impacts the amount of exploration expense we record. We have an active exploration program, which can result in periodic dry hole expense.

Impairment of proved properties. We recorded no impairment of proved properties for the three months ended September 30, 2014. We recorded a \$5.9 million impairment of proved properties in the third quarter of 2013, related to our decision to no longer pursue the development of certain underperforming assets. We expect impairments of proved properties to be more likely to occur in periods of low commodity prices.

Abandonment and impairment of unproved properties. We recorded \$15.5 million of abandonment and impairment of unproved properties expense for the three months ended September 30, 2014, related to acreage we no longer intended to develop as a result of unsuccessful exploratory activities. For the same period in 2013, we recorded \$3.8 million of expense related to acreage we no longer intended to develop. We expect our abandonment and impairment of unproved properties expense to fluctuate with the timing of lease expirations and unsuccessful exploratory activities.

General and administrative. G&A expense increased 23 percent for the three months ended September 30, 2014, compared with the same period of 2013. The increase is due to an increase in employee headcount, which increased base and equity compensation, benefits, and general corporate office expenses incurred. Please refer to A three-month and nine-month overview of selected production and financial information, including trends above for discussion of G&A expense on a per BOE basis.

Change in Net Profits Plan liability. This non-cash item generally relates to the change in the estimated value of the associated noncurrent liability between reporting periods. For the three months ended September 30, 2014, we recorded a non-cash benefit of \$6.4 million. For the same period in 2013, we recorded a non-cash expense of \$940,000. The change in our liability is subject to estimation and may change from period to period based on assumptions used for production rates, reserve quantities, commodity pricing, discount rates, and production costs. We expect the change in our Net Profits Plan liability to correlate with fluctuations in commodity prices. Payments made to participants as a result of divestitures and ongoing operations also impact the change in the Net Profits Plan liability.

Derivative (gain) loss. We recognized a derivative gain of \$190.7 million for the three-month period ended September 30, 2014, which is comprised of a \$274,000 loss on cash settlements and a \$190.9 million increase in the fair value of commodity derivative contracts during the period. This compares to a loss of \$39.9 million for the same period in 2013, which is comprised of a \$38.6 million decrease in the fair value of commodity derivative contracts and a \$1.3 million loss on cash settlements during the period. The decrease in commodity strip prices, particularly oil, during the three months ended September 30, 2014, resulted in a significant increase in the net derivative asset position as of quarter-end. Please refer to Note 10 - Derivative Financial Instruments in Part I, Item 1 of this report for additional information.

Other operating expenses. This line item includes marketed gas system expense, which decreased\$9.5 million for the three months ended September 30, 2014, compared to the same period in 2013. This decrease occurred as a result of our Anadarko Basin divestiture in December 2013, which reduced marketed gas system volumes and the overall significance of marketed gas system revenues and expenses. Additionally, during the three months ended September 30, 2013, other operating expenses included \$3.6 million of expense related to an agreed clarification concerning royalty payment provisions of various leases on certain South Texas & Gulf Coast acreage.

Income tax expense. We recorded income tax expense of \$124.7 million for the three-month period ended September 30, 2014, compared to expense of \$42.3 million for the same period in 2013, resulting in effective tax rates of 37.4 percent and 37.5 percent, respectively.

Comparison of Financial Results and Trends Between the Nine Months Ended September 30, 2014, and 2013

Oil, gas, and NGL production revenue and costs. The following table presents the regional changes in our production, and oil, gas, and NGL production revenue and costs between the nine months ended September 30, 2014, and 2013:

	Average Net Daily Production Added (Lost)	Oil, Gas, & NGL Production Revenue Added (Lost)	Production Costs Increase (Decrease)
	(MBOE/d)	(in millions)	(in millions)
South Texas & Gulf Coast	23.7	\$ 328.0	\$ 75.6
Rocky Mountain	2.0	40.1	21.3
Permian	0.9	16.0	_
Mid-Continent	(12.5)	(95.0)	(11.5)
Total	14.1	\$ 289.1	\$ 85.4

In our South Texas & Gulf Coast region, average net daily production increased29 percent between the two periods as a result of drilling activity in our Eagle Ford shale program. Average daily production increased 10 percent in our Rocky Mountain region as a result of drilling activity in our Bakken/Three Forks program. This significant production growth in our Eagle Ford shale and Bakken/Three Forks programs exceeded the production decrease in our Mid-Continent region resulting from our Anadarko Basin asset divestiture in December 2013.

An 11 percent increase in production on a BOE basis combined with asix percent increase in the realized price per BOE resulted in an 18 percent increase in oil, gas, and NGL production revenue between the two periods.

Other operating revenues. Gains and losses on divestiture activity are recorded to other operating revenues. The gain realized on the sale of properties in our Rocky Mountain region during the second quarter of 2014 was mostly offset by losses recorded on assets classified as held for sale in the second and third quarters of 2014. Please refer to Third Quarter 2014 Highlights and Outlook for the Remainder of 2014 above for further discussion of the gain recorded on our divestiture of assets in our Rocky Mountain region. Other operating revenues also includes marketed gas system revenues, which decreased \$30.8 million for the nine months ended September 30, 2014, compared to the same period in 2013. This decrease occurred as a result of our Anadarko Basin divestiture in December 2013. This divestiture reduced marketed gas volumes and the overall significance of marketed gas system revenues and expenses. Partially offsetting this decrease was a \$10.7 million gain recorded in the second quarter of 2014 related to our settlement with Endeavour.

Oil, gas, and NGL production expense. Total production costs increased 20 percent for the nine months ended September 30, 2014, compared with the same period of 2013, as a result of an 11 percent increase in net equivalent production volumes as well as an overall increase in transportation costs in our South Texas & Gulf Coast region. Please refer to the caption A three-month and nine-month overview of selected production and financial information, including trendsabove for discussion of production costs on a per BOE basis.

Depletion, depreciation, amortization, and asset retirement obligation liability accretion. DD&A expense decreased 12 percent to \$548.3 million for the nine-month period ended September 30, 2014, compared with \$620.2 million for the same period in 2013. Please refer to A three-month and nine-month overview of selected production and financial information, including trends above for discussion of DD&A on a per BOE basis, as our lower DD&A rate for the nine months ended September 30, 2014, has resulted in decreased absolute DD&A expense compared to the same period in 2013.

Exploration. The components of exploration expense are summarized as follows:

	Fo	r the Nine Months Ended September 30,			
		2014	2013		
		(in millions)			
Geological and geophysical expenses	\$	7.2	\$	3.2	
Exploratory dry hole expense		22.8		5.9	
Overhead and other expenses		50.2		43.2	
Total	\$	80.2	\$	52.3	

Exploration expense for the nine months ended September 30, 2014, increased 53 percent compared to the same period in 2013 due to an exploratory dry hole being expensed in each of the second and third quarters of 2014. During the first quarter of 2014, we performed a seismic study, which increased geological and geophysical ("G&G") expenses. We have also experienced an overall increase in exploration overhead. Please refer to Comparison of Financial Results and Trends Between the Three Months Ended September 30, 2014, and 2013 above for additional discussion of exploration expense.

Impairment of proved properties. We recorded no impairment of proved properties expense for the nine months ended September 30, 2014, compared to \$61.7 million for the same period in 2013. In addition to the explanation under Comparison of Financial Results and Trends Between the Three Months Ended September 30, 2014, and 2013, above, we recorded a \$21.2 million impairment of proved properties in the first quarter of 2013 related to Olmos interval, dry gas assets in our South Texas & Gulf Coast region as a result of commencing a plugging and abandonment program. We also recorded a \$34.6 million impairment of proved properties in the second quarter of 2013 related to our decision to no longer pursue the development of certain underperforming assets.

Abandonment and impairment of unproved properties. For the nine months ended September 30, 2014, and 2013, we recorded abandonment and impairment of unproved properties expense of \$18.5 million and \$8.5 million, respectively, related to acreage we no longer intended to develop. Please refer to Comparison of Financial Results and Trends Between the Three Months Ended September 30, 2014, and 2013 above for additional discussion.

General and administrative. G&A expense increased 13 percent to \$114.9 million for the nine months ended September 30, 2014, compared with \$101.6 million for the same period of 2013, as a result of increased employee headcount, which increased base and equity compensation, benefits, and general corporate office expenses incurred. Please refer to A three-month and nine-month overview of selected production and financial information, including trendsabove for additional discussion of G&A on a per BOE basis.

Change in Net Profits Plan liability. For the nine months ended September 30, 2014, and 2013, we recorded a non-cash benefit of \$15.3 million and \$6.4 million, respectively. The increase in the benefit between these two periods is mostly due to payments made as a result of the divestiture of properties in our Rocky Mountain region during the second quarter of 2014. Please refer to Comparison of Financial Results and Trends Between the Three Months Ended September 30, 2014, and 2013 above for additional discussion

Derivative (gain) loss. We recognized a derivative loss of \$33.5 million for the nine-month period ended September 30, 2014, which is comprised of a \$62.9 million loss on cash settlements and a \$29.4 million increase in the fair value of commodity derivative contracts during the period. This compares to a gain of \$14.7 million for the same period in 2013, which consists of a \$12.7 million gain on cash settlements and a \$2.0 million increase in the fair value of commodity derivative contracts during the period. The loss on derivative cash settlements during the nine months ended September 30, 2014, is partially offset by the decline in oil commodity strip pricing during the period resulting in an increase in the fair value of derivative contracts. The derivative gain for the nine months ended September 30, 2013, is driven by favorable cash settlements on gas contracts. Please refer to Note 10 - Derivative Financial Instruments in Part I, Item 1 of this report for additional information.

Other operating expenses. This line item includes marketed gas system expense, which decreased \$28.3 million in the first nine months of 2014 compared to the same period in 2013. This decrease occurred as a result of our Anadarko Basin divestiture in December 2013, reducing marketed gas system volumes and the overall significance of marketed gas system revenues and expenses. Additionally, during the nine months ended September 30, 2013, other operating expenses included \$17.8 million of expense related to an agreed clarification concerning royalty payment provisions of various leases on certain South Texas & Gulf Coast acreage.

Income tax expense. We recorded income tax expense of \$199.7 million for the nine-month period ended September 30, 2014, compared to expense of \$98.9 million for the same period in 2013, resulting in effective tax rates of 37.4 percent and 37.6 percent, respectively. The decrease in the rate is partially attributable to the Anadarko Basin divestiture that closed at the end of 2013. The sale of assets in a higher rate state caused a decrease in the composition of our blended state tax rate for future years. However, state cash taxes are higher as a result of the estimated Texas margin tax.

Overview of Liquidity and Capital Resources

We believe we have sufficient liquidity and capital resources to execute our business plan for the foreseeable future. We continue to manage the duration and level of our drilling and completion service commitments in order to provide flexibility to reduce activity and capital expenditures in periods of prolonged commodity price decline.

Sources of Cash

We currently expect our remaining 2014 capital program to be funded by cash flows from operations and proceeds from 2014 divestitures, supplemented by borrowings under our credit facility. Although we anticipate cash flow from these sources will be sufficient to fund our remaining expected 2014 capital program, we may also elect to access the capital markets, depending on prevailing market conditions, as well as divest of additional non-strategic oil and gas properties to provide additional sources of funding. From time to time, we may enter into carrying cost funding and sharing arrangements with third parties for particular exploration and/or development programs. All of our sources of liquidity can be impacted by the general condition of the broader economy and by fluctuations in commodity prices, operating costs, and volumes produced, all of which affect us and our industry. We have no control over the market prices for oil, gas, or NGLs, although we are able to influence the amount of our realized revenues from our oil, gas, and NGL sales through the use of derivative contracts as part of our commodity price risk management program. Historically, decreases in commodity prices have limited our industry's access to capital markets. The borrowing base under our credit facility could be reduced as a result of lower commodity prices, divestitures of proved properties, or newly issued debt.

In late 2011, we consummated our Acquisition and Development Agreement with Mitsui, pursuant to which Mitsui funded, or carried, 90 percent of certain drilling and completion costs attributable to our remaining interest in our non-operated Eagle Ford shale acreage until \$680.0 million was expended on our behalf. The remaining carry was utilized during the second quarter of 2014, at which point we became responsible for funding our share of drilling and completion costs.

Proposals to reform the IRC, which include eliminating or reducing current tax deductions for intangible drilling costs, depreciation of equipment acquisition costs, the domestic production activities deduction, and percentage depletion, continue to circulate. We expect that future legislation modifying or eliminating these deductions would reduce net operating cash flows over time, thereby reducing funding available for our exploration and development capital programs, as well as funding available to our peers in the industry. If enacted at some point in the future, these funding reductions could have a significant adverse effect on drilling in the United States for a number of years.

Credit Facility

During the second quarter of 2013, we and our lenders entered into a Fifth Amended and Restated Credit Agreement. The credit facility has a maximum loan amount of \$2.5 billion, current aggregate lender commitments of \$1.3 billion, and a maturity date of April 12, 2018. The borrowing base under the credit facility as of the filing date of this report is \$2.4 billion and is subject to regular semi-annual redeterminations. We believe the current commitment amount is sufficient to meet our anticipated liquidity and operating needs. No individual bank participating in our credit facility represents more than 10 percent of the lending commitments under the credit facility. Borrowings under our credit facility are secured by mortgages on at least 75 percent of our proved oil and gas properties. Please refer to *Note 5 - Long-term Debt* in Part I, Item 1 of this report for additional discussion as well as the presentation of the outstanding balance, total amount of letters of credit, and available borrowing capacity under our credit facility as of October 22, 2014, September 30, 2014, and December 31, 2013.

We are subject to customary covenants under our credit facility, including limitations on dividend payments and requirements to maintain certain financial ratios, which include debt to adjusted EBITDAX, as defined by our credit agreement as the ratio of debt to 12-month trailing adjusted EBITDAX, of less than 4.0, and an adjusted current ratio, as defined by our credit agreement, of no less than 1.0. As of September 30, 2014, and as of the filing date of this report, we are in compliance with all covenants under our credit facility.

Operating cash flow and cash received from the divestiture of properties were sufficient in meeting our capital expenditure needs through the first half of 2014. During the third quarter of 2014, we began to draw upon our credit facility, primarily to fund acquisitions of oil and gas properties. Our daily weighted-average credit facility balance was \$41.7 million and \$14.1 million for the three and nine months ended September 30, 2014, respectively. Our daily weighted-average credit facility balance was \$72.8 million and \$239.2 million for the three and nine months ended September 30, 2013, respectively. Our daily weighted-average credit facility balance was lower throughout 2014 as a result of proceeds received from property divestitures in the fourth quarter of 2013. Cash flows provided by our operating activities, proceeds received from divestitures of properties, and the amount of our capital expenditures all impact the amount we have borrowed under our credit facility.

Weighted-Average Interest Rates

Our calculated weighted-average interest rates include paid and accrued interest, cash fees on the unused portion of the credit facility's aggregate commitment amount, letter of credit fees, and the non-cash amortization of deferred financing costs. Our calculated weighted-average borrowing rates include paid and accrued interest only.

The following table presents our weighted-average interest rates and our weighted-average borrowing rates for the three andnine months ended September 30, 2014, and 2013:

	For the Three Months F 30,	nded September	For the Nine Months En	ded September 30,
	2014	2013	2014	2013
Weighted-average interest rate	6.6%	6.5 %	6.7%	6.2 %
Weighted-average borrowing rate	5.9%	5.9%	6.0%	5.6%

Our weighted-average interest rates and weighted average borrowing rates in 2013 and 2014 have been impacted by the issuance of the 2024 Notes in the second quarter of 2013. This event, as well as the divestiture of properties, including our Anadarko Basin divestiture in December 2013, impacted the average balance on our revolving credit facility and the fees paid on the unused portion of our aggregate commitment.

Uses of Cash

We use cash for the acquisition, exploration, and development of oil and gas properties and for the payment of operating and G&A costs, income taxes, dividends, and debt obligations, including interest. Expenditures for the exploration and development of oil and gas properties are the primary use of our capital resources. In the first nine months of 2014, we spent \$1.8 billion for exploration and development capital activities and proved and unproved oil and gas property acquisitions. This amount differs from the cost incurred amount, which is accrual-based and includes asset retirement obligation, G&G, and exploration overhead amounts.

The amount and allocation of future capital expenditures will depend upon a number of factors, including the number and size of available acquisition and drilling opportunities, our ability to assimilate acquisitions and execute our drilling program, and our cash flows from operating, investing, and financing activities. In addition, the impact of oil, gas, and NGL prices on investment opportunities, the availability of capital, and the timing and results of our operated and non-operated development and exploratory activities may lead to changes in funding requirements for future development. We periodically review our capital expenditure budget to assess changes in current and projected cash flows, acquisition and divestiture activities, debt requirements, and other factors.

We may from time to time repurchase certain amounts of our outstanding debt securities for cash and/or through exchanges for other securities. Such repurchases or exchanges may be made in open market transactions, privately negotiated transactions, or otherwise. Any such repurchases or exchanges will depend on prevailing market conditions, our liquidity requirements, contractual restrictions, compliance with securities laws, and other factors. The amounts involved in any such transaction may be material.

As of the filing date of this report, we could repurchase up to 3,072,184 shares of our common stock under our stock repurchase program, subject to the approval of our Board of Directors. Shares may be repurchased from time to time in the open market, or in privately negotiated transactions, subject to market conditions and other factors, including certain provisions of our credit facility, the indentures governing our Senior Notes, compliance with securities laws, and the terms and provisions of our stock repurchase program. Our Board of Directors reviews this program as part of the allocation of our capital. We currently do not plan to repurchase any shares in 2014.

The following table presents changes in cash flows between thenine months ended September 30, 2014, and 2013. The analysis following the table should be read in conjunction with our condensed consolidated statements of cash flows in Part I, Item 1 of this report.

	For the	For the Nine Months Ended September 30,				mount Change	Percent Change	
		2014		2013		etween Periods	Between Periods	
	'-	(in m	illions)				
Net cash provided by operating activities	\$	1,075.1	\$	1,000.9	\$	74.2	7 %	
Net cash used in investing activities	\$	(1,736.0)	\$	(1,166.4)	\$	(569.6)	49 %	
Net cash provided by financing activities	\$	378.9	\$	159.8	\$	219.1	137 %	

Analysis of Cash Flow Changes Between the Nine Months Ended September 30, 2014, and 2013

Operating activities. Cash received from oil, gas, and NGL production revenues, net of transportation costs and production taxes, and including derivative cash settlements, increased \$182.2 million, or 14 percent, to \$1.5 billion for the first nine months of 2014, compared to the same period in 2013. This increase was due to an increase in production volumes and an increase in our realized equivalent price, including the effects of derivative cash settlements. Cash paid for LOE increased \$4.2 million to \$170.6 million for the first nine months of 2014, compared to the same period in 2013, due to increased production and increased workover activity in our Rocky Mountain region, as well as the timing of cash payments. Cash paid for interest, net of capitalized interest, during the first nine months of 2014 increased \$19.3 million compared to the same period in 2013, due to the first interest payment on our 2024 Notes being made in the first quarter of 2014. Additionally, cash bonuses paid in 2014 for the 2013 performance year were \$41.7 million compared to \$16.0 million paid in 2013 for the 2012 performance year.

Investing activities. Capital expenditures for the first nine months of 2014 increased 18 percent compared with the same period in 2013 due to increased spending in our Eagle Ford shale and Bakken/Three Forks programs. Acquisitions of proved and unproved properties increased \$397.3 million as a result of property acquisitions in our Gooseneck prospect area and Powder River Basin in the second and third quarters of 2014.

Financing activities. We had net borrowings under our credit facility of \$390.0 million during the nine months ended September 30, 2014, and net repayments of \$312.0 million during the same period in 2013.

Interest Rate Risk and Commodity Price Risk

Our credit agreement allows us to fix the interest rate for all or a portion of the principal balance of our revolving credit facility for a period up to six months; however, our borrowings are generally made with interest rates fixed for one month. Therefore, to the extent we do not repay the principal, our borrowings are rolled over and the interest rate is reset based on the current LIBOR or Alternate Base Rate ("ABR") as applicable. As a result, changes in interest rates can impact results of operations and cash flows. Changes in interest rates do not impact the amount of interest we pay on our fixed-rate Senior Notes, but can impact their fair market values. As of September 30, 2014, we had \$1.6 billion of fixed-rate debt outstanding. Please refer to Note 11 - Fair Value Measurements in Part I, Item 1 of this report for additional discussion on the fair value of our Senior Notes. As of September 30, 2014, we had \$390.0 million of floating-rate debt outstanding. The carrying amount of our floating rate debt at September 30, 2014, approximates its fair value.

The prices we receive for our oil, gas, and NGL production heavily impact our revenue, overall profitability, access to capital, and future rate of production growth. Oil, gas, and NGL prices are subject to wide fluctuations in response to changes in supply and demand. Historically, the markets for oil, gas, and NGLs have been volatile, and these markets will likely continue to be volatile in the future. The prices we receive for our production depend on numerous factors beyond our control.

We enter into commodity derivative contracts in order to reduce the impact of fluctuations in commodity prices. Please refer to *Note 10 - Derivative Financial Instruments* in Part I, Item 1 of this report for information about our oil, gas, and NGL derivative contracts.

There has been no material change to the interest rate risk analysis or oil and gas price sensitivity analysis previously disclosed. Please refer to Interest Rate Risk and Commodity Price Risk in Management's Discussion and Analysis of Financial Condition and Results of Operations in Part II, Item 7 of our 2013 Form 10-K for further discussion.

Off-Balance Sheet Arrangements

We have not participated in transactions that generate relationships with unconsolidated entities or financial partnerships, such as entities often referred to as structured finance or special purpose entities, which would have been established for the purpose of facilitating off-balance sheet arrangements or other contractually narrow or limited purposes. We evaluate our transactions to determine if any variable interest entities exist. If we determine that we are the primary beneficiary of a variable interest entity, that entity is consolidated into our consolidated financial statements.

Critical Accounting Policies and Estimates

Please refer to the corresponding section in Part II, Item 7 of our 2013 Form 10-K and to the footnote disclosures included in Part I, Item 1 of this report for a discussion of our accounting policies and estimates.

New Accounting Pronouncements

Please refer to Note 2 - Basis of Presentation, Significant Accounting Policies, and Recently Issued Accounting Standards under Part I, Item 1 of this report for new accounting matters.

Non-GAAP Financial Measures

Adjusted EBITDAX represents income before interest expense, other non-operating income or expense, income taxes, depreciation, depletion, amortization, and accretion expense, exploration expense, property impairments, non-cash stock compensation expense, derivative gains and losses net of cash settlements, change in the Net Profits Plan liability, and gains and losses on divestitures. Adjusted EBITDAX excludes certain items that we believe affect the comparability of operating results and can exclude items that are generally one-time in nature or whose timing and/or amount cannot be reasonably estimated. Adjusted EBITDAX is a non-GAAP measure that is presented because we believe that it provides useful additional information to investors and analysts, as a performance measure, for analysis of our ability to internally generate funds for exploration, development, acquisitions, and to service debt. We are also subject to a financial covenant under our credit facility based on our debt to adjusted EBITDAX ratio. In addition, adjusted EBITDAX is widely used by professional research analysts and others in the valuation, comparison, and investment recommendations of companies in the oil and gas exploration and production industry, and many investors use the published research of industry research analysts in making investment decisions. Adjusted EBITDAX should not be considered in isolation or as a substitute for net income, income from operations, net cash provided by operating activities, or profitability or liquidity measures prepared under GAAP. Because adjusted EBITDAX excludes some, but not all items that affect net income and may vary among companies, the adjusted EBITDAX amounts presented may not be comparable to similar metrics of other companies. The following table provides reconciliations of our net income and net cash provided by operating activities to adjusted EBITDAX for the periods presented:

	For the Three Months Ended September 30,				For the Nine I Septem			
	2014	2013		2014			2013	
	 (in thou			usands)				
Net income (GAAP)	\$ 208,938	\$	70,690	\$	334,325	\$	163,939	
Interest expense	22,621		24,488		70,851		65,170	
Other non-operating (income) expense, net	672		(28)		2,493		(64)	
Income tax expense	124,748		42,334		199,660		98,921	
Depreciation, depletion, amortization, and asset retirement obligation liability accretion	183,259		195,792		548,255		620,232	
Exploration (1)	32,155		14,176		74,696		45,783	
Impairment of proved properties	_		5,935		_		61,706	
Abandonment and impairment of unproved properties	15,522		3,818		18,487		8,459	
Stock-based compensation expense	10,227		7,427		24,568		25,495	
Derivative (gain) loss	(190,661)		39,933		33,470		(14,685)	
Derivative cash settlement gain (loss)	(274)		(1,288)		(62,894)		12,715	
Change in Net Profits Plan liability	(6,399)		940		(15,280)		(6,423)	
(Gain) loss on divestiture activity (2)	5,432		6,216		(52)		510	
Adjusted EBITDAX (Non-GAAP)	406,240		410,433		1,228,579		1,081,758	
Interest expense	(22,621)		(24,488)		(70,851)		(65,170)	
Other non-operating income (expense), net	(672)		28		(2,493)		64	
Income tax expense	(124,748)		(42,334)		(199,660)		(98,921)	
Exploration (1)	(32,155)		(14,176)		(74,696)		(45,783)	
Exploratory dry hole expense	16,385		(8)		22,844		5,878	
Amortization of deferred financing costs	1,479		1,474		4,433		3,914	
Deferred income taxes	124,269		42,380		198,180		98,619	
Plugging and abandonment	(2,974)		(3,707)		(6,193)		(7,453)	
Changes in current assets and liabilities	(7,127)		37,752		(22,087)		25,034	
Other, net	1,893		(2,840)		(2,934)		2,929	
Net cash provided by operating activities (GAAP)	\$ 359,969	\$	404,514	\$	1,075,122	\$	1,000,869	

⁽¹⁾ Stock-based compensation expense is a component of exploration expense and general and administrative expense on the accompanying statements of operations. Therefore, the exploration line items shown in the reconciliation above will vary from the amount shown on the accompanying statements of operations because of the component of stock-based compensation expense recorded to exploration.

^{(2) (}Gain) loss on divestiture activity is included within the other operating revenues line item in the accompanying statements of operations.

Cautionary Information about Forward-Looking Statements

This report contains "forward-looking statements" within the meaning of Section 27A of the Securities Act and Section 21E of the Securities Exchange Act of 1934 (the "Exchange Act"). All statements, other than statements of historical facts, included in this report that address activities, events, or developments with respect to our financial condition, results of operations, or economic performance that we expect, believe, or anticipate will or may occur in the future, or that address plans and objectives of management for future operations, are forward-looking statements. The words "anticipate," "assume," "believe," "budget," "estimate," "expect," "forecast," "intend," "plan," "project," "will," and similar expressions are intended to identify forward-looking statements. Forward-looking statements appear in a number of places in this report, and include statements about such matters as:

- the amount and nature of future capital expenditures and the availability of liquidity and capital resources to fund capital
 expenditures;
- the drilling of wells and other exploration and development activities and plans, as well as possible future acquisitions;
- the possible divestiture or farm-down of, or joint venture relating to, certain properties;
- proved reserve estimates and the estimates of both future net revenues and the present value of future net revenues associated with those proved reserve estimates;
- pending acquisitions of oil and gas assets;
- future oil, gas, and NGL production estimates:
- our outlook on future oil, gas, and NGL prices, well costs, and service costs;
- cash flows, anticipated liquidity, and the future repayment of debt:
- business strategies and other plans and objectives for future operations, including plans for expansion and growth of operations or to defer capital investment, and our outlook on our future financial condition or results of operations; and
- other similar matters such as those discussed in the Management's Discussion and Analysis of Financial Condition and Results of Operations section of this report.

Our forward-looking statements are based on assumptions and analyses made by us in light of our experience and our perception of historical trends, current conditions, expected future developments, and other factors that we believe are appropriate under the circumstances. These statements are subject to a number of known and unknown risks and uncertainties, which may cause our actual results and performance to be materially different from any future results or performance expressed or implied by the forward-looking statements. Some of these risks are described under *Risk Factors* in Item 1A of our 2013 Form 10-K, and include such factors as:

- the volatility of oil, gas, and NGL prices, and the effect it may have on our profitability, financial condition, cash flows, access to capital, and ability to grow
 production volumes and/or proved reserves;
- weakness in economic conditions and uncertainty in financial markets:
- our ability to replace reserves in order to sustain production;
- our ability to raise the substantial amount of capital that is required to develop and/or replace our reserves:
- our ability to compete against competitors that have greater financial, technical, and human resources;
- our ability to attract and retain key personnel;
- the imprecise estimations of our actual quantities and present value of proved oil, gas, and NGL reserves:
- the uncertainty in evaluating recoverable reserves and estimating expected benefits or liabilities;

- the possibility that exploration and development drilling may not result in commercially producible reserves:
- our limited control over activities on non-operated properties;
- our reliance on the skill and expertise of third-party service providers on our operated properties;
- the possibility that title to properties in which we have an interest may be defective;
- the possibility that our planned drilling in existing or emerging resource plays using some of the latest available horizontal drilling and completion techniques is subject to drilling and completion risks and may not meet our expectations for reserves or production;
- the uncertainties associated with divestitures, joint ventures, farm-downs, farm-outs, and similar transactions with respect to certain assets, including whether such transactions will be consummated or completed in the form or timing and for the value that we anticipate;
- the uncertainties associated with enhanced recovery methods:
- our commodity derivative contracts may result in financial losses or may limit the prices that we receive for oil, gas, and NGL sales;
- the inability of one or more of our service providers, customers, or contractual counterparties to meet their
 obligations;
- our ability to deliver necessary quantities of oil, gas, or NGLs to contractual counterparties;
- price declines or unsuccessful exploration efforts resulting in write-downs of our asset carrying values:
- the impact that lower oil, gas, or NGL prices could have on the amount we are able to borrow under our credit facility:
- the possibility that our amount of debt may limit our ability to obtain financing for acquisitions, make us more vulnerable to adverse economic conditions, and make it more difficult for us to make payments on our debt;
- the possibility that covenants in our debt agreements may limit our discretion in the operation of our business, prohibit us from engaging in beneficial transactions, or lead to the accelerated payment of our debt;
- operating and environmental risks and hazards that could result in substantial losses:
- the impact of seasonal weather conditions and lease stipulations on our ability to conduct drilling activities:
- our ability to acquire adequate supplies of water and dispose of or recycle water we use at a reasonable cost in accordance with environmental and other applicable rules:
- complex laws and regulations, including environmental regulations, that result in substantial costs and other risks:
- the availability and capacity of gathering, transportation, processing, and/or refining facilities:
- our ability to sell and/or receive market prices for our oil, gas, and NGLs:
- new technologies may cause our current exploration and drilling methods to become
 absolute:
- the possibility of security threats, including terrorist attacks and cybersecurity breaches, against, or otherwise impacting, our facilities and systems;
- litigation, environmental matters, the potential impact of legislation and government regulations, and the use of management estimates regarding such matters

We caution you that forward-looking statements are not guarantees of future performance and that actual results or performance may be materially different from those expressed or implied in the forward-looking statements. The forward-looking statements in

this report speak	as of the f	iling date of this report.	Although we may	from time to ti	me voluntarily	update our pri	or forward-	looking statements,	we disclaim any	commitment commitment
to do so except as	required	by securities laws.								

ITEM 3. QUANTITATIVE AND QUALITATIVE DISCLOSURES ABOUT MARKET RISK

The information required by this item is provided under Interest Rate Risk and Commodity Price Risk in Item 2 above and is incorporated herein by reference. Please also refer to the information under Interest Rate Risk and Commodity Price Risk in Management's Discussion and Analysis of Financial Condition and Results of Operations in Part II, Item 7 of our 2013 Form 10-K.

ITEM 4. CONTROLS AND PROCEDURES

Evaluation of Disclosure Controls and Procedures

We maintain a system of disclosure controls and procedures that are designed to reasonably ensure that information required to be disclosed in our SEC reports is recorded, processed, summarized, and reported within the time periods specified in the SEC's rules and forms, and to reasonably ensure that such information is accumulated and communicated to our management, including the Chief Executive Officer and the Chief Financial Officer, as appropriate, to allow for timely decisions regarding required disclosure.

Our management, including our Chief Executive Officer and Chief Financial Officer, does not expect that our disclosure controls and procedures (as defined in Rules 13a-15(e) and 15d-15(e) of the Exchange Act) ("Disclosure Controls") will prevent all errors and all fraud. A control system, no matter how well conceived and operated, can provide only reasonable, not absolute, assurance that the objectives of the control system are met. Further, the design of a control system must reflect the fact that there are resource constraints, and the benefits of controls must be considered relative to their costs. Because of the inherent limitations in all control systems, no evaluation of controls can provide absolute assurance that all control issues and instances of fraud, if any, within the Company have been detected. These inherent limitations include the realities that judgments in decision-making can be faulty, and that breakdowns can occur because of simple error or mistake. Additionally, controls can be circumvented by the individual acts of some persons, by collusion of two or more people, or by management override of the control. The design of any system of controls also is based in part upon certain assumptions about the likelihood of future events and there can be no assurance that any design will succeed in achieving its stated goals under all potential future conditions. Because of the inherent limitations in a cost-effective control system, misstatements due to error or fraud may occur and not be detected. We monitor our Disclosure Controls and make modifications as necessary; our intent in this regard is that the Disclosure Controls will be modified as systems change and conditions warrant.

An evaluation of the effectiveness of the design and operation of our Disclosure Controls was performed as of the end of the period covered by this report. This evaluation was performed under the supervision and with the participation of our management, including our Chief Executive Officer and Chief Financial Officer. Based upon that evaluation, our Chief Executive Officer and Chief Financial Officer concluded that our Disclosure Controls are effective at a reasonable assurance level.

Changes in Internal Control Over Financial Reporting

There have been no changes during the third quarter of 2014 that have materially affected, or are reasonably likely to materially affect, our internal control over financial reporting.

PART II. OTHER INFORMATION

ITEM 1. LEGAL PROCEEDINGS

Except as noted in Part II, Item 1 of the Quarterly Report on Form 10-Q for the quarter ended June 30, 2014, there have been no other material changes to the legal proceedings as previously disclosed in our 2013 Form 10-K, under Part I, Item 3. See *Note 6 - Commitments and Contingencies*, in Part I, Item 1 of this report, for additional discussion.

ITEM 1A. RISK FACTORS

There have been no material changes to the risk factors as previously disclosed in our2013 Form 10-K.

ITEM 2. UNREGISTERED SALES OF EQUITY SECURITIES AND USE OF PROCEEDS

(c) The following table provides information about purchases by the Company or any "affiliated purchaser" (as defined in Rule 10b-18(a)(3) under the Exchange Act) during the fiscal quarter ended September 30, 2014, of shares of the Company's common stock, which is the sole class of equity securities registered by the Company pursuant to Section 12 of the Exchange Act:

PURCHASES OF EQUITY SECURITIES BY ISSUER AND AFFILIATED PURCHASERS

	(a)	(b)	(c)	(d)		
Period	Total Number of Shares Purchased ⁽¹⁾	Average Price Paid per Share	Total Number of Shares Purchased as Part of Publicly Announced Program	Maximum Number of Shares that May Yet Be Purchased Under the Program ⁽²⁾		
07/01/14 - 07/31/14	124,698	\$ 84.15	_	3,072,184		
08/01/14 - 08/31/14	212	78.70		3,072,184		
09/01/14 - 09/30/14	32	86.53		3,072,184		
Total:	124,942	\$ 84.14	_	3,072,184		

- (1) All shares purchased in the third quarter of 2014 were to offset tax withholding obligations that occurred upon the delivery of outstanding shares underlying RSUs and PSUs delivered under the terms of grants under our Equity Incentive Compensation Plan.
- (2) In July 2006, our Board of Directors approved an increase in the number of shares that may be repurchased under the original August 1998 authorization to up to 6,000,000 shares as of the effective date of the resolution. Accordingly, as of the date of this filing, we may repurchase up to 3,072,184 shares of common stock on a prospective basis, subject to the approval of our Board of Directors. The shares may be repurchased from time to time in open market transactions or privately negotiated transactions, subject to market conditions and other factors, including certain provisions of our credit facility, the indentures governing our Senior Notes and compliance with securities laws. Stock repurchases may be funded with existing cash balances, internal cash flow, or borrowings under our credit facility. The stock repurchase program may be suspended or discontinued at any time.

Our payment of cash dividends to our stockholders is subject to covenants under the terms of our credit facility that limit our annual dividend payments to no more than \$50.0 million per year. We are also subject to certain covenants under our Senior Notes that restrict certain payments, including dividends; provided, however, that the first \$6.5 million of dividends paid each year are not restricted by these covenants. We do not anticipate that these restrictions will limit our payment of dividends at our current rate for the foreseeable future if declared by our Board of Directors.

ITEM 6. EXHIBITS

The following exhibits are filed or furnished with or incorporated by reference into this report:

Exhibit	Description
2.1*†	Purchase Agreement dated July 29, 2014, between Baytex Energy USA LLC and SM Energy Company
31.1*	Certification of Chief Executive Officer pursuant to Section 302 of the Sarbanes - Oxley Act of 2002
31.2*	Certification of Chief Financial Officer pursuant to Section 302 of the Sarbanes - Oxley Act of 2002
32.1**	Certification pursuant to 18 U.S.C. Section 1350 as adopted pursuant to Section 906 of the Sarbanes - Oxley Act of 2002
101.INS*	XBRL Instance Document
101.SCH*	XBRL Schema Document
101.CAL*	XBRL Calculation Linkbase Document
101.LAB*	XBRL Label Linkbase Document
101.PRE*	XBRL Presentation Linkbase Document
101.DEF*	XBRL Taxonomy Extension Definition Linkbase Document
*	Filed with this report.
**	Furnished with this report.
†	Confidential Treatment has been requested with respect to portions of the exhibit. Such portions have been redacted and filed separately with the SEC.

SIGNATURES

Pursuant to the requirements of the Securities Exchange Act of 1934, the registrant has duly caused this report to be signed on its behalf by the undersigned hereunto duly authorized.

SM ENERGY COMPANY

October 29, 2014 By: /s/ ANTHONY J. BEST

Anthony J. Best Chief Executive Officer (Principal Executive Officer)

October 29, 2014 By: /s/ A. WADE PURSELL

A. Wade Pursell

Executive Vice President and Chief Financial Officer

(Principal Financial Officer)

October 29, 2014 By: /s/ MARK T. SOLOMON

Mark T. Solomon

Vice President - Controller and Assistant Secretary

(Principal Accounting Officer)

Portions of this Exhibit have been redacted pursuant to a request for confidential treatment under Rule 24b-2 of the General Rules and Regulations under the Securities Exchange Act. Omitted information marked "[***]" in this Exhibit has been filed with the Securities and Exchange Commission together with such request for confidential treatment.

Execution Version

PURCHASE AGREEMENT
BY AND AMONG
BAYTEX ENERGY USA LLC
AS SELLER
AND
SM ENERGY COMPANY
AS BUYER

DATED AS OF JULY 29, 2014

TABLE OF CONTENTS

ARTICLE	1 CERTAIN DEFINITIONS	1
1.1	Certain Defined Terms	1
1.2	References, Gender and Number	1
ARTICLE	2 PURCHASE AND SALE OF MEMBERSHIP INTERESTS; CONSIDERATION AND PAYMENT	2
2.1	Purchase and Sale of the Membership Interests	2
2.2	Consideration	2
2.3	Escrow	2
2.4	Payment of Closing Consideration	2
2.5	Purchase Price Adjustment	2
2.6	Post Closing Review	5
ARTICLE	3 REPRESENTATIONS AND WARRANTIES	5
3.1	Representations and Warranties of Seller Regarding the Subject Assets and the Company	5
3.2	Representations and Warranties of Seller Regarding Seller	11
3.3	Representations and Warranties of Buyer	12
ARTICLE	4 INVESTIGATION OF SUBJECT ASSETS; CONFIDENTIALITY	14
4.1	Investigation of Subject Assets	14
4.2	Obligations Under Confidentiality Agreement	14
ARTICLE	5 TITLE ADJUSTMENTS	15
5.1	General Disclaimer; No Title Warranty or Representations	15
5.2	Buyer's Title Review	15
5.3	Determination of Title Defects	18
5.4	Seller Title Credit	18
5.5	Deferred Claims and Disputes	19
5.6	No Duplication	20
5.7	Title Defect Deductible	20
ARTICLE	6 ENVIRONMENTAL MATTERS	21
6.1	Environmental Defects; Environmental Defect Amount	21
6.2	Environmental Defect Notice	21
6.3	Effect of Environmental Defect Notice	21
6.4	Seller's Right to Cure	22
6.5	Environmental Defect Adjustments	22
6.6	Environmental Defect Deductible	22
6.7	Subject Assets Exclusion	22
6.8	Deferred Environmental Claims and Disputes	22
6.9	Sole Remedies	24
6.1	NORM	24
ARTICLE	7 COVENANTS OF SELLER AND BUYER	24
7.1	Conduct of Business	24
7.2	Certain Restrictions Regarding the Conduct of Business	25
7.3	Press Releases	26
7.4	Records	26

7.5	Further Assurances	26
7.6	Transfer Requirements	26
7.7	Casualty and Condemnation	26
7.8	Seller Marks	27
7.9	Certain Tax Matters	27
7.1	Asset and Liability Transfer	30
7.11	No Shop/No Talk	31
7.12	Employee Covenant	31
ARTICLE	8 CLOSING CONDITIONS	33
8.1	Seller's Closing Conditions	33
8.2	Buyer's Closing Conditions	33
ARTICLE	9 CLOSING	35
9.1	Closing	35
9.2	Seller's Closing Obligations	35
9.3	Buyer's Closing Obligations	35
ARTICLE	10 SURVIVAL	36
10.1	Survival	36
10.2	Acknowledgement of Limitation of Recovery	36
ARTICLE	11 LIMITATIONS	36
11.1	Disclaimer of Warranties	36
11.2	Damages	37
ARTICLE	12 INDEMNIFICATION	38
12.1	• •	38
12.2	·	38
12.3		38
12.4	Seller's General Liability Limitation	40
12.5	·	41
12.6	Determination of Damages Related to Breach	41
12.7	·	41
ARTICLE	13 TERMINATION; REMEDIES	41
13.1	Termination	41
13.2	Remedies	42
ARTICLE	14 MISCELLANEOUS	44
14.1	Counterparts	44
14.2	Notices	44
14.3		45
14.4	Entire Agreement	45
14.5	Expenses	46
14.6	Successors and Assigns	46
14.7	Amendments and Waivers	46
14.8	Attorneys' Fees	46
14.9	Severability	46

14.1	Appendix, Schedules and Exhibits	46
14.11	Interpretation	47
14.12	Agreement for the Parties' Benefit Only	47
14.13	No Recordation	47
14.14	Time of Essence	48

EXHIBITS AND SCHEDULES

Exhibits

Exhibit A

Part I Leases

Part II Allocated Values

Part II(A) Bakken Drillsite Spacing Units and Wells

Part III Surface Contracts

Part IV Wells
Part V Equipment

Exhibit B Deposit Escrow Agreement
Exhibit C General Conveyance
Exhibit D Assumption Agreement

Exhibit E Assignment of Membership Interests
Exhibit F Certificate of Non-Foreign Status

Exhibit G Organizational Documents of the Company

Exhibit H Parent Guaranty

Schedules

Schedule 2.5(c) Newly Acquired Leases

Schedule 2.5(e)(v) Special Adjustment to the Purchase Price

Schedule 3.1(d) Actions

Schedule 3.1(e) Compliance with Laws
Schedule 3.1(g) Transfer Requirements
Schedule 3.1(h)(1) Material Contracts

Schedule 3.1(h)(2) Breaches of Material Contracts

Schedule 3.1(i) Tax Matters

Schedule 3.1(k) Proposed Operations or Expenditures

Schedule 3.1(1) **Environmental Matters** Schedule 3.1(n) Payments for Production Schedule 3.1(p) Calls on Production Schedule 3.1(s) Non-consent Elections Schedule 3.1(u) Advance Payments Schedule 3.1(v) Payout Balances Schedule 3.2(e) No Conflict Schedule 3.2(f) Seller Consents

Schedule 5.3(i)Certain Mergers not of RecordSchedule 6.1Excluded Environmental DefectsSchedule 7.2Certain Operational Covenants

Schedule 7.9 Certain Tax Covenants

Schedule A-1 Additional Permitted Encumbrances

PURCHASE AGREEMENT

THIS PURCHASE AGREEMENT (this "Agreement"), dated July 29, 2014 (the "Execution Date"), is by and among

BAYTEX ENERGY USA LLC, a Colorado limited liability company ("Seller"), and

SM ENERGY COMPANY, a Delaware corporation ("Buyer").

Seller and Buyer are hereinafter referred to individually as a "Party" and collectively as the "Parties."

WHEREAS, Baytex Energy USA Ltd. was the owner of certain assets, contracts, benefits and obligations associated with certain oil and gas operations in North Dakota and described herein, and on July 14, 2014, Baytex Energy USA Ltd. merged with and into Seller; as used in this Agreement, all references to "Seller" include Baytex Energy USA Ltd.;

WHEREAS, immediately prior to the Closing, the Seller will create a new Delaware limited liability company (the "Company") and will contribute the Subject Assets to the Company and cause the Company to assume the Assumed Liabilities; and

WHEREAS, following the transfer of the Subject Assets to the Company and the assumption by the Company of the Assumed Liabilities, Seller desires to sell and transfer to Buyer, and Buyer desires to purchase from Seller, all of the Membership Interests upon the terms and subject to the conditions set forth herein.

NOW, THEREFORE, in consideration of the mutual covenants and agreements hereinafter set forth, the Parties agree as follows:

ARTICLE 1 CERTAIN DEFINITIONS

1.1 Certain Defined Terms

Unless the context otherwise requires, the respective terms defined in <u>Appendix A</u> attached hereto shall, when used herein, have the respective meanings therein specified, with each such definition to be equally applicable both to the singular and the plural forms of the term so defined.

1.2 References, Gender and Number

All references in this Agreement to an "Article," "Section," or "subsection" shall be to an Article, Section, or subsection of this Agreement, unless the context requires otherwise. Unless the context requires otherwise, the words "this Agreement," "hereof," "herein," "hereby," or words of similar import shall refer to this Agreement as a whole and not to a particular Article, Section, subsection, clause or other subdivision hereof. Whenever the context requires, the words used herein shall include the masculine, feminine and neuter gender, and the singular and the plural.

ARTICLE 2 PURCHASE AND SALE OF MEMBERSHIP INTERESTS; CONSIDERATION AND PAYMENT

2.1 Purchase and Sale of the Membership Interests

Upon and subject to the terms and conditions set forth in this Agreement, at the Closing, Seller shall sell, transfer and deliver to Buyer, free and clear of all Liens (other than restrictions on the transfer of securities arising under applicable federal and state securities Laws), and Buyer shall purchase from Seller, all right, title and interest in and to the Membership Interests.

2.2 Consideration

The consideration (the "**Purchase Price**") for the sale of the Membership Interests to Buyer shall be Three Hundred Thirty Million, Four Hundred Eighty Thousand Dollars and No/100 (\$330,480,000.00). The "**Adjusted Purchase Price**" shall be the Purchase Price as adjusted under **Section 2.5**.

2.3 Escrow

Contemporaneously with the execution of this Agreement, (a) Buyer, Seller and Wells Fargo Bank, National Association (the " **Escrow Agent**") entered into the Deposit Escrow Agreement in the form attached hereto as **Exhibit B** (the "**Deposit Escrow Agreement**"), and (b) Buyer deposited an amount equal to seven and one-half percent (7.5%) of the Purchase Price with the Escrow Agent into an account (the "**Escrow Account**") as a deposit hereunder (such amount, together with all interest earned thereon, the "**Cash Deposit**"). The Cash Deposit shall be held and distributed by the Escrow Agent in accordance with the terms of this Agreement and the Deposit Escrow Agreement.

2.4 Payment of Closing Consideration

At the Closing, (a) Buyer shall wire transfer the Adjusted Purchase Price minus the Cash Deposit and any amounts to be deposited with the Escrow Agent pursuant to **Section 5.5(a)** and **Section 6.8(a)** (such amount, the "**Closing Amount**") in immediately available funds to one or more accounts specified by Seller, and (b) Buyer and Seller shall provide written instructions to the Escrow Agent directing the Escrow Agent to disburse the Cash Deposit and any interest earned thereon pursuant to the Deposit Escrow Agreement to Seller.

2.5 Purchase Price Adjustment

(a) The Purchase Price shall be adjusted according to this **Section 2.5**. Seller shall deliver to Buyer not less than three (3) Business Days immediately preceding the Closing Date a statement (the "**Preliminary Settlement Statement**") setting forth Seller's preliminary good faith determination of the Adjusted Purchase Price, together with reasonably supporting documentation of the same. The Preliminary Settlement Statement shall set forth the Adjusted Purchase Price as provided in this **Section 2.5** using the best information available at the date Seller delivers the Preliminary Settlement Statement. In calculating the various adjustments to the Purchase Price pursuant to this **Section 2.5**, no single item shall be given duplicative effect. After delivery of the Preliminary Settlement Statement by Seller, Buyer will have an opportunity to review such Preliminary Settlement Statement, and Buyer and Seller shall work together in good faith to address any reasonable requests for adjustments to the Preliminary Settlement Statement made by Buyer on or before the Closing Date; *provided* that if Buyer and Seller are unable to resolve any disagreement regarding the Preliminary

Settlement Statement prior to Closing, the Preliminary Settlement Statement delivered by Seller shall be used for purposes of calculating the Closing Amount.

- For purposes of determining the amounts of the adjustments to the Purchase Price provided for in Section 2.5, the principles set forth in this (b) Section 2.5(b) shall apply. Buyer shall be entitled to all production of hydrocarbons from or attributable to the Subject Assets at and after the Effective Time (and all products and proceeds attributable thereto), and to all other income, proceeds, receipts and credits earned with respect to the Subject Assets at or after the Effective Time, and shall be responsible for (and entitled to any refunds with respect to) all Property Expenses (as defined in Section 2.5(c)) incurred at and after the Effective Time. Seller shall be entitled to all hydrocarbon production from or attributable to Subject Assets prior to the Effective Time (and all products and proceeds attributable thereto), and to all other income, proceeds, receipts and credits earned with respect to the Subject Assets prior to the Effective Time, and shall be responsible for (and entitled to any refunds with respect to) all Property Expenses incurred prior to the Effective Time. "Earned" and "incurred", as used in the Agreement shall be interpreted in accordance with GAAP and Council of Petroleum Accountants Society standards, except as otherwise specified herein. For purposes of allocating production (and proceeds and accounts receivable with respect thereto), under this Section 2.5, (i) liquid hydrocarbons shall be deemed to be "from or attributable to" the Subject Assets when they pass through the pipeline connecting into the storage facilities into which they are run and (ii) gaseous hydrocarbons shall be deemed to be "from or attributable to" the Subject Assets when they pass through the royalty measurement meters, delivery point sales meters or custody transfer meters on the gathering lines or pipelines through which they are transported (whichever meter is closest to the well). Seller shall utilize reasonable interpolative procedures, consistent with industry practice, to arrive at an allocation of production when exact meter readings or gauging and strapping data are not available. As part of the Preliminary Settlement Statement, Seller shall provide to Buyer such data as is reasonably necessary to support any estimated allocation, for purposes of establishing the Adjusted Purchase Price.
- (c) The term "Property Expenses" means all (i) direct costs and expenses (including capital expenditures and lease operating expenses) of ownership, exploration, maintenance, development, production and operation of the Subject Assets or acquiring new oil and gas leases after the Effective Time but on or before the Execution Date, which are within the Bakken Drillsite Spacing Units and listed on Schedule 2.5(c) (including, for this purpose, prepayments of any such costs or expenses), which costs and expenses shall include (x) Asset Taxes, but exclude any other Taxes and also exclude any costs and expenses of operating the office of Seller in Denver, Colorado, or the costs to the Company or its Affiliates of employees in that office; and (y) with respect to the Subject Assets operated by the Company or its Affiliates, all direct costs and expenses (including capital expenditures, lease operating expenses, facilities and plant expenses, and joint interest billings) chargeable to the joint account under the applicable Contract or in accordance with any Law of any Governmental Authority in connection with the exploration, maintenance, development, production and operation of such Subject Assets, and (ii) all overhead charges paid by the Seller, the Company or any of their Affiliates before the Closing Time to any third-party operator of any of the Subject Assets and incurred with respect to the time period (the "Adjustment Period") beginning at the Effective Time and ending at 7:00 A.M. (local time) on the Closing Date (such time being the "Closing Time").
- (d) <u>Upward Adjustments</u>. To calculate the Adjusted Purchase Price, the Purchase Price shall be adjusted upward by the following:

- (i) Proceeds received by Buyer (less any royalties and other non-Tax burdens and Asset Taxes measured by or payable out of such production to the extent not included in Property Expenses), from the sale of any hydrocarbons or otherwise that were produced from or attributable to the Subject Assets prior to the Effective Time.
- (ii) An amount equal to all Property Expenses attributable to the Subject Assets during the Adjustment Period that were paid by the Company, Seller or any of their Affiliates.
- (iii) An amount equal to all amounts paid by Seller or any of its Affiliates as a performance bond or security deposit outstanding at the Closing Date and attributable to the Subject Assets.
- (iv) An amount equal to, with respect to the Subject Assets operated by the Seller, (i) if there is an existing operating agreement in place for such Subject Assets, the overhead chargeable to the joint account pursuant to each of the existing operating agreements for such Subject Assets and incurred with respect to the Adjustment Period, and (ii) if, during the Adjustment Period, there is no existing operating agreement in place for such Subject Assets, an overhead amount equal to the Seller's Working Interest portion of \$12,000 per Well per month undergoing drilling or completion operations and \$1,200 per producing Well per month, and for which, in the case of clause (i) or (ii), neither the Company, the Seller nor any of their Affiliates have received reimbursement before the Closing from the other parties in the case of clause (i) bound by any such operating agreement or in the case of clause (ii) in the unit for such Well.
- (v) Any other amount provided for in this Agreement as payable by Buyer to Seller or as may otherwise be agreed to by Buyer and Seller.
- (e) <u>Downward Adjustments</u>. To calculate the Adjusted Purchase Price, the Purchase Price shall be adjusted downward by the following:
 - An amount equal to the Title Defect Adjustment, if any.
 - (ii) An amount equal to the Environmental Defect Adjustment, if any.
 - (iii) An amount equal to the Excluded Environmental Adjustment, if any.
 - (iv) Gross proceeds (less any royalties and other non-Tax burdens and Asset Taxes measured by or payable out of such production to the extent not included in Property Expenses) received by the Seller, the Company or their Affiliates (A) from the sale or disposition of hydrocarbons produced from the Subject Assets during the Adjustment Period, (B) from the rental, sale, salvage, or other disposition of any other Subject Assets on or after the Effective Time or (C) otherwise attributable to the ownership of the Subject Assets during the Adjustment Period (excluding, however, all Property Expenses for which the Company or any of its Affiliates receives reimbursement before the Closing Time from the other parties bound by any such operating agreement).
 - (v) If applicable, the amount provided in <u>Schedule 2.5(e)</u> (v).
 - (vi) Any other amount provided for in this Agreement as payable by Seller to Buyer or as may otherwise be agreed to by Buyer and Seller.

2.6 Post Closing Review

After the Closing, Seller shall review the Preliminary Settlement Statement and determine the actual Adjusted Purchase Price pursuant to Section 2.5. On or prior to the ninetieth (90th) day after the Closing Date, Seller shall present Buyer with a statement of the actual Purchase Price adjustments and such supporting documentation as is reasonably necessary to support the Adjusted Purchase Price shown therein (the "Final Settlement Statement"). Buyer will give representatives of Seller or its Affiliates reasonable access to its premises and to the books and records of Buyer and the Company for purposes of preparing the Final Settlement Statement and will cause appropriate personnel of Buyer and, if applicable, the Company to assist Seller, its Affiliates, and its and their representatives, at no cost to Seller, in the preparation of the Final Settlement Statement. Seller will, and will cause its Affiliates to, give representatives of Buyer and, if applicable, the Company, reasonable access to its and their premises and to its and their books and records for purposes of reviewing the calculation of the Adjusted Purchase Price set forth in the Final Settlement Statement and will cause appropriate personnel of Seller and its Affiliates to assist Buyer and its representatives, at no cost to Buyer or the Company, in verification of such calculation. The Final Settlement Statement shall become final and binding on Seller and Buyer sixty (60) days following the date the Final Settlement Statement is received by Buyer, except to the extent that prior to the expiration of such sixty (60) day period, Buyer shall deliver to Seller notice, as hereinafter required, of its disagreement with the contents of the Final Settlement Statement. Such notice shall be in writing and set forth all of Buyer's disagreements with respect to any portion of the Final Settlement Statement, together with Buyer's proposed changes thereto, and shall include an explanation in reasonable detail of, and such supporting documentation as is reasonably necessary to support, such changes. If Buyer has timely delivered such a notice of disagreement to Seller, then, upon written agreement between Buyer and Seller resolving all disagreements of Buyer set forth in such notice, the Final Settlement Statement will become final and binding upon Buyer and Seller. If the Final Settlement Statement has not become final and binding by the ninetieth (90th) day following its receipt by Buyer, then Buyer or Seller may submit any unresolved disagreements of Buyer set forth in such notice to the Houston, Texas office of KPMG LLP (the "Independent Accounting Firm") who shall be designated to act as an arbitrator and to decide all points of disagreement with respect to the Final Settlement Statement, such decision to be binding upon Buyer and Seller. Buyer and Seller shall use commercially reasonable efforts to cause such arbitrator to render its decision within fifteen (15) Business Days after such points of disagreement are submitted to such arbitrator. The costs and expenses of such arbitrator shall be borne equally by Seller and Buyer. Upon resolution of such unresolved disagreements, the Final Settlement Statement shall be final and binding upon the Parties. The date upon which such agreement is reached or upon which the final Adjusted Purchase Price is established shall be herein called the "Final Settlement Date." If the Adjusted Purchase Price shown in the Final Settlement Statement is more than the Adjusted Purchase Price shown in the Preliminary Settlement Statement, Buyer shall pay to Seller the amount of such difference by wire transfer in immediately available funds no later than five (5) Business Days after the Final Settlement Date. If the Adjusted Purchase Price shown in the Final Settlement Statement is less than the Adjusted Purchase Price shown in the Preliminary Settlement Statement, Seller shall pay the amount of such difference, to Buyer by wire transfer in immediately available funds no later than five (5) Business Days after the Final Settlement Date.

ARTICLE 3 REPRESENTATIONS AND WARRANTIES

3.1 Representations and Warranties of Seller Regarding the Subject Assets and the Company

Seller represents and warrants to Buyer as follows:

- (a) Organization and Qualification. At Closing, the Company will be a limited liability company duly organized, validly existing and in good standing under the laws of the State of Delaware and will have the requisite limited liability company power to carry on its business as it will be conducted at the Closing. The Company will be duly qualified to do business and be in good standing in the State of North Dakota.
- (b) Membership Interests. At Closing, all of the Membership Interests will be duly authorized and validly issued, fully paid and non-assessable, and will be issued in accordance with the Organizational Documents of the Company. The Membership Interests will constitute 100% of the total issued and outstanding membership interests or other equity interests in the Company. At Closing, there will be no outstanding or authorized options, warrants, convertible securities or other rights, agreements, arrangements or commitments of any character relating to the Membership Interests or obligating the Company to issue or sell any Membership Interests or any other interest in, the Company. There will be no voting trusts, member agreements, proxies or other agreements or understandings in effect with respect to the voting or transfer of any of the Membership Interests. The Seller will be the sole record and beneficial owner of the Membership Interests or the other equity interests in the Company and, immediately upon consummation of the transactions contemplated by this Agreement, Buyer will own all of the Membership Interests free and clear of all Liens (other than restrictions on transfer of securities arising under applicable federal and state securities Laws).
- (c) <u>Subsidiaries</u>. From the date of its formation through the Closing, the Company will not own, either directly or indirectly, any equity interest in any Person.
- (d) Actions. Except as set forth on Schedule 3.1(d), there are no Actions ongoing or, to the Knowledge of Seller, pending or threatened against the Company, Seller or any Affiliate of Seller with respect to the Subject Assets, or which relate to the transactions contemplated by this Agreement, and which, if determined adversely, would affect Seller's ability to consummate the transactions contemplated by this Agreement or have a Material Adverse Effect. Except as set forth on Schedule 3.1(d), there is no outstanding order, writ, judgment, injunction, decree, stipulation, determination or award entered by or with any Governmental Authority and no unsatisfied judgments, penalties or awards issued by any Governmental Authority against or affecting the Company or any of the Subject Assets.
- (e) Compliance with Laws; Permits. Except with respect to (i) the matters set forth on Schedule 3.1(d) and Schedule 3.1(e), (ii) compliance with Laws concerning Taxes (as to which representations and warranties are made solely pursuant to Section 3.1(i)) and (iii) compliance with Environmental Laws (as to which representations and warranties are made solely pursuant to Section 3.1(l)), the Subject Assets operated by the Seller, and, to the Knowledge of Seller, the operation of each of the other Subject Assets are, in each case, in material compliance with all applicable Laws. The Seller has all approvals, authorizations, consents, licenses, orders, franchises, rights, registrations and permits of all Governmental Authorities (the "Permits") that are material to the Seller's operatorship of the Subject Assets as presently operated. Except as would not reasonably be expected to have a Material Adverse Effect, (x) each Permit is in full force and effect and has been duly and validly issued, (y) there are no uncured violations of the terms and provisions of such Permits and (z) except as set forth on Schedule 3.1(e), with respect to each such Permit, the Company has not received written notice from any Governmental Authority of any violation by the Company of such Permit that remains uncured. To the Knowledge of Seller, Seller is not in violation of the terms of the Permits in any material respect.
- (f) <u>Preference Rights</u>. There are no Preference Rights with respect to the Subject Assets.

- (g) <u>Transfer Requirements</u>. Except as set forth on <u>Schedule 3.1(g)</u>, and to the Knowledge of Seller, there are no agreements containing a Transfer Requirement that is applicable to the transactions contemplated by this Agreement.
- (h) Material Contracts. Schedule 3.1(h)(1) sets forth a list of all Material Contracts to which the Seller or any of its Affiliates is a party as of the Execution Date. True and complete copies of all Material Contracts, and all amendments thereto, have been provided to Buyer. Each Material Contract is valid and binding on the Seller or its Affiliates, as applicable, in accordance with its terms and is in full force and effect. Except as set forth on Schedule 3.1(h)(2), Seller is not in material breach of, or material default under, and to the Knowledge of Seller, no other Person is in material breach of, or material default under, any Material Contract, and there does not exist under any provision thereof or, to the Knowledge of Seller, any event that, with the giving of notice or the lapse of time or both, would constitute such a material breach or material default by any Person.

(i) <u>Tax</u> Matters.

- (i) Except with respect to items set forth on Schedule 3.1(i), (A) during the period of the Seller's and the Company's ownership of the Subject Assets, all Asset Taxes and Transfer Taxes (other than Transfer Taxes resulting from the transactions contemplated by this Agreement) that have become due and payable before the Effective Time have been properly paid, whether or not shown on any Tax Return), (B) there are no suits or proceedings ongoing, or to the Knowledge of the Seller, any claims, investigations, audits, inquiries pending or threatened against the Seller or the Company in respect of Taxes of the Company or Asset Taxes related to the Subject Assets, and (C) none of the Subject Assets are subject to any partnership within the meaning of subchapter K of Chapter 1 of Subtitle A of the Code. The Company is, and has been throughout its existence, disregarded as an entity separate from its owner for United States federal Income Tax purposes.
- (ii) The Company has not been required to file any Tax Returns under applicable law and regulations. No claim has ever been made by an authority in a jurisdiction where the Company does not file Tax Returns that the Company or any of the Subject Assets is or may be subject to taxation by that jurisdiction. There are no Liens for Taxes (other than Taxes not yet due and payable) upon any of the Subject Assets.
- (iii) The Company has withheld and paid all Taxes required to have been withheld and paid in connection with any amounts paid or owing to any employee, independent contractor, creditor, member or other third party.
- (iv) Neither the Company nor the Seller has received from any federal, state, local or non-US Taxing Authority (including jurisdictions where the Company has not filed Tax Returns) any (i) written notice indicating an intent to open an audit or other review with respect to Taxes of the Company or Asset Taxes or Transfer Taxes that relate to the Subject Assets, (ii) request for information related to Taxes of the Company or Asset Taxes or Transfer Taxes that relate to the Subject Assets, or (iii) notice of deficiency or proposed adjustment for any amount of Tax proposed, asserted, or assessed by any Taxing Authority against the Company. Schedule 3.1(i) lists all Tax Returns filed with respect to Asset Taxes or Transfer Taxes related to the Subject Assets for taxable periods ended on or after December 31, 2011, indicates those Tax Returns that have been audited, and indicates those Tax Returns that currently are the subject of audit. The Seller has delivered to Buyer correct and complete

- copies of all federal income Tax Returns, examination reports and statements of deficiencies assessed against or agreed to by the Company or the Seller with respect to the Company or the Subject Assets filed or received since December 31, 2011.
- (v) Neither the Company nor the Seller has waived any statute of limitations in respect of Taxes from or attributable to the Company or the Subject Assets or agreed to any extension of time with respect to a Tax assessment or deficiency against the Company or related to the Subject Assets.
- (vi) The Company is not a party to or bound by any Tax allocation or sharing agreement. The Company does not have any Liability for the Taxes of any Person (other than the Company) under 26 C.F.R. Reg 1.1502-6 (or any similar provision of state, local or non-U.S. law) as a transferee or successor, by contract, or otherwise.

Notwithstanding anything to the contrary set forth in this Agreement, this **Section 3.1(i)** contains the sole and exclusive representations and warranties of Seller relating to Taxes.

- (j) Wells. There are no Wells in respect of which Seller or any of its Affiliates have received an unresolved order from any Governmental Authority requiring that such Wells be plugged and abandoned. All plugged and abandoned Wells that have been plugged and abandoned by Seller have been plugged and abandoned in material compliance with applicable Laws. Except for the Wells described on **Part IV** of **Exhibit A**, Seller and its Affiliates have no duty to plug any wells located on the Lands. To Seller's Knowledge, all Wells have been drilled and completed within the limits permitted by all applicable Leases and pooling or unit agreements or orders.
- (k) <u>Proposed Operations or Expenditures</u>. Except as set forth on <u>Schedule 3.1(k)</u>, as of the Execution Date, there are no outstanding authorizations for expenditure or other commitments to conduct any operations or expend any amount of money on or with respect to the Subject Assets which are binding on the Seller or the Subject Assets and will be binding on the Company after Closing and which Seller reasonably anticipates will require the expenditure of in excess of Two Hundred Fifty Thousand and No/100 Dollars (\$250,000.00) per item (net to the Seller's interest).

(l) Environmental.

- (i) Except as set forth on Schedule 3.1(1) and except as would not reasonably be expected to have a Material Adverse Effect:
 - (A) the Subject Assets operated by the Seller or any of its Affiliates and all other Subject Assets comprising the Business are currently in compliance with Environmental Laws;
 - (B) all Permits required by Environmental Laws (the "Environmental Permits") with respect to the ownership or operation of the Subject Assets operated by the Seller or any of its Affiliates, and all other Subject Assets comprising the Business, are in full force and effect, and the Subject Assets operated by the Seller or any of its Affiliates and, to the Knowledge of Seller, the Subject Assets operated by any third party are in compliance with the Environmental Permits;
 - (C) there has been no Release or, to the Knowledge of Seller, threatened Release of Hazardous Materials in connection with the Subject Assets operated by the Seller

- or any of its Affiliates or any other Subject Assets comprising the Business (including, in each case, any Release or threatened Release at any offsite location where the Seller or its Affiliates transported or disposed, or arranged for the transport or disposal, of Hazardous Materials) that is reasonably likely to give rise to a material liability to the Company or the Seller or its Affiliates; and
- (D) there are no Actions pending or, to the Knowledge of Seller, threatened against the Company or Seller with respect to the Business by or before a Governmental Authority pursuant to any Environmental Laws or relating to an Environmental Permit.
- (ii) As of the date hereof, neither Seller nor any of its Affiliates have received any unresolved written notice from any Governmental Authority of any release, disposal, event, condition, circumstance, activity, practice or incident concerning any Subject Assets that alleges non-compliance by Seller or its Affiliates with any Environmental Law.

The representations and warranties made in this **Section 3.1(I)** are the only representations and warranties of Seller with respect to Environmental Laws, Hazardous Materials, Remediation and other environmental matters.

- (m) Special Warranty. Neither the Seller nor any Person claiming by, through or under the Seller, but not otherwise, has taken any action that would result in the Company failing to own Defensible Title following the Closing.
- (n) Payments for Production. Except as set forth in Schedule 3.1(n) and except for amounts being held in suspense as permitted pursuant to applicable Law, all rentals, royalties, excess royalty, overriding royalty interests, hydrocarbon production payments and all other payments due and/or payable by Seller or the Company or their Affiliates to overriding royalty holders and other interest owners with respect to the Subject Assets and the hydrocarbons produced therefrom prior to the Effective Time have been properly and timely paid in all material respects.
- (o) <u>Imbalances</u>. Except for normal immaterial pipeline imbalances that are adjusted by the pipeline each month, as of the date hereof, there are no Imbalances or other imbalances attributable to the Wells which require payment from Buyer to any Person or for which Buyer would otherwise be responsible.
- (p) <u>Calls on Production</u>. Except as described in <u>Schedule 3.1(p)</u> or <u>Schedule 3.1(h)(1)</u>, to Seller's Knowledge, the Subject Assets are not subject to any calls on production.
- (q) <u>Hedge Agreements</u>. Neither the Subject Assets nor the production therefrom are subject to any Hedge Agreements with respect to which Buyer or the Company will have any obligations.
- (r) Operatorship. As of the date hereof, (i) neither Seller nor any of its Affiliates have received written notice pursuant to and in accordance with the terms of any operating agreement under which the Seller or any of its Affiliates is currently designated as the operator or, to the Knowledge of Seller, any other notice, with respect to any pending or requested vote to remove the Seller or any of its Affiliates as the named "operator" of any of the Subject Assets for which the Seller or any of its Affiliates is currently designated as "operator", and (ii) the Seller and any of its applicable Affiliates have not failed or refused to carry out, and at all times during the two years prior to the Execution Date have not failed or refused to carry out, their duties as "operator" under the terms of the applicable joint operating agreements that cover the Subject Assets and under which the Seller or any of its

- Affiliates is currently, or at any time during the two years prior to the date hereof has been, designated as the named "operator" such that there exists cause for removal of Seller or any of its Affiliates as the operator.
- (s) Non-consent Elections. Except as set forth in Schedule 3.1(s), as of the date hereof no operations are being conducted or, since the Effective Time, have been conducted on, or with respect to, the Subject Assets with respect to which the Seller has elected to be a non-consenting party under the applicable operating agreement or any forced pooling and with respect to which all of the Seller's rights have not yet reverted to such Person.
- (t) <u>Bank Accounts; Power of Attorney</u>. The Company will hold no bank accounts or investment accounts. There are no powers of attorney issued by the Company in effect.
- (u) Advance Payments. Except as set forth on Schedule 3.1(u), the Seller is not obligated by virtue of any take-or-pay payment, advance payment or other similar payment (other than royalties, overriding royalties and similar arrangements reflected with respect to the Net Revenue Interests for the Wells set forth in Part II of Exhibit A and gas balancing arrangements), to deliver hydrocarbons attributable to the Subject Assets, or proceeds from the sale thereof, at some future time without receiving payment therefore at or after the time of delivery.
- (v) <u>Payout Balances.</u> <u>Schedule 3.1(v)</u> identifies the Seller-operated Wells subject to a reversion or other adjustment at some level of cost recovery or payout (or passage of time or other event other than a termination of a Lease by its terms), and the list of the "payout" balance has been maintained by the Seller in the ordinary course of business of each such Seller-operated Wells, as of the Effective Time.
- (w) Operations of the Company. The Company will be formed solely for the purpose of engaging in the transactions contemplated hereby. Prior to the Closing, the Company will not undertake any activities other than as contemplated by this Agreement and the Seller will not cause the Company to assume any liabilities other than as provided in the General Conveyance and Assumption Agreement.
- (x) Employee Matters. At no time has the Company had employees, independent contractors or consultants. At no time has the Company had any pension, benefit, retirement, compensation, employment, consulting, profit-sharing, deferred compensation, incentive, bonus, performance award, phantom equity or other equity, change in control, retention, severance, vacation, paid time off, welfare, fringe-benefit and other similar agreement, plan, policy, program or arrangement, in each case whether or not reduced to writing and whether funded or unfunded.
- (y) <u>Sufficiency of Subject Assets</u>. The Seller has, and the Subject Assets include, all material surface rights, permits, consents and other assets and rights necessary to access, operate, maintain and repair the Wells and Equipment in the ordinary course of business as currently conducted by Seller.
- (z) <u>Insurance</u>. All insurance policies of the Seller or any of its Affiliates relating to the Subject Assets are in full force and effect and insure the Subject Assets in reasonably sufficient amounts against normal risks usually insured against by persons operating oil and gas properties in the same area.
- (aa) No Other Representations. Except for the representations and warranties contained in this Section 3.1 (including the related portions of the disclosure schedules), the Seller makes no express or implied representation or warranty, either written or oral, on behalf of Seller or the Company including any representation or warranty as to the accuracy or completeness of any information regarding the Subject Assets furnished or made available to Buyer and its representatives. Except

as provided in **Section 3.1(r)**, Seller makes no representations or warranties to Buyer as to the continued operatorship after the date of the Asset and Liability Transfer of the Subject Assets which Seller or its Affiliates currently operate.

3.2 Representations and Warranties of Seller Regarding Seller

Seller represents and warrants to Buyer as follows:

- (a) <u>Organization and Qualification</u>. Seller is a limited liability company duly organized, validly existing and in good standing under the laws of the State of Colorado.
- (b) <u>Authority</u>. Seller has all requisite limited liability company power and authority to execute and deliver this Agreement and each of the agreements required to be executed and delivered by Seller under this Agreement and to perform its obligations hereunder and thereunder and to consummate the transactions contemplated hereby and thereby. The execution, delivery and performance of this Agreement and the agreements to be executed and delivered by Seller hereunder and the transactions contemplated hereby and thereby have been duly and validly authorized by all requisite limited liability company action on the part of Seller.
- (c) Enforceability. This Agreement has been duly executed and delivered by Seller, and (assuming due authorization, execution, and delivery by Buyer) constitutes, and at Closing each of the agreements required to be executed and delivered by Seller hereunder (assuming due authorization, execution, and delivery by Buyer) will constitute, a legal, valid and binding agreement of Seller enforceable against Seller in accordance with its terms, subject to (i) applicable bankruptcy, insolvency, reorganization, moratorium and other similar laws of general application with respect to creditors, (ii) general principles of equity and (iii) the power of a court to deny enforcement of remedies generally based upon public policy.
- (d) <u>Title to Membership Interests</u>. At Closing, Seller will be the record and beneficial owner of and have good and valid title to the Membership Interests, free and clear of any Liens, except as are imposed by applicable securities Laws or such Liens that will be released in full prior to the Closing.
- (e) No Conflict. Except as set forth on Schedule 3.2(e), neither the execution and delivery of this Agreement or the agreements required to be executed and delivered by Seller hereunder nor the consummation of the transactions and performance of the terms and conditions contemplated hereby or thereby by Seller will (i) violate, conflict with or constitute a default or an event that, with notice or lapse of time or both, would be a default, breach or violation under any term or provision of the certificate of incorporation, bylaws or other governing documents of Seller or the Company; (ii) materially violate, conflict in any material respect with or constitute a material default or event that, with notice or lapse of time or both, would be a material default, breach or violation under any term or provision of any commitment, license, promissory note, conditional sales contract, indenture, Lien, lease or other agreement, instrument or arrangement to which Seller or the Company is a party or by which Seller or the Company or any of the Subject Assets is bound; (iii) violate, conflict with or constitute a breach of any Law in any material respect applicable to Seller or the Company or by which Seller or the Company or any of the Subject Assets is bound; or (iv) result in the creation or imposition of any adverse material claim or interest, or any Lien, charge, equity or restriction of any material nature whatsoever, on or affecting Seller or the Company or any of the Subject Assets.

- (f) Consents. Except with respect to (i) any Transfer Requirements set forth on Schedule 3.1(g), (ii) such filings with the SEC as may be required by applicable Law or pursuant to regulation of any applicable securities exchange in the United States of America or Canada, and (iii) such matters set forth on Schedule 3.2(f), no consent, approval, authorization or permit of, or filing with or notification to, any Person is required for or in connection with the execution and delivery of this Agreement by Seller or for or in connection with the consummation of the transactions and performance of the terms and conditions contemplated hereby and thereby by Seller.
- (g) Actions. There are no Actions ongoing or, to the Knowledge of Seller, pending or threatened against Seller which relate to the transactions contemplated by this Agreement other than Actions which would not if determined adversely, materially affect Seller's ability to consummate the transactions contemplated by this Agreement or to perform its obligations hereunder.
- (h) <u>Foreign Person</u>. Seller is not a foreign person (as that term is used in Section 1.1445-2 of the Treasury Regulations).
- (i) <u>Brokerage Fees and Commissions</u>. Neither Seller nor any Affiliate of Seller has incurred an obligation or entered into any agreement for any investment banking, brokerage or finder's fee or commission in respect of the transactions contemplated by this Agreement for which Buyer or, after the Closing, the Company, shall incur any liability.
- (j) <u>Bankruptcy</u>. There are no bankruptcy, reorganization, receivership or arrangement proceedings instituted or, to the Knowledge of Seller, pending or threatened in writing against the Seller. The Seller is not insolvent or generally unable to pay its debts as they become due.

3.3 Representations and Warranties of Buyer

Buyer represents and warrants to Seller as follows:

- (a) Organization and Qualification. Buyer is a corporation duly formed, validly existing and in good standing under the laws of Delaware and has the requisite corporate power to carry on its business as it is now being conducted. Buyer is duly qualified to do business and is in good standing in each jurisdiction in which the Subject Assets make such qualification necessary.
- (b) Authority. Buyer has all requisite corporate power and authority to execute and deliver this Agreement and each of the agreements required to be executed and delivered by Buyer under this Agreement and to perform its obligations hereunder and thereunder. The execution, delivery and performance of this Agreement and the agreements to be executed and delivered by Buyer hereunder and the transactions contemplated hereby and thereby have been duly and validly authorized by all requisite corporate action on the part of Buyer.
- (c) Enforceability. This Agreement has been duly executed and delivered by Buyer, and (assuming due authorization, execution and delivery by Seller) constitutes, and at Closing each of the agreements required to be executed and delivered by Buyer hereunder (assuming due authorization, execution and delivery by the other parties thereto) will constitute, a valid and binding agreement of Buyer enforceable against Buyer in accordance with its terms, subject to (i) applicable bankruptcy, insolvency, reorganization, moratorium and other similar laws of general application with respect to creditors, (ii) general principles of equity and (iii) the power of a court to deny enforcement of remedies generally based upon public policy.

- (d) No Conflict. Neither the execution and delivery of this Agreement or the agreements required to be executed and delivered by Buyer hereunder nor the consummation of the transactions and performance of the terms and conditions contemplated hereby or thereby by Buyer will (i) violate, conflict with or constitute a default or an event that, with notice or lapse of time or both, would be a default, breach or violation under any term or provision of the Certificate of Incorporation or other governing documents of Buyer, (ii) materially violate, conflict in any material respect with or constitute a material default or event that with notice or lapse of time or both, would be a material default, breach or violation under any term or provision of any commitment, license, promissory note, conditional sales contract, indenture, mortgage, deed of trust, lease or other agreement, instrument or arrangement to which Buyer is a party or by which Buyer is bound; (iii) violate, conflict with or constitute a breach of any Law in any material respect applicable to Buyer or by which Buyer is bound; or (iv) result in the creation or imposition of any adverse material claim or interest, or any Lien, charge, equity or restriction of any material nature whatsoever, on or affecting Buyer.
- (e) <u>Consents.</u> Except with respect to such filings with the SEC as may be required by applicable Law or pursuant to regulation of any applicable securities exchange in the United States of America, no consent, approval, authorization or permit of, or filing with or notification to, any Person is required for or in connection with the execution and delivery of this Agreement by Buyer or for or in connection with the consummation of the transactions and performance of the terms and conditions contemplated hereby by Buyer.
- (f) <u>Actions</u>. There are no Actions ongoing or, to the Knowledge of Buyer, pending or threatened against Buyer which relate to the transactions contemplated by this Agreement other than Actions which are not reasonably expected by Buyer to materially impair the ability of Buyer to consummate the transactions contemplated by this Agreement or to perform its obligations hereunder.
- (g) <u>Brokerage Fees and Commissions</u>. Neither Buyer nor any Affiliate of Buyer has incurred an obligation or entered into any agreement for any investment banking, brokerage or finder's fee or commission in respect of the transactions contemplated by this Agreement for which Seller shall incur any liability.
- (h) <u>Funds</u>. Buyer has sufficient funds or sufficient capacity to borrow under its existing revolving credit agreement available to enable Buyer to consummate the transactions contemplated hereby and to pay all related fees and expenses of Buyer.
- (i) Securities Laws. Buyer is an "accredited investor," as such term is defined in Regulation D of the Securities Act and will acquire the Membership Interests for its own account and not with a view to a sale or distribution thereof in violation of the Securities Act and the rules and regulations thereunder, any state blue sky Laws or any other securities Laws. Buyer is acquiring the Membership Interests solely for its own account for investment purposes and not with a view to, or for offer or sale in connection with, any distribution thereof. Buyer acknowledges that the Membership Interests are not registered under the Securities Act or any state securities Laws, and that the Membership Interests may not be transferred or sold except pursuant to the registration provisions of the Securities Act or pursuant to an applicable exemption therefrom and subject to state securities Laws. Buyer is able to bear the economic risk of holding the Membership Interests for an indefinite period (including total loss of its investment), and has sufficient knowledge and experience in financial and business matters so as to be capable of evaluating the merits and risk of its investment.

(j) <u>Independent Evaluation</u>. Buyer is an experienced and knowledgeable investor in the oil and gas business. Prior to entering into this Agreement, Buyer was advised by its legal, Tax, and other professional counsel and such other Persons it has deemed appropriate concerning this Agreement and the transactions contemplated by this Agreement and has relied solely on (i) its own independent valuation of the Company and the Subject Assets including the geologic and geophysical characteristics of the Subject Interests, the estimated reserves recoverable therefrom, and the price and expense assumptions applicable thereto, and (ii) the representations and warranties made by Seller in this Agreement.

ARTICLE 4 INVESTIGATION OF SUBJECT ASSETS; CONFIDENTIALITY

4.1 Investigation of Subject Assets

Promptly following the execution of this Agreement and until the Defect Notice Date (or earlier termination of this Agreement), Seller (a) shall permit Buyer and its representatives at reasonable times to examine, in Seller's offices, all Records including abstracts of title, title opinions, title files, ownership maps, lease files, assignments, division orders, and similar documents relating to the Subject Assets insofar as the same are in the possession or control of Seller or any Affiliate of Seller and insofar as Seller or any Affiliate of Seller may do so without (i) violating any legal constraint or confidentiality obligation to a third Person (provided that Seller shall use all commercially reasonable efforts to obtain any necessary waivers of such obligations to allow disclosure of any Records) or (ii) waiving any attorney/client privilege and (b) subject to any required consent of any third Person (provided that Seller shall use all commercially reasonable efforts to obtain any such consents), shall permit Buyer and its representatives at reasonable times and at Buyer's sole risk, cost and expense, to conduct, in the presence of Seller's representatives, reasonable inspections of the Subject Assets: provided that Buyer shall not conduct any Phase II environmental testing or other invasive surface or subsurface testing without Seller's prior written consent, which may not be unreasonably withheld, conditioned or delayed; provided, further, Buyer does hereby indemnify and hold harmless the Seller Indemnified Parties from and against any and all Covered Liabilities caused by Buyer's inspection of the Subject Assets, REGARDLESS OF ANY CONCURRENT NEGLIGENCE OR STRICT LIABILITY ON THE PART OF ANY SELLER INDEMNIFIED PARTY AND REGARDLESS OF THE FORM OF CLAIM WHETHER AT COMMON LAW, STRICT LIABILITY, NEGLIGENCE OR UNDER ANY STATUTE OR REGULATION, EXCLUDING, HOWEVER, COVERED LIABILITIES RESULTING FROM AND TO THE EXTENT CAUSED BY THE GROSS NEGLIGENCE OR WILLFUL MISCONDUCT OF ANY SELLER INDEMNIFIED PARTY. Unless caused by Buyer's inspection of the Subject Assets, prior to the Closing, Buyer shall not be required to defend, indemnify and hold harmless the Seller Indemnified Parties pursuant to this Section 4.1 from and against any Covered Liabilities arising out of, resulting from or relating to any pre-existing Environmental Defect identified by or on behalf of Buyer as a result of any physical inspection, due diligence activities or access granted to Buyer and Buyer's representatives pursuant hereto; provided that if Closing occurs, this Section 4.1 shall not limit Buyer's assumption of the Assumed Liabilities and Seller's right to indemnification pursuant to Article 12.

4.2 Obligations Under Confidentiality Agreement

(k) Prior to the Closing, Buyer shall maintain all information made available to it pursuant to the Confidentiality Agreement or **Section 4.1** confidential and otherwise to comply in all respects with, and to cause their respective officers, employees, representatives, consultants and advisors to maintain all such information confidential and otherwise to comply in all respects with the provisions

regarding confidentiality in the Confidentiality Agreement; *provided, however*, that the Parties hereby acknowledge and agree that nothing in this **Section 4.2(a)** shall affect or limit any of the (i) other provisions set forth in the Confidentiality Agreement (including all provisions regarding non-solicitation and standstill) or (ii) generally applicable provisions set forth in the Confidentiality Agreement to the extent such provisions relate to such other provisions.

- (1) Without limiting Seller's right to issue a press release pursuant to Section 7.3, for a period of two (2) years from and after the Closing and except as otherwise required by any requirements of Law, Seller shall maintain all information relating to the Company with respect to the Subject Assets confidential and otherwise to comply in all respects with, and to cause its officers, employees, representatives, consultants and advisors to maintain all information relating to the Subject Assets confidential; provided, however, that the information subject to such confidentiality restrictions shall not include information that is or becomes generally available to the public other than as a result of disclosure by Seller or its Affiliates.
- (m) Without limiting Buyer's right to issue a press release pursuant to **Section 7.3**, for a period of two (2) years from and after the Closing and except as otherwise required by any requirements of Law, Buyer shall maintain all information relating to all Subject Assets retained by Seller (or an Affiliate designated by Seller) pursuant to **Section 6.7** confidential and otherwise to comply in all respects with, and to cause its officers, employees, representatives, consultants and advisors to maintain all information relating to such assets confidential; *provided, however*, that the information subject to such confidentiality restrictions shall not include information that is or becomes generally available to the public other than as a result of disclosure by Buyer or its Affiliates.

ARTICLE 5 TITLE ADJUSTMENTS

5.1 General Disclaimer; No Title Warranty or Representations

Without limiting Buyer's right to receive an adjustment to the Purchase Price by operation of Section 5.2, and except for the special warranty of Defensible Title contained in Section 3.1(m), Seller makes no warranty or representation, express, implied, statutory or otherwise, with respect to Seller's or the Company's title to any of the Subject Assets and Buyer hereby acknowledges and agrees that, except as provided above, Buyer's sole remedy for any defect of title, including any Title Defect, with respect to any of the Subject Assets shall be pursuant to the procedures set forth in this ARTICLE 5.

5.2 Buyer's Title Review

(f) <u>Buyer's Assertion of Title Defects</u>. Prior to 5:00 P.M., Mountain Time, on September 19, 2014 (the " **Defect Notice Date**"), Buyer shall have the right, but not the obligation, to notify Seller in writing of any matters which, in Buyer's reasonable opinion, constitute Title Defects and which Buyer intends to assert as a Title Defect with respect to any Property Subdivision pursuant to this **ARTICLE 5**. For all purposes of this Agreement, Buyer shall be deemed to have waived any Title Defect which Buyer fails to assert as a Title Defect by written notice given to Seller on or before 5:00 P.M., Mountain Time, on the Defect Notice Date. To be effective, Buyer's written notice of a Title Defect must by asserted in good faith and include (i) a description of the alleged Title Defect Property, (ii) the Allocated Value of the alleged Title Defect Property, (iii) a brief description of the matter constituting the asserted Title Defect and the basis for such Title Defect, (iv) the Title Defect Amount attributable thereto claimed in good faith by Buyer, (v) the computations for such Title Defect Amount, (vi) to the extent then known by Buyer, the necessary curative for each Title Defect, and

(vii) supporting documents reasonably necessary for Seller (as well as any title attorney or examiner hired by Seller) to verify the existence of such asserted Title Defect; *provided*, *however*, that to the extent that Seller and Buyer are in disagreement as to whether or not Buyer's written notice of Title Defect is effective as provided above, Seller and Buyer shall resolve such dispute pursuant to Section 5.5 of this Agreement. To give Seller an opportunity to commence reviewing and curing Title Defects, on a weekly basis, Buyer shall give Seller written notice of any Title Defect that Buyer determines exists, which notice may be preliminary in nature and supplemented prior to 5:00 P.M., Mountain Time, on the Defect Notice Date; *provided*, *however*, any failure of Buyer to deliver such notice shall not constitute a breach of this Agreement.

- (g) <u>Allocation of Consideration</u>. Solely for the purposes of **ARTICLE 5** and **ARTICLE 6**, the allocated value of each Property Subdivision shall be the value specified for such Property Subdivision in <u>Part II of Exhibit A</u> (the "Allocated Value").
- (h) <u>Seller's Opportunity to</u> Cure.
 - (i) Seller shall have until one (1) day before the Closing Date, at its sole cost and expense, if it so elects but without obligation, to cure all or a portion of such asserted Title Defects. Subject to Seller's continuing right to dispute the existence of a Title Defect and/or the Title Defect Amount asserted with respect thereto, if Seller within such time fails to cure any Title Defect of which Buyer has given a timely and effective written notice as required above and Buyer has not and does not waive same on or before the day immediately preceding the Closing Date, the Property Subdivision affected by such uncured and unwaived Title Defect shall be a "Title Defect Property".
 - (ii) To the extent that a Title Defect is based on Seller's failure to have Defensible Title to a Bakken Drillsite Spacing Unit, Seller may cure such defect by securing, at its sole cost and expense, in the Seller's name new or renewal leases with reasonable primary terms and royalty rates covering a sufficient number of net mineral acres within the Bakken Drillsite Spacing Unit to elevate Seller's Net Revenue in such unit to an amount not less than set forth on **Part II(A) of Exhibit A** for such unit.
 - (iii) In addition, Seller may attempt to cure any Title Defect at any time after Closing but prior to the deadline for submitting a Final Settlement Statement under **Section 2.6**. Upon curing of any such Title Defect, Seller may include in the Final Settlement Statement an amount due Seller arising from such title curative.
- (i) <u>Buyer's Title Adjustments</u>. Subject to (w) **Section 5.5**, (x) the special warranty of Defensible Title contained in **Section 3.1(m)**, (y) Buyer's termination right as provided in **Section 13.1(a)**, and (z) Seller's termination right as provided in **Section 13.2(b)**, as Buyer's sole and exclusive remedy with respect to Title Defects, Buyer shall be entitled to reduce the Purchase Price by an amount equal to the amount, if any, by which the aggregate amount of Title Defect Amounts with respect to all Title Defect Properties exceeds the Title Defect Deductible (such amount in excess of the Title Defect Deductible, the "**Title Defect Adjustment**"). "**Title Defect Amount**" shall mean, with respect to a Title Defect Property, the amount by which the value of such Title Defect Property is impaired as a result of the existence of one or more uncured and unwaived Title Defects, which amount shall be determined as follows:
 - (iii) If Buyer and Seller agree on the Title Defect Amount, then that amount shall be the Title Defect Amount.

- (iv) If the Title Defect results from the Seller owning a lesser Net Revenue Interest in such Title Defect Property than the Net Revenue Interest specified therefor in Part II(A) of Exhibit A, and the Working Interest attributable to such Property Subdivision has been reduced proportionately, the Title Defect Amount shall be equal to the product obtained by multiplying the Allocated Value specified on Part II(A) of Exhibit A for such Title Defect Property by a fraction, the numerator of which is the difference, if any, between (A) the Net Revenue Interest for such Title Defect Property in Part II(A) of Exhibit A. Title Defect Property in Part II(A) of Exhibit A.
- (v) If the Title Defect results from the existence of a Lien, the Title Defect Amount shall be an amount sufficient to fully discharge such Lien.
- (vi) If the Title Defect results from any matter not described in paragraphs (i), (ii), or (iii) above, the Title Defect Amount shall be determined by the Parties in good faith, taking into account all relevant factors, including the following: (A) the Allocated Value of the affected Title Defect Property; (B) the portion of the Title Defect Property affected by the Title Defect; (C) the legal effect of the Title Defect; (D) the potential economic effect of the Title Defect over the life of the Title Defect Property; (E) the values placed upon the Title Defect by Buyer and Seller; and (F) such other reasonable factors as are necessary to make a proper evaluation.
- (vii) If a Title Defect does not affect a Title Defect Property throughout the entire productive life of such Title Defect Property, such fact shall be taken into account in determining the Title Defect Amount.
- (viii) The Title Defect Amount with respect to a Title Defect Property shall be determined without duplication of any costs or losses included in another Title Defect Amount hereunder. For example, but without limitation, if a Lien affects more than one Title Defect Property or the curative work with respect to one Title Defect results (or is reasonably expected to result) in the curing of any other Title Defect affecting the same or another Title Defect Property, the amount necessary to discharge such Lien or the cost and expense of such curative work shall only be included in the Title Defect Amount for one Title Defect Property and only once in such Title Defect Amount.
- (ix) The Title Defect Amount attributable to a Title Defect Property shall not exceed the Allocated Value of such Title Defect Property. For example, but without limitation, if the Seller does not own fifty percent (50%) of the Net Revenue Interest specified in **Part II(A) of Exhibit A** for a Title Defect Property and such unowned fifty percent (50%) interest is also burdened by a Lien, the Title Defect Amount for such Title Defect Property shall not exceed the Allocated Value allocable to such fifty percent (50%) interest notwithstanding that it may be affected by multiple Title Defects.
- (x) Notwithstanding the foregoing, if the Title Defect Amount determined pursuant to the foregoing with respect to a Title Defect is [***] or less for any Property Subdivision, then the Title Defect Amount with respect to such Title Defect shall be deemed to be zero.

5.3 Determination of Title Defects

A portion of a Property Subdivision shall be deemed to have a "Title Defect" if the Seller fails to have Defensible Title thereto as of the Defect Notice Date. Notwithstanding any other provision in this Agreement to the contrary, the following matters shall not be asserted as, and shall not constitute Title Defects: (a) defects in the early chain of title consisting of the mere failure to recite marital status in a document or omissions of successions of heirship proceedings, unless Buyer provides affirmative evidence that such failure or omission results in another Person's superior claim of title to the relevant Property Subdivision or portion thereof or that another Person is claiming the same; (b) defects arising out of lack of survey unless a survey is expressly required by Laws; (c) defects arising out of lack of corporate authorization, unless Buyer provides affirmative evidence that such corporate action was not authorized and results in another Person's superior claim of title to the relevant Property Subdivision or portion thereof or that another Person is claiming the same; (d) defects that have been cured by possession under the applicable statutes of limitations; (e) defects based solely on the existence of prior oil and gas leases relating to the Subject Interests that are expired and no longer in force and legal effect but not surrendered of record; (f) defects arising from a mortgage encumbering the oil, gas or mineral estate of any lessor if such mortgage would customarily be accepted in taking oil and gas leases and for which the lessee would not customarily seek a subordination of such mortgage to the Subject Interests affected thereby prior to conducting drilling activities on the Subject Interests (unless a complaint of foreclosure has been filed or any similar action taken by the mortgagee thereunder and in such case such mortgage has not been subordinated to the Subject Interests affected thereby or unless the mortgage requires notice to or consent from the mortgagee prior to entering into any oil and gas lease and such notice or consent was not delivered or obtained); (g) defects based solely on lack of information in files of Seller or the Company or references to documents if such documents are not in files of Seller or the Company; (h) Permitted Encumbrances; (i) defects based on a failure to record name changes or mergers set forth on Schedule 5.3(i) in the applicable records; (j) defects to Wells and Bakken Drillsite Spacing Units that result from an adjustment in the acreage contained within such Bakken Drillsite Spacing Unit as set forth on Part II(A) of Exhibit A resulting from the presence of irregular lots in the governmental survey of such Bakken Drillsite Spacing Unit; and (k) defects asserting that the Seller's ownership in a Well is based on a right under a Contract or under a Law of any Governmental Authority entitling the Seller to a non-consenting owner's share in such Well.

5.4 Seller Title Credit

- (a) <u>Seller Title Credit</u>. A "Seller Title Credit" shall mean, (x) with respect to each Property Subdivision described in <u>Part II(A) of Exhibit A</u>, the amount by which the value of such Property Subdivision is enhanced by virtue of the Seller owning a greater Net Revenue Interest in such Property Subdivision than the Net Revenue Interest specified therefor in <u>Part II(A) of Exhibit A</u>, or (y) with respect to each Property Subdivision described in <u>Part II(A) of Exhibit A</u>, the amount by which the value of such Property Subdivision is enhanced by virtue of the Seller owning a lesser Working Interest in such Property Subdivision than the Working Interest specified therefor in <u>Part II(A) of Exhibit A</u>, in each case, as of the Closing Date, which amount shall be determined as follows (such amount, the "Seller Title Credit Amount"):
 - (i) If Buyer and Seller agree on the amount of the Seller Title Credit, then that amount shall be the amount of the Seller Title Credit.
 - (ii) If the Seller Title Credit results from Seller owning a greater Net Revenue Interest in such Property Subdivision than the Net Revenue Interest specified therefor in <u>Part II(A) of Exhibit A</u>, and the Working Interest attributable to such Property Subdivision has been increased proportionately, the Seller Title Credit shall be equal to the product obtained by multiplying the Allocated Value specified on <u>Part II(A) of Exhibit A</u> for such Property

- Subdivision by a fraction, the numerator of which is the difference, if any, between (A) the actual Net Revenue Interest for such Title Defect Property, *minus* (B) the Net Revenue Interest for such Title Defect Property in <u>Part II(A) of Exhibit A</u> and the denominator of which is the Net Revenue Interest specified for such Property Subdivision in <u>Part II(A) of Exhibit A</u>.
- (iii) If the Seller Title Credit results from any matter not described in paragraphs (i) or (ii) above, the Seller Title Credit shall be determined by the Parties in good faith, taking into account all relevant factors.
- (b) <u>Notice of Seller Title Credit</u>. On or before September 19, 2014, Seller shall have the right, but not the obligation, to deliver to Buyer a Seller Title Credit Notice.
- (c) <u>Contents of Seller Title Credit Notice</u>. The Seller Title Credit Notice must be in writing and by asserted in good faith and include (1) a description of the Seller Title Credit, (2) the Property Subdivisions affected by the Seller Title Credit, (3) the Allocated Values of the Property Subdivisions subject to such Seller Title Credit, and (4) the amount by which Seller reasonably believes the Allocated Value of those Property Subdivisions is increased by the Seller Title Credit, and the computations and information upon which Seller's belief is based. Seller shall be deemed to have waived all Seller Title Credits of which it has not given notice on or before the Defect Notice Date.
- (d) <u>Seller Title Credit Amount</u>. Any Seller Title Credit Amounts shall be offset against the aggregate of all Title Defect Amounts, if any, but shall not result in any upward adjustment to the Purchase Price under **Section 2.5**.

5.5 Deferred Claims and Disputes

(a) Deferred Adjustment Claim. In the event that Buyer and Seller have not agreed upon one or more adjustments, credits or offsets claimed by Buyer or Seller pursuant to and in accordance with the requirements of this ARTICLE 5, any such claim (a "Deferred Adjustment Claim") shall be settled pursuant to this Section 5.5 and, except as provided in Section 8.1(e) and Section 8.2(e), shall not prevent or delay Closing; provided, however, if either Party claims that its closing conditions have not been satisfied under Section 8.1(e) or Section 8.2(e), as applicable, and such Party's claim under such Section relies upon alleged Title Defects and/or Title Defect Amounts that are Deferred Adjustment Claims in order to reach the threshold set forth therein, then the Parties will resolve all such Deferred Adjustment Claims under this Section 5.5 prior to Closing, and until all such Deferred Adjustment Claims are resolved, neither Party will be entitled to rely on the failure of the condition in Section 8.1(e) or Section 8.2(e), as applicable, to terminate this Agreement. With respect to each potential Deferred Adjustment Claim, Buyer and Seller shall deliver to the other a written notice describing each such potential Deferred Adjustment Claim, the amount in dispute and a statement setting forth the facts and circumstances that support such Party's position with respect to such Deferred Adjustment Claim. At Closing, the Purchase Price shall not be adjusted on account of, and, except as provided in Section 8.1(e) and Section 8.2(e), no effect shall be given to, the Deferred Adjustment Claim, except, Buyer shall pay that portion of the Purchase Price related to the Deferred Adjustment Claim into the Escrow Account. On or prior to the thirtieth (30th) consecutive calendar day after the Closing Date (the "Deferred Matters Date"), Seller and Buyer shall attempt in good faith to reach agreement on the Deferred Adjustment Claims and, ultimately, to resolve by written agreement all disputes regarding the Deferred Adjustment Claims. Following any written agreement between Seller and Buyer regarding the Deferred Adjustment Claims, Seller and Buyer shall

promptly jointly instruct the Escrow Agent to release to the entitled party the amount of any reduction or increase in the Purchase Price to which Buyer or Seller, as applicable, is entitled.

- Disputed Deferred Adjustment Claims. If Buyer and Seller are unable to resolve all Deferred Adjustment Claims or any disputes regarding any (b) post closing cure of any Title Defects by Seller by the Deferred Matters Date, then such claims shall be resolved by a single arbitrator, who shall be an attorney with at least fifteen (15) years' experience in oil and gas title and transactional matters, as selected by mutual agreement of Buyer and Seller, and if the Parties are not able to mutually agree, then as selected by the Denver office of the Judicial Arbiter Group, Inc. (the "Title Arbitrator"). The arbitration proceeding shall be held in Denver, Colorado, and shall be conducted in accordance with the Commercial Arbitration Rules of the American Arbitration Association, to the extent such rules do not conflict with the terms of this Section 5.5. Buyer and Seller shall exercise commercially reasonable efforts to cause the Title Arbitrator to render his or her determination within fifteen (15) Business Days after a disputed matter is submitted to the Title Arbitrator and such determination shall be final and binding upon the Parties, without right of appeal. In making his or her determination, the Title Arbitrator shall be bound by the rules set forth in this ARTICLE 5 and may consider such other matters as in the opinion of the Title Arbitrator are necessary or helpful to make a proper determination. The Title Arbitrator shall act as an expert for the limited purpose of determining the specific disputed Deferred Adjustment Claims submitted by Buyer or Seller and may not award damages, interest or penalties to either Party with respect to any matter. Seller and Buyer shall each bear (i) its own legal fees and other costs of presenting its case and (ii) one-half of the fees, costs and expenses of the Title Arbitrator. Following the receipt of any final and binding written decision of the Title Arbitrator, Seller and Buyer shall promptly (x) jointly instruct the Escrow Agent to release to the entitled party the amount of any reduction or increase in the Purchase Price to which Buyer or Seller, as applicable, is entitled and (y) pay any amounts due to the other party as a result of the Title Arbitrator's decision.
- (c) <u>Certain Seller Rights</u>. Notwithstanding anything herein provided to the contrary, including **Section 5.2(c)**, at any time prior to the point in time when a final and binding written decision of the Title Arbitrator is made with respect to any Title Defect which constitutes a Deferred Adjustment Claim in accordance with this **Section 5.5**, Seller shall be entitled (i) to cure any such Title Defect and (ii) to resolve all disputes relating to the Deferred Adjustment Claims by payment, by joint instruction to the Escrow Agent instructing the release of such funds to Buyer, of the amount by which the Purchase Price would have been reduced at Closing on account of the Title Defects which constitute Deferred Adjustment Claims if same did not constitute Deferred Adjustment Claims.

5.6 No Duplication

Notwithstanding anything herein provided to the contrary, if a Title Defect results from any matter which could also result in the breach of any representation or warranty of Seller set forth in **Section 3.1** (other than **Section 3.1(m)**), then Buyer shall only be entitled to assert such matter as a Title Defect pursuant to this **ARTICLE 5** and shall be precluded from also asserting such matter as the basis of the breach of any such representation or warranty.

5.7 Title Defect Deductible

Notwithstanding anything herein provided to the contrary, there will be no reduction of the Purchase Price on account of a Title Defect and no amount owing by Seller unless and until the aggregate amount of all undisputed Title Defect Amounts with respect to the Title Defect Properties exceeds [***] (the "Title Defect Deductible"), and in such event, the reduction of the Purchase Price will be limited to the amount in excess of the Title Defect Deductible.

ARTICLE 6 ENVIRONMENTAL MATTERS

6.1 Environmental Defects; Environmental Defect

"Environmental Defect" means any condition or circumstance (other than any condition or circumstance set forth on Schedule 6.1) that, with notice to the Governmental Authority with jurisdiction, would require Remediation under Environmental Laws in, on or under a particular Well and related facilities (including air, land, soil, surface and subsurface strata, surface water, groundwater or sediments) or would cause a particular Well and related facilities to be in violation of any applicable requirements of Environmental Laws or Environmental Permits, in either case that is reasonably expected to require more than [***] to Remediate as to a particular Well and related facilities. "Environmental Defect Amount" means the cost to Remediate the Environmental Defect. "Environmental Defect Properties" means the Wells to which an Environmental Defect relates. Notwithstanding anything to the contrary, if the Environmental Defect Amount with respect to an Environmental Defect Property is [***] or less, then the Environmental Defect Amount with respect to such Environmental Defect shall be deemed to be zero.

6.2 Environmental Defect Notice

Prior to 5:00 P.M., Mountain Time, on the Defect Notice Date, Buyer shall have the right, but not the obligation to notify Seller in writing of any matters that, in Buyer's reasonable opinion, constitute an Environmental Defect (an "Environmental Defect Notice"). Each Environmental Defect Notice shall be in writing and shall contain the following: (i) a description of the asserted Environmental Defect, (ii) the identity of the specific Subject Assets to which such Environmental Defect relates, (iii) the reason Buyer believes such Subject Assets to have an Environmental Defect, (iv) the Environmental Defect Amount attributable thereto claimed in good faith by Buyer, and (v) documentation supporting Buyer's asserted Environmental Defect. To be valid, any Environmental Defect Notice must be received by Seller prior to 5:00 P.M., Mountain Time, on the Defect Notice Date.

6.3 Effect of Environmental Defect Notice

Buyer waives for all purposes all objections associated with the environmental condition of the Subject Assets (including Environmental Defects), unless raised in an Environmental Defect Notice prior to 5:00 P.M., Mountain Time, on the Defect Notice Date. If the aggregate amount of all Environmental Defects Amounts with respect to Environmental Defects, as applicable, timely asserted by Buyer is equal to or less than the Environmental Defect Deductible based on the estimates in the Environmental Defect Notice, Buyer shall be deemed to have (i) accepted the environmental condition(s) in, on and under the affected Subject Assets, (ii) waived its right to claim an Environmental Defect with respect to that particular condition in, on or under the affected Subject Assets, and (iii) included the particular environmental condition(s) as part of the Assumed Liabilities.

6.4 Seller's Right to Cure

Seller shall have until the Closing Date, at its cost and expense, if it so elects but without obligation, to cure all or a portion of asserted Environmental Defects prior to such date with no adjustment to the Purchase Price in respect of such cured Environmental Defect.

6.5 Environmental Defect Adjustments

If Seller and Buyer agree that an Environmental Defect exists and agree on the Environmental Defect Amount, and Seller has been unable, or elects not, to cure such Environmental Defect to the satisfaction of the Parties pursuant to **Section 6.4**, then Buyer shall be entitled, at its sole and exclusive option, to:

- (a) waive the Environmental Defect that remains uncured and reduce the Purchase Price by the amount, if any, by which the aggregate amount of all undisputed Environmental Defect Amounts exceeds the Environmental Defect Deductible (such amount in excess of the Environmental Defect Deductible, the "Environmental Defect Adjustment"), or
- (b) exclude the Environmental Defect Properties affected by such Environmental Defect from the Subject Assets.

6.6 Environmental Defect Deductible

Notwithstanding anything herein provided to the contrary, there will be no reduction of the Purchase Price on account of an Environmental Defect and no amount owing by Seller unless and until the aggregate amount of all undisputed Environmental Defect Amounts with respect to the Environmental Defect Properties (not counting any Environmental Defect Properties that are excluded by Seller pursuant to Section 6.7) exceeds [***] (the "Environmental Defect Deductible"), and in such event, the reduction of the Purchase Price will be limited to the amount in excess of the Environmental Defect Deductible.

6.7 Subject Assets Exclusion

Notwithstanding anything herein provided to the contrary, with respect to any Environmental Defect Properties, Seller may elect prior to Closing to exclude such Environmental Defect Properties from the Subject Assets conveyed to the Company immediately prior to the Closing. If an Environmental Defect Property is excluded under **Section 6.5(b)** or this **Section 6.7**, as the case may be, then (a) all Subject Assets comprising such Environmental Defect Property together with a pro rata share of all incidental rights, oil, gas and other hydrocarbons and other assets attributable or appurtenant thereto, shall be retained by the Seller and shall not be considered part of the Subject Assets and (b) the Purchase Price shall be reduced by the Allocated Value attributable to such Subject Asset (the "**Excluded Environmental Adjustment**").

6.8 Deferred Environmental Claims and Disputes

(a) <u>Deferred Environmental Adjustment Claim</u>. In the event that Buyer and Seller have not agreed on or prior to the Closing Date as to whether an Environmental Defect exists or, if Buyer and Seller agree that an Environmental Defect exists but disagree as to whether it has been cured or on the amount of the Environmental Defect Amount attributable to such Environmental Defect, any such claim (a "**Deferred Environmental Adjustment Claim**") shall be settled pursuant to this **Section 6.8** and, except as provided in **Section 8.1(e)** and **Section 8.2(e)**, shall not prevent or delay Closing; *provided, however*; if either Party claims that its closing conditions have not been satisfied under **Section 8.1(e)** or **Section 8.2(e)**, as applicable, and such Party's claim under such Section relies

upon alleged Environmental Defects and/or Environmental Amounts that are Deferred Environmental Adjustment Claims in order to reach the threshold set forth therein, then the Parties will resolve all such Deferred Environmental Adjustment Claims under this Section 6.8 prior to Closing, and until all such Deferred Environmental Adjustment Claims are resolved, neither Party will be entitled to rely on the failure of the condition in Section 8.1(e) or Section 8.2(e), as applicable, to terminate this Agreement. With respect to each potential Deferred Environmental Adjustment Claim, Buyer and Seller shall deliver to the other a written notice describing each such potential Deferred Environmental Adjustment Claim, the amount in dispute and a statement setting forth the facts and circumstances that support such Party's position with respect to such Deferred Environmental Adjustment Claim, except Buyer shall pay that portion of the Purchase Price related to the Deferred Environmental Adjustment Claim into the Escrow Account. At Closing, the Purchase Price shall not be adjusted on account of, and, except as provided in Section 8.1(e) and Section 8.2(e), no effect shall be given to, the Deferred Environmental Adjustment Claim. On or prior to Deferred Matters Date, Seller and Buyer shall attempt in good faith to reach agreement on the Deferred Environmental Adjustment Claims and, ultimately, to resolve by written agreement all disputes regarding the Deferred Environmental Adjustment Claims. Following any written agreement regarding the Deferred Environmental Adjustment Claims, Seller and Buyer shall promptly jointly instruct the Escrow Agent to release to the entitled party the amount of escrow funds to which Buyer or Seller, as applicable, is entitled.

- (b) Disputed Deferred Environmental Adjustment Claims. If Buyer and Seller are unable to resolve all Deferred Environmental Adjustment Claims by the Deferred Matters Date, then such matters shall be resolved by a single arbitrator, who shall be an attorney with at least fifteen (15) years experience in oil and gas environmental matters, as selected by mutual agreement of Buyer and Seller, and if the Parties are unable to mutually agree, then as selected by the Denver office of the Judicial Arbiter Group, Inc. (the "Environmental Arbitrator"). The arbitration proceeding shall be held in Denver, Colorado, and shall be conducted in accordance with the Commercial Arbitration Rules of the American Arbitration Association, to the extent such rules do not conflict with the terms of this Section 6.8(b). Buyer and Seller shall exercise commercially reasonable efforts to cause the Environmental Arbitrator to render his or her determination within fifteen (15) Business Days after a disputed matter is submitted to the Environmental Arbitrator and such determination shall be final and binding upon the Parties, without right of appeal. In making his or her determination, the Environmental Arbitrator shall be bound by the rules set forth in this ARTICLE 6 and may consider such other matters as in the opinion of the Environmental Arbitrator are necessary or helpful to make a proper determination. The Environmental Arbitrator shall act as an expert for the limited purpose of determining the specific disputed Deferred Environmental Adjustment Claims submitted by either Buyer or Seller and may not award damages, interest or penalties to a Party with respect to any matter. Seller and Buyer shall each bear (i) its own legal fees and other costs of presenting its case and (ii) one-half of the fees, costs and expenses of the Environmental Arbitrator. Following the receipt of any final and binding written decision of the Environmental Arbitrator, Seller and Buyer shall promptly jointly instruct the Escrow Agent to release to the entitled party the amount of escrow funds to which Buyer or Seller, as applicable, is entitled.
- (c) <u>Certain Seller Rights</u>. Notwithstanding anything herein provided to the contrary, at any time prior to the point in time when a final and binding written decision of the Environmental Arbitrator is made with respect to any Environmental Defect which constitutes a Deferred Environmental Adjustment Claim in accordance with this **Section 6.8**, Seller shall be entitled, (i) prior to the Closing Date, to cure any such Environmental Defect or (ii) to resolve all disputes relating to a Deferred Environmental Adjustment Claim by payment, by joint instruction to the Escrow Agent instructing

the release of such funds to Buyer, of the amount by which the Purchase Price would have been reduced at Closing on account of the Environmental Defects which constitute Deferred Environmental Adjustment Claims if same did not constitute Deferred Environmental Adjustment Claims.

6.9 Sole Remedies

Subject to (i) Buyer's termination right as provided in **Section 13.1(a)**, (ii) **Section 6.5**, and (iii) any remedy for breach of any representations and warranties set forth in **Section 3.1**, the procedures set forth in this **ARTICLE 6** shall be Buyer's sole remedy for the existence of any objections associated with the environmental condition of the Subject Assets (including Environmental Defects).

6.10 NORM

Buyer acknowledges that the Subject Assets have been used for exploration, development and production of oil and gas and that there may be petroleum, produced water, wastes, or other substances or materials located in, on or under the Subject Assets or associated with the Subject Assets. Equipment and sites included in the Subject Assets may contain asbestos, NORM or other Hazardous Materials. NORM may affix or attach itself to the inside of wells, materials, and equipment as scale, or in other forms. The wells, materials and equipment located on the Subject Assets or included in the Subject Assets may contain NORM and other wastes or Hazardous Materials. NORM containing material and/or other wastes or Hazardous Materials may have come in contact with various environmental media, including water, soils or sediment. Special procedures may be required for the assessment, remediation, removal, transportation, or disposal of environmental media, wastes, asbestos, NORM and other Hazardous Materials from the Subject Assets. Notwithstanding anything herein to the contrary, Buyer shall not be permitted to claim any Environmental Defect or breach of a representation or warranty set forth in Section 3.1 on the account of the presence of NORM on the properties underlying the Subject Assets.

ARTICLE 7 COVENANTS OF SELLER AND BUYER

7.1 Conduct of Business

- (a) Except as set forth in Section 6.7, Section 7.2, and Section 7.10, from the Execution Date through the Closing, Seller covenants and agrees that it will, and will cause its Affiliates to, (i) conduct the operations with respect to the Subject Assets in good and workmanlike manner in the ordinary course of business consistent with its past practice and good oilfield practices, (ii) not fail or refuse to carry out its duties as "operator" under the terms of the applicable joint operating agreements covering the Subject Assets for which the Seller or any of its Affiliates is currently designated as "operator" such that there exists cause for removal of Seller or any of its Affiliates as operator, (iii) maintain the books of account and records relating to the Subject Assets in the usual, regular and ordinary manner, in accordance with its usual accounting and recordkeeping practices (iv) use commercially reasonable efforts to defend any action to remove Seller or any of its Affiliates as "operator" under the terms of the joint operating agreements covering the Subject Assets and (iv) maintain all insurance coverage as is currently in effect for the Subject Assets.
- (b) From the Execution Date through the Closing, Seller or the Company may acquire new oil and gas leases to cure Title Defects under Section 5.2(e)(ii) at Seller's sole cost and expense.

(c) From the Execution Date until the Closing Date, Seller will notify Buyer (i) if written notice with respect to any alleged material breach of any Lease or Material Contract or any pending claim, suit or action is received or delivered by Seller or its Affiliates; (ii) of any material injury to any employee or contractor of Seller or any of its Affiliates related to the Subject Assets; (iii) of any Casualty Loss; or (iv) of the occurrence of any Material Adverse Effect or any event likely to result in a Material Adverse Effect.

7.2 Certain Restrictions Regarding the Conduct of Business

Unless Buyer shall otherwise consent in writing (such consent not to be unreasonably delayed, conditioned or withheld), and except as set forth in **Section 7.10** and those matters described on **Schedule 3.1(k)** or **Schedule 7.2**, Seller will not, and shall cause the Company not to, from the Execution Date through the Closing:

- (e) convey, encumber or otherwise dispose of any part of the Subject Assets, other than the sale of oil, gas or other hydrocarbons produced from the Subject Interests in the ordinary course of business and the replacement or sale of machinery and equipment in the ordinary course of business;
- (f) modify in any material respect, Willfully Breach or terminate any Material Contract:
- (g) enter, or agree to enter, into any Contract that, if in existence as of the date hereof, would be a Material Contract;
- (h) (i) propose or commit to any single operation, or series of related operations, reasonably anticipated to require capital expenditures by the Company with respect to the Subject Assets in excess of Two Hundred Fifty Thousand and No/100 Dollars (\$250,000.00), (ii) except to the extent required under the terms of an operating agreement with respect to an authority for expenditure issued thereunder as of the date hereof, make any capital expenditures with respect to the Subject Assets and any operation, or series of related operations, relating thereto in excess of Two Hundred Fifty Thousand and No/100 Dollars (\$250,000.00) or (iii) propose or, except to the extent required under the terms of an operating agreement with respect to an authority for expenditure issued thereunder as of the date hereof, make capital expenditures subject to clauses (i) or (ii) in excess of One Million and No/100 Dollars (\$1,000,000.00) in the aggregate;
- (i) enter into any agreement or otherwise settle any Tax claim or assessment relating to the Company;
- (j) make, revoke or amend any tax election with respect to Taxes, adopt, change or modify any accounting method, enter into any settlement of any material issue with respect to Taxes, file any amended Tax Return, or execute or consent to any waivers extending the statutory period of limitations with respect to the collection of any Taxes, in each case, to the extent such action would bind or otherwise affect Buyer, its Affiliates, or the Company at or after the Effective Time; or
- (k) authorize or agree, in writing or otherwise, to take any of the actions prohibited by this **Section 7.2**.

7.3 Press Releases

Seller and Buyer shall promptly consult with each other with regard to all press releases or other public announcements concerning this Agreement or the transactions contemplated herein, and, except as may be required by applicable Laws or the applicable rules and regulations of any Governmental Authority or stock exchange, neither Buyer nor Seller shall issue any such press release or other public announcement

concerning this Agreement or the transactions contemplated hereby that names the other Party without the prior written consent of the other Party, which shall not be unreasonably withheld or delayed. For the avoidance of doubt, either Party may disclose in a filing before the SEC or applicable Canadian securities regulatory authorities that a purchase and sale agreement has been signed and the associated purchase price.

7.4 Records

Within 15 Business Days after the Closing, Seller shall deliver all Records to Buyer's offices as directed by Buyer.

7.5 Further Assurances

Seller and Buyer each agree that, from time to time, whether at or after the Closing Date, each of them will execute and deliver or cause their respective Affiliates to execute and deliver such further instruments of conveyance and transfer, including instruments of conveyance from Baytex Energy USA Ltd., and take such other action as may be reasonably necessary to carry out the purposes and intents of this Agreement, including (a) the completion of the Asset and Liability Transfer, (b) with regard to any Title Defects or Environmental Defects cured on or after Closing, as it relates to Seller, working with Buyer to assure that any cured Subject Asset shall be transferred to Buyer or the appropriate Affiliate of Buyer (as designated by Buyer) and (c) the fulfilling of the conditions to Closing set forth in **Section 8.1** and **Section 8.2**.

7.6 Transfer Requirements

Seller shall (and shall cause its respective Affiliates to) use commercially reasonable efforts to obtain, comply with or otherwise satisfy all Transfer Requirements set forth on Schedule 3.1(g). Seller shall not be obligated to pay any consideration to (or to incur any cost or expenses for the benefit of) the holder of any Transfer Requirements in order to obtain, comply with or otherwise satisfy the same.

7.7 Casualty and Condemnation

If after the Execution Date and prior to the Closing any part of the Subject Assets shall be destroyed or damaged by fire or other casualty or if any part of the Subject Assets shall be taken in condemnation or under the right of eminent domain or if proceedings for such purposes shall be pending or threatened (a "Casualty Loss"), this Agreement shall remain in full force and effect notwithstanding any such destruction, damage, taking or proceeding or the threat thereof. To the extent insurance proceeds, condemnation awards or other payments are not committed, used or applied by Seller prior to the Closing Date to repair, restore or replace such destroyed, damaged or taken Subject Assets, Seller shall at the Closing (a) assign to the Company Seller's or any of its Affiliates' right to receive all insurance or condemnation proceeds, awards or payments owed to Seller or any of its Affiliates by reason of such Casualty Loss, less any reasonable costs and expenses incurred by Seller or any of its Affiliates in collecting same or in connection with such proceedings or the threat thereof, and (b) pay to the Company all insurance or condemnation proceeds, awards or payments theretofore paid to Seller or any of its Affiliates by reason of such Casualty Loss, less any reasonable costs and expenses incurred by Seller or any of its Affiliates in collecting same or in connection with such proceedings or the threat thereof.

7.8 Seller Marks

Buyer shall obtain no right, title, interest, license or any other right whatsoever to use the word "Baytex" or any derivative thereof of any trademarks containing or comprising the foregoing, or any

trademark confusingly similar thereto or dilutive thereof (collectively, the "Seller Marks"). From and after the Closing, Buyer agrees (a) to cease using and to cause the Company to cease using, the Seller Marks in any manner, directly or indirectly, (b) to remove, strike over or otherwise obliterate all Seller Marks from the assets of the Company and from all assets and all other materials owned, possessed or used by the Company within ninety (90) days after the Closing Date, (c) to use commercially reasonable efforts to cause any third parties identified by Seller prior to Closing that use or license Seller Marks in respect of the assets of the Company or on behalf of or with the consent of the Company, to remove, strike over or otherwise obliterate all Seller Marks from all materials owned, possessed or used by such third parties within ninety (90) days after the Closing Date and (d) to change the name of the Company to remove the name "Baytex" and to amend the Organizational Documents of the Company to reflect such name change within ninety (90) days after the Closing Date and (e) to file such documents, reports and affidavits required by the State of North Dakota to evidence the change of name of the Company. The Parties agree, because damages would be an inadequate remedy, that a Party seeking to enforce this Section 7.8 shall be entitled to seek specific performance and injunctive relief as remedies for any breach thereof in addition to other remedies available at law or in equity.

7.9 Certain Tax Matters

- (a) Intended Tax Treatment. The Parties agree that because the Company is a single-member limited liability company that is wholly-owned by Seller, based on Seller's representations in Section 3.1(i), the acquisition by Buyer of the Membership Interests is treated as an acquisition by Buyer of the assets of the Company for United States federal Income Tax purposes (collectively, the "Intended Tax Treatment"). Neither Buyer nor Seller shall take any actions which are inconsistent with the Intended Tax Treatment unless required by applicable Law. Neither Buyer nor Seller shall take any position on any Tax Return or other similar document, or take any other Tax reporting position, for United States federal, state and local Income Tax purposes, that is inconsistent with the Intended Tax Treatment.
- (b) <u>Apportionment</u> of <u>Property</u> <u>Taxes</u>.
 - (i) Property Taxes shall be deemed attributable to the period during which ownership of the applicable Subject Assets gives rise to liability for such Property Taxes, and liability therefor allocated to Seller for all periods ending prior to the Effective Time and to Buyer for all periods beginning on or after the Effective Time. The amount of Property Taxes allocable to the Seller shall be equal to the amount of such Property Taxes for a taxable period beginning before and ending after the Effective Time allocated to the Seller shall be equal to the amount of such Property Taxes for the entire taxable period multiplied by a fraction the numerator of which is the number of days in the taxable period ending on the Closing Date and the denominator of which is the number of days in such taxable period. To the extent the actual amount of Property Tax that is attributable to a period ending prior to the Effective Time (but not due and payable prior to the Closing) is not determinable at the time an adjustment to the Purchase Price is to be made with respect to such Property Tax pursuant to Section 2.5 and/or Section 2.6, as applicable, the Parties shall utilize the most recent information available in estimating the amount of such Property Tax for purposes of such adjustment, and the amount of such Property Tax used for purposes of Section 2.6 shall be considered full and final settlement of such Property Tax without regard to the actual Tax rates or assessments (the "Settled Property Taxes").

- (ii) Buyer shall be entitled to all rights to any refunds of (1) Property Taxes allocable to Buyer pursuant to this **Section 7.9(b)** regardless of when received, and (2) Settled Property Taxes. Seller shall be entitled to all rights to any refunds of Property Taxes allocated to Seller pursuant to this **Section 7.9(b)** other than refunds of Settled Property Taxes. After Closing, Buyer shall timely file or cause to be filed all Tax Returns for Property Taxes required to be filed after the Closing and shall timely pay or cause to be paid to the taxing authorities all Property Taxes that become due and payable after the Closing. Any penalty, addition to Tax, or interest levied or assessed with respect to any Property Tax shall be allocated to, and shall be payable by, the Party to which the Tax to which such penalty, addition to Tax or interest relates is allocated, regardless of when such penalty, addition to Tax, or interest is levied or assessed; *provided, however*, that the liability for any penalty, addition to Tax, or interest levied or assessed with respect to any failure of Buyer to comply with the previous sentence shall be allocated to, and shall be payable by, Buyer.
- (c) Apportionment of Severance Taxes. Severance Taxes shall be deemed attributable to the period during which the production of the hydrocarbons with respect to such Severance Taxes occurred, and liability therefor shall be allocated to Seller for pre-Effective Time Severance Taxes and to Buyer for post-Effective Time Severance Taxes. For the purpose of calculating the Initial Adjustment Amount under Section 2.5(a) and for purposes of the Final Settlement Statement under Section 2.6, Severance Taxes for which the actual liability amount is not known as of such time shall be estimated based on the applicable current Severance Tax rates, with the resulting Adjusted Purchase Price under Section 2.6 to be considered full and final settlement of all such Taxes without regard to the actual Tax rates or assessments; provided that, if the actual amounts of the applicable Severance Tax rates are not known at the time of the Closing, the amounts shall be re-estimated based on the best information available at the time of the Final Settlement Statement, and such estimated amounts shall thereupon become full and final settlement of such Severance Taxes (the "Settled Severance Taxes," and together with the Settled Property Taxes, the "Settled Asset Taxes") for purposes of this Agreement. Buyer shall be entitled to all rights to any refunds of (a) Severance Taxes allocable to Buyer pursuant to this Section 7.9(c) regardless of when received, and (b) Settled Severance Taxes. Seller shall be entitled to all rights to any refunds of Severance Taxes allocated to Seller pursuant to this Section 7.9(c) other than refunds of Settled Severance Taxes. After Closing, Buyer shall timely file or cause to be filed all Tax Returns for Severance Taxes required to be filed after the Closing and shall timely pay or cause to be paid to the taxing authorities all Severance Taxes that become due and payable after the Closing. Any penalty, addition to Tax, or interest levied or assessed with respect to any Severance Tax shall be allocated to, and shall be payable by, the Party to which the Tax to which such penalty, addition to Tax or interest relates is allocated, regardless of when such penalty, addition to Tax, or interest is levied or assessed; provided, however, that the liability for any penalty, addition to Tax, or interest levied or assessed with respect to any failure of Buyer to comply with the previous sentence shall be allocated to, and shall be payable by, Buyer.
- (d) <u>Income Taxes</u>. Each Party shall be responsible for its own Income Taxes.
- (e) <u>Transfer Taxes</u>. All Transfer Taxes (including any related interest, penalties or legal costs, but excluding any recordation fees incurred in connection with the transfer of the Subject Assets) that may be imposed on any transfer of the membership interests in the Company (or the Subject Assets) pursuant to this Agreement shall be borne and paid 50% by Buyer and 50% by Seller. The Parties shall cooperate with one another in the preparation of any Tax Returns and other related documentation with respect to such Transfer Taxes (including any exemption certificates and forms as each may request to establish an exemption from (or otherwise reduce) or make a report with

respect to Transfer Taxes). Any expenses, fees or costs incurred by Seller in connection with preparing or filing such Tax Returns shall be paid by Buyer.

- (f) Allocations for Federal Income Tax Purposes. Buyer and Seller acknowledge that, under Section 1060 of the Code, Buyer and Seller must report information regarding the allocation of the Purchase Price and any other amounts treated as consideration for federal Income Tax purposes (collectively, the "Allocation Amount") to the United States Secretary of Treasury by attaching Department of Treasury, Internal Revenue Service, Form 8594 to their federal Income Tax returns for the tax period which includes the Closing Date. Buyer and Seller agree that the Allocation Amount shall be allocated among the Subject Assets for Tax purposes in accordance with an allocation schedule substantially consistent with Schedule 7.9(f) (the "Allocation Schedule"). Buyer and Seller shall negotiate in good faith to revise the Allocation Schedule in a manner consistent with Section 1060 of the Code and the Treasury Regulations promulgated thereunder to take into account any adjustments to the Purchase Price pursuant to this Agreement, including any indemnification payments pursuant to ARTICLE 12. Buyer and Seller shall each prepare their respective Forms 8594 in a manner consistent with the Allocation Schedule. The Parties shall not take any Tax position (whether in audits, on Tax Returns, or otherwise) that is inconsistent with such allocation unless required to do so by applicable Law; provided, however, that nothing contained herein shall prevent Buyer or Seller from settling any proposed deficiency or adjustment by any Taxing Authority based upon or arising out of such Allocation Schedule, and neither Buyer nor Seller shall be required to litigate before any court any proposed deficiency or adjustment by any Taxing Authority challenging such allocation.
- (g) <u>Post-Closing Tax Matters</u>. After the Closing, each of Buyer and Seller shall:
 - (i) reasonably cooperate and assist the other (1) in preparing any Tax Returns relating to any Tax imposed with respect to the Subject Assets or the Company, or the Transaction, and (2) in qualifying for any exemption or reduction in Tax that may be available;
 - (ii) reasonably cooperate in preparing for any audits, examinations or other tax proceedings by, or disputes with, taxing authorities regarding any Tax relating to the Subject Assets, the Company, or the Transaction;
 - (iii) make available to the other, and to any Taxing Authority as reasonably requested, any information, records, and documents relating to a Tax incurred or imposed in connection with the Subject Assets or the Transaction;
 - (iv) provide timely notice to the other in writing of any pending or threatened Tax audit, examination, or assessment that could reasonably be expected to affect the other's Tax liability under applicable Law or this Agreement (a "Tax Controversy"), and to promptly furnish the other with copies of all correspondence with respect to any Tax Controversy; and
 - (v) allow the other to participate, at its own expense, in any Tax Controversy, and not settle any Tax Controversy without the prior written consent of the other, which may not be unreasonably withheld, conditioned, or delayed.
- (h) <u>Effect of Indemnity</u>. Seller and Buyer shall treat any indemnity payment made pursuant to this Agreement as an adjustment to the Purchase Price for Tax purposes unless otherwise required by Law.

7.10 Asset and Liability Transfer

- (a) Prior to the Closing, the Seller shall undertake the actions contemplated by this **Section 7.10**, which collectively shall be referred to as the "**Asset and Liability Transfer**".
- (b) Prior to the Closing, Seller shall create the Company as a limited liability company under the laws of the State of Delaware with the Organizational Documents, and will cause the Company to issue the Membership Interests to the Seller. The Seller shall not cause or permit the Company to issue any equity interests other than the Membership Interests.
- (c) Immediately prior to the Closing, Seller shall execute, acknowledge and deliver to the Company, and shall cause the Company to execute, acknowledge and deliver to the Seller, an Assignment, Bill of Sale and Conveyance in the form attached as **Exhibit C** (the "**General Conveyance**"), and any applicable forms of any Governmental Authorities, conveying the Subject Assets to the Company as of the Effective Time, with a special warranty of title by, through and under Seller but not otherwise.
- (d) Immediately prior to the Closing, Seller shall execute, acknowledge and deliver to the Company, and shall cause the Company to execute, acknowledge and deliver to the Seller, an Assignment and Assumption Agreement in the form attached as **Exhibit D** (the "**Assumption Agreement**") pursuant to which the Company shall assume and agree to pay, perform, fulfill and discharge the Assumed Liabilities.
- (e) Any time after the formation of the Company, the Seller shall not, and shall cause the Company not to:
 - (i) amend the Organizational Documents;
 - (ii) split, combine or reclassify any membership interests in the Company;
 - (iii) sell or create any Lien on any membership or other equity interest in the Company;
 - (iv) incur, assume, or guarantee any indebtedness for borrowed money;
 - (v) adopt any plan of merger, consolidation, reorganization, liquidation or dissolution or filing of a petition in bankruptcy under any provisions of federal or state bankruptcy Law or consent to the filing of any bankruptcy petition against it under any similar Law; or
 - (vi) authorize or agree, in writing or otherwise, to take any of the actions prohibited by this Section 7.10(e).
- (f) The Company shall not undertake any activities other than as contemplated by this Agreement and the Seller shall not cause the Company to assume any liabilities other than as provided in the General Conveyance and Assumption Agreement.

7.11 No Shop/No Talk

The Seller shall not, and shall cause its Affiliates and the Company not to, directly or indirectly, through any representative or otherwise, solicit or entertain offers from, provide any information to, negotiate with or in any manner encourage, or accept any proposal of any other Person relating to the acquisition of all, substantially all, or any material portion of the Subject Assets of the Seller or equity interests the Company,

or their assets or businesses, whether directly or indirectly, through purchase, merger, consolidation or otherwise.

7.12 Employee Covenant

- (a) Seller waives the restrictions contained in Section 20 of the Confidentiality Agreement with regard to the employees of Seller, other than the officers of Seller (each a "Business Employee" and, collectively, the "Business Employees"). Buyer may, on or after the date hereof, interview any Business Employee and conduct any such standard employee screening and eligibility procedures with regard to such Business Employees as Buyer conducts with respect to candidates for employment in Buyer's ordinary course of business. Subject to applicable Law and, with the prior written consent of each Business Employee, as applicable, Buyer shall be entitled to review copies of the personnel records maintained by Seller with respect to such Business Employee, and discuss such records and such Business Employee with Seller's representatives. Buyer's interviews of the Business Employees as set forth in this Section 7.12 shall be conducted so as not to unreasonably interfere with the Business of Seller prior to the Closing, and Seller shall exercise commercially reasonable efforts to cooperate with Buyer and to facilitate Buyer's completion of its interview and screening procedures hereunder.
- (b) After the date hereof, Buyer shall be authorized to communicate with any Business Employee regarding such Business Employee's potential employment with Buyer and Buyer shall be authorized to communicate any offer of employment to any Business Employees. Buyer shall, not later than 5 days prior to the Closing, deliver to Seller a final written list containing the name of each Business Employee Buyer desires to hire. Each Business Employee that is offered employment with Buyer and that accepts such offer (each a "Continuing Employee") and, collectively, the "Continuing Employees") shall, as of the Closing be terminated by Seller and become an employee of Buyer.
- (c) Seller shall be responsible for all compensation and benefits owing to Business Employees up to and including the Closing Date, and Buyer shall be responsible for all compensation and benefits owing to Continuing Employees after the Closing Date under Buyer's compensation and benefits programs (on and subject to the terms of employment agreed upon between Buyer and such Continuing Employee). Buyer shall have no obligations with respect to any Business Employee that is not hired by Buyer at the Closing.
- (d) With respect to each Continuing Employee:
 - (vi) In the event that, prior to the 6-month anniversary of the Closing Date, Buyer terminates the employment of any Continuing Employee, other than for cause (as determined pursuant to Buyer's applicable employment policies), Buyer shall pay to such Continuing Employee, within five (5) days following termination of such Continuing Employee, the Severance Amount.
 - (vii) In accordance with and solely to the extent permitted by and in accordance with Buyer's current policies and programs (i) Buyer shall grant such Continuing Employee credit for their service with Seller for purposes of eligibility to participate in and vesting credits in Buyer's service award plans, vacation programs and policies and retirement plans; (ii) for the calendar year during which such coverage begins, Buyer shall credit under any medical, dental or vision benefit plans maintained by Buyer, all Continuing Employees with any deductibles and co-payments already incurred during such year under Seller's benefit plans, and waive any pre-existing conditions, exclusions or limitations as to coverage, any

evidence-of-insurability provisions and any waiting-period requirements under such plans; and (iii) Buyer shall apply towards any deductible requirements and out-of-pocket maximum limits under such welfare benefit plans that are applicable to 2014, any similar deductible requirements and out-of-pocket maximum limits satisfied by any Continuing Employee under Seller's welfare benefit plans. Prior to Closing, Seller shall prepare and deliver, or cause to be prepared and delivered, to each Continuing Employee, and request such Continuing Employee to deliver to Buyer, a schedule setting forth the amount of the deductible and out-of-pocket maximum limits satisfied by each Continuing Employee under Seller welfare benefit plans with the information available as of the date the schedule is prepared, which schedule shall be updated through the Closing Date as soon as reasonably practicable after the Closing.

- (viii) To the extent permitted by Buyer's current 401(k) plan, Buyer shall cause the trustee or administrator of Buyer's 401(k) plan to accept direct rollovers from Seller's 401(k) plan trust for each Continuing Employee electing the same with respect to a distribution of his or her vested account thereunder. If any Continuing Employee does not rollover his or her account to Buyer's 401(k) plan, such Continuing Employee may continue to participate in Seller's 401(k) plan as a terminated participant in accordance with the terms thereof, and Seller shall retain all responsibility for the management and administration of such Continuing Employee's account balances under Seller's 401(k) plan.
- (ix) Each Continuing Employee shall, on the Closing Date, begin accruing vacation time at the level to which they would be entitled under Buyer's vacation policies for 2014. Seller will pay to each Business Employee an amount equal to any accrued and unused vacation days at the Closing in accordance with its policies.
- (e) For the 6 month period following the Closing Date, if Buyer or any of its Affiliates directly or indirectly retains the services of any Business Employee to whom Seller paid the Severance Amount because the employment of such Business Employee was terminated, Buyer shall immediately reimburse Seller for the Severance Amount, except to the extent (a) that such Business Employee applies for employment with Buyer in response to a general solicitation or posting by Buyer of an open position, or (b) such Business Employee was not solicited by Buyer.
- (f) Notwithstanding anything in this Agreement to the contrary, (i) nothing in this Agreement shall create any obligation on the part of Buyer to continue the employment of any employee for any definite period following the Closing Date, and (ii) nothing in this Agreement shall preclude Buyer from altering, amending, or terminating any of its welfare benefit plans, or the participation of any of its employees in such welfare benefit plans, at any time.
- (g) No provision in this Agreement shall create any third party beneficiary or other right in any Person (including any beneficiary or dependent thereof) for any reason in respect of continued employment or new employment with Seller, the Company or Buyer or in respect of any benefits that may be provided, directly or indirectly, under any plan or arrangement maintained by Seller, the Company or Buyer.

ARTICLE 8 CLOSING CONDITIONS

8.1 Seller's Closing Conditions

The obligation of Seller to proceed with the Closing is subject, at the option of Seller, to the satisfaction on or prior to the Closing Date of all of the following conditions:

- (1) Representations, Warranties, Covenants and Agreements.
 - (i) The representations and warranties of Buyer contained in **Section 3.3** shall be true and correct in all material respects (and in all respects, in the case of representations and warranties which are qualified by materiality) on and as of the Closing Date as though made as of the Closing Date (except to the extent such representations and warranties expressly relate to an earlier date, in which case such representations and warranties shall be deemed to refer to such earlier date referenced in such representation and warranty); and
 - (ii) The covenants and agreements of Buyer to be performed on or before the Closing Date in accordance with this Agreement shall have been duly performed in all material respects.
- (m) Officer's Certificate. Seller shall have received a certificate dated as of the Closing Date, executed by a duly authorized officer of Buyer, to the effect that the conditions set forth in Section 8.1(a) have been satisfied.
- (n) <u>Closing Deliveries</u>. On or prior to the Closing Date, Buyer shall have delivered, or be standing ready to deliver at Closing, all agreements, instruments and other documents or items required to be delivered by Buyer pursuant to **Section 9.3**.
- (o) No Action. On the Closing Date, no material Action (excluding any such matter initiated by Seller or any of its Affiliates) shall be pending or threatened before any Governmental Authority of competent jurisdiction seeking to enjoin or restrain the consummation of the Closing or recover damages from Seller resulting therefrom, and no Governmental Authority shall have enacted, issued, promulgated, enforced or entered any order which is in effect and has the effect of making the transactions contemplated by this Agreement illegal, or otherwise restraining or prohibiting consummation of such transactions or causing any of the transactions contemplated hereunder to be rescinded following completion hereof.
- (p) <u>Adjustments</u>. The sum of the (i) reduction in the Purchase Price on account of the aggregate amount of all undisputed Title Defect Amounts, (ii) aggregate amount of Title Defect Amounts claimed by Buyer with respect to unresolved Deferred Adjustment Claims, (iii) reduction in the Purchase Price on account of the aggregate amount of all undisputed Environmental Defect Amounts pursuant to **Section 6.5** and the exclusion of Environmental Defect Properties pursuant to **Section 6.7** and (iv) aggregate amount of Environmental Defect Amounts claimed by Buyer with respect to unresolved Deferred Environmental Adjustment Claims, shall not exceed [***] of the Purchase Price.

8.2 Buyer's Closing Conditions

The obligation of Buyer to proceed with the Closing is subject, at the option of Buyer, to the satisfaction on or prior to the Closing Date of all of the following conditions:

(d) Representations, Warranties, Covenants and Agreements.

- (i) The representations and warranties of Seller contained in Section 3.1(a), Section 3.1(b), Section 3.1(c), Section 3.1(w), Section 3.2(a), Section 3.2(b), Section 3.2(c), Section 3.2(d), Section 3.2(h) and Section 3.2(j) (collectively, the "Flat Reps") shall be true and correct on and as of the Closing Date as though made as of the Closing Date (except to the extent such representations and warranties expressly relate to an earlier date, in which case such representations and warranties shall be deemed to refer to such earlier date referenced in such representation and warranty);
- (ii) The representations and warranties of Seller contained in **Section 3.1** and **Section 3.2** (other than the Flat Reps) shall be true and correct on and as of the Closing Date as though made as of the Closing Date (except to the extent such representations and warranties expressly relate to an earlier date, in which case such representations and warranties shall be deemed to refer to such earlier date referenced in such representation and warranty), except where the failure of such representations and warranties to be true and correct would not have a Material Adverse Effect; and
- (iii) The covenants and agreements of Seller to be performed on or before the Closing Date in accordance with this Agreement shall have been duly performed in all material respects.
- (e) Officer's Certificate. Buyer shall have received a certificate dated as of the Closing Date, executed by a duly authorized officer of Seller, to the effect that the conditions set forth in **Section 8.2(a)** have been satisfied.
- (f) <u>Closing Deliveries</u>. On or prior to the Closing Date, Seller shall have delivered, or be standing ready to deliver at Closing, all agreements, instruments and other documents or items required to be delivered by Seller pursuant to **Section 9.2**.
- (g) No Action. On the Closing Date, no material Action (excluding any such matter initiated by Buyer or any of its Affiliates) shall be pending or threatened before any Governmental Authority of competent jurisdiction seeking to enjoin or restrain the consummation of the Closing or recover damages from Buyer resulting therefrom, and no Governmental Authority shall have enacted, issued, promulgated, enforced or entered any order which is in effect and has the effect of making the transactions contemplated by this Agreement illegal, or otherwise restraining or prohibiting consummation of such transactions or causing any of the transactions contemplated hereunder to be rescinded following completion hereof.
- (h) Adjustments. The sum of the (i) reduction in the Purchase Price on account of the aggregate amount of all undisputed Title Defect Amounts, (ii) aggregate amount of Title Defect Amounts claimed by Buyer with respect to unresolved Deferred Adjustment Claims, (iii) reduction in the Purchase Price on account of the aggregate amount of all undisputed Environmental Defect Amounts pursuant to **Section 6.5** and the exclusion of Environmental Defect Properties pursuant to **Section 6.7** and (iv) aggregate amount of Environmental Defect Amounts claimed by Buyer with respect to unresolved Deferred Environmental Adjustment Claims, shall not exceed [***] of the Purchase Price.
- (i) <u>Asset and Liability Transfer</u>. The Asset and Liability Transfer shall have been completed in the manner set forth in **Section 7.10**.

ARTICLE 9 CLOSING

9.1 Closing

The Closing shall be held at 10:00 A.M., Mountain Time, at the offices of Davis Graham & Stubbs LLP, at 1550 17 th Street, Suite 500, Denver, Colorado, on September 24, 2014 (the "**Scheduled Closing Date**"), or, if all of the conditions to Closing set forth in **Article 8** have been satisfied or waived (other than conditions which, by their nature, are to be satisfied on the Closing Date) as of such date, on the date that is five (5) days following satisfaction (or waiver) of such conditions, or at such other time or on such other date at such other place as Seller and Buyer may otherwise agree in writing (the day on which the Closing takes place being the "**Closing Date**").

9.2 Seller's Closing Obligations

At Closing, Seller shall:

- (c) execute and deliver the Assignment of Membership Interests in the form of **Exhibit E** duly executed by Seller, which Membership Interests shall be free and clear of all Liens;
- (d) execute and deliver a certificate of non-foreign status as described in Treasury Regulations § 1.1445-2(b)(2), substantially in the form attached hereto as **Exhibit F**;
- (e) execute and deliver the officer's certificate referred to in **Section 8.2(b)**;
- (f) deliver copies of all consents and approvals of third parties, including Governmental Authorities, in form and content reasonably acceptable to Buyer, the granting or receipt of which are necessary for the consummation of the transactions contemplated by this Agreement;
- (g) deliver written resignations and mutual releases of the officers and managers of the Company, effective as of the Closing Date;
- (h) deliver evidence to the effect that the Company shall have been released as a guarantor under any instrument governing the terms of Seller's or its Affiliates' debt instruments;
- (i) deliver a recordable release of any trust, mortgages, financing statements, fixture filings and security agreements made by Seller or its Affiliates affecting the Subject Assets;
- (j) execute and deliver the Parent Guaranty;
- (k) execute and deliver joint instructions to the Escrow Agent regarding the Cash Deposit as agreed to by Seller and Buyer pursuant to Section 2.4, Section 5.5 and Section 6.8; and
- (1) execute and deliver any other agreements, instruments and documents which are required by other terms of this Agreement to be executed and/or delivered by Seller to Buyer at the Closing.

9.3 Buyer's Closing Obligations

At Closing, Buyer shall:

- (a) pay the Closing Amount by wire transfer of immediately available funds and as otherwise described in ARTICLE
 2;
- (b) execute and deliver, or cause to be executed and delivered, to Seller the officer's certificate of Buyer referred to in Section 8.1(b);
- (c) execute and deliver joint instructions to the Escrow Agent regarding the Cash Deposit as agreed to by Seller and Buyer pursuant to Section 2.4, Section 5.5 and Section 6.8; and
- (d) execute and deliver any other agreements, instruments and documents which are required by other terms of this Agreement to be executed and/or delivered by Buyer to Seller at the Closing.

ARTICLE 10 SURVIVAL

10.1 Survival

No representation, warranty, covenant or agreement made herein shall survive the Closing except as provided in this Section 10.1. Each representation, warranty, covenant and agreement made herein shall terminate and cease to be of further force and effect as of the Closing or such later date after Closing as is expressly stipulated in this Section 10.1 for the survival thereof. Following the Closing or such later date stipulated in this Section 10.1 for the survival thereof, such representation, warranty, covenant or agreement shall not form the basis for or give rise to any claim, demand, cause of action, counterclaim, defense, damage, indemnity, obligation or liability which is asserted, claimed, made or filed following the Closing or such later date stipulated in this Section 10.1 for survival. It is expressly agreed that the terms and provisions of (a) ARTICLE 3 shall survive for twelve (12) months following the Closing, except that (i) Section 3.1(i) shall survive the Closing until sixty (60) days following the expiration of the applicable statutes of limitations (including extensions) for the subject Taxes and (ii) Seller's representations and warranties set forth set forth in Section 3.1(a), Section 3.1(b), Section 3.1(m), Section 3.1(w), Section 3.1(x), Section 3.2(a), Section 3.2(b), Section 3.2(c) and Section 3.2(i) (the "Fundamental Representations") and Buyer representations and warranties set forth in Section 3.3 shall, in each case, survive the Closing indefinitely, (b) Section 2.5, Section 2.6, Section 4.2, ARTICLE 5, ARTICLE 6, ARTICLE 10, ARTICLE 11, ARTICLE 12, ARTICLE 14, and Buyer's indemnity and hold harmless of the Seller Indemnified Parties under Section 4.1, shall survive the Closing indefinitely or for such shorter period of time as may be stipulated in such provisions and (c) the covenants contained herein shall survive in accordance with their terms. In addition, the definitions set forth in Appendix A to this Agreement which are used in the representations, warranties, covenants and agreements which survive the Closing pursuant to this Section 10.1 shall survive the Closing to the extent necessary to give operative effect to such surviving representations, warranties, covenants and agreements.

10.2 Acknowledgement of Limitation of Recovery

Both Parties acknowledge that the limitations on survival in **Section 10.1** are a contractual statute of limitations that limits such Party's ability to make a claim against the other Party or, with respect to Buyer, the Company, which such Party may otherwise have available under Law.

ARTICLE 11 LIMITATIONS

11.1 Disclaimer of Warranties

NOTWITHSTANDING ANYTHING CONTAINED TO THE CONTRARY IN ANY OTHER PROVISION OF THIS AGREEMENT. IT IS THE EXPLICIT INTENT OF EACH PARTY THAT SELLER IS NOT MAKING ANY REPRESENTATION OR WARRANTY WHATSOEVER, EXPRESS, IMPLIED, STATUTORY OR OTHERWISE, BEYOND THOSE REPRESENTATIONS AND WARRANTIES EXPRESSLY GIVEN IN THIS AGREEMENT AND THE SPECIAL WARRANTY OF TITLE PURSUANT TO THE GENERAL CONVEYANCE. AND IT IS UNDERSTOOD THAT, WITHOUT LIMITING SUCH EXPRESS REPRESENTATIONS AND WARRANTIES, SELLER'S INTEREST IN THE COMPANY AND THE ASSETS OF THE COMPANY ARE BEING TRANSFERRED AS IS AND WHERE IS AND WITH ALL FAULTS. WITHOUT LIMITING THE GENERALITY OF THE IMMEDIATELY PRECEDING SENTENCE. SELLER HEREBY (a) EXPRESSLY DISCLAIMS AND NEGATES ANY REPRESENTATION OR WARRANTY, EXPRESS OR IMPLIED, AT COMMON LAW, BY STATUTE OR OTHERWISE, RELATING TO (i) THE CONDITION OF THE SUBJECT ASSETS (INCLUDING, WITHOUT LIMITATION, ANY IMPLIED OR EXPRESS WARRANTY OF MERCHANTABILITY OR FITNESS FOR A PARTICULAR PURPOSE, OR OF CONFORMITY TO MODELS OR SAMPLES OF MATERIALS, OR THE PRESENCE OR ABSENCE OF ANY HAZARDOUS MATERIALS IN OR ON, OR DISPOSED OF OR DISCHARGED FROM, THE SUBJECT ASSETS) OR (ii) ANY INFRINGEMENT BY SELLER OR ANY OF ITS AFFILIATES OF ANY PATENT OR PROPRIETARY RIGHT OF ANY PERSON; AND (b) NEGATES ANY RIGHTS OF BUYER UNDER STATUTES TO CLAIM DIMINUTION OF CONSIDERATION AND ANY CLAIMS BY BUYER FOR DAMAGES BECAUSE OF DEFECTS, WHETHER KNOWN OR UNKNOWN, IT BEING THE INTENTION OF SELLER AND BUYER THAT SELLER'S INTEREST IN THE COMPANY AND THE ASSETS OF THE COMPANY ARE BEING TRANSFERRED TO AND ACCEPTED BY BUYER IN THEIR PRESENT CONDITION AND STATE OF REPAIR.

11.2 Damages

NOTWITHSTANDING ANYTHING CONTAINED TO THE CONTRARY IN ANY OTHER PROVISION OF THIS AGREEMENT, SELLER AND BUYER AGREE THAT, EXCEPT FOR THE AMOUNT SPECIFICALLY STIPULATED IN SECTION 13.2, THE RECOVERY BY ANY PARTY OF ANY DAMAGES SUFFERED OR INCURRED BY IT AS A RESULT OF ANY BREACH BY ANOTHER PARTY OF ANY OF ITS REPRESENTATIONS, WARRANTIES OR OBLIGATIONS UNDER THIS AGREEMENT SHALL BE LIMITED TO THE ACTUAL DAMAGES SUFFERED OR INCURRED BY THE NON-BREACHING PARTY (AND THE INDEMNIFIED PERSONS TO WHICH SUCH OBLIGATIONS MAY EXTEND UNDER THE EXPRESS TERMS HEREOF) AS A RESULT OF THE BREACH BY THE BREACHING PARTY OF ITS REPRESENTATIONS, WARRANTIES OR OBLIGATIONS HEREUNDER AND IN NO EVENT SHALL THE BREACHING PARTY BE LIABLE TO THE NON-BREACHING PARTY OR ANY INDEMNIFIED PERSON FOR ANY INDIRECT, CONSEQUENTIAL, SPECIAL, EXEMPLARY OR PUNITIVE DAMAGES (INCLUDING ANY DAMAGES ON ACCOUNT OF LOST PROFITS OR OPPORTUNITIES, BUSINESS INTERRUPTION OR LOST OR DELAYED PRODUCTION) SUFFERED OR INCURRED BY THE NON-BREACHING PARTY OR ANY INDEMNIFIED PERSON AS A RESULT OF THE BREACH BY THE BREACHING PARTY OF ANY OF ITS REPRESENTATIONS, WARRANTIES OR OBLIGATIONS HEREUNDER. For purposes of the foregoing, actual damages may, however, include indirect, consequential, special, exemplary

or punitive damages to the extent (a) the injuries or losses resulting in or giving rise to such damages are incurred or suffered by a Person which is not a Seller Indemnified Party, a Buyer Indemnified Party or an Affiliate of any of the foregoing and (b) such damages are recovered against an Indemnified Person by a Person which is not a Seller Indemnified Party, a Buyer Indemnified Party or an Affiliate of any of the foregoing. This **Section 11.2** shall operate only to limit a Party's liability and shall not operate to increase or expand any contractual obligation of a Party or cause any contractual obligation of a Party to survive longer than provided in **Section 10.1**.

ARTICLE 12 INDEMNIFICATION

12.1 Indemnification By Buyer

FROM AND AFTER THE CLOSING, SUBJECT TO THE LIMITATIONS OF **SECTION 10.1** AND **ARTICLE 11**, BUYER SHALL INDEMNIFY, DEFEND AND HOLD HARMLESS THE SELLER INDEMNIFIED PARTIES FROM AND AGAINST ANY AND ALL (a) ASSUMED LIABILITIES INCURRED BY OR ASSERTED AGAINST ANY OF THE SELLER INDEMNIFIED PARTIES, INCLUDING ANY ASSUMED LIABILITY BASED ON NEGLIGENCE, GROSS NEGLIGENCE OR STRICT LIABILITY OF THE COMPANY OR ANY SELLER INDEMNIFIED PARTY OR ANY OTHER THEORY OF LIABILITY, WHETHER IN LAW (WHETHER COMMON OR STATUTORY) OR EQUITY (SO LONG AS SUCH ASSUMED LIABILITIES ARE NOT COVERED LIABILITIES FOR WHICH BUYER IS ENTITLED TO INDEMNIFICATION PURSUANT TO **SECTION 12.2**), AND (b) COVERED LIABILITIES RESULTING FROM ANY MISREPRESENTATION, BREACH OF ANY REPRESENTATION OR WARRANTY OR NONFULFILLMENT OF ANY COVENANT OR AGREEMENT ON THE PART OF BUYER INCLUDED IN THIS AGREEMENT OR THE CERTIFICATES DELIVERED UNDER **SECTION 8.2(b)**.

12.2 Indemnification By Seller

FROM AND AFTER THE CLOSING, SELLER SHALL INDEMNIFY, DEFEND AND HOLD HARMLESS THE BUYER INDEMNIFIED PARTIES FROM AND AGAINST ANY AND ALL (a) RETAINED LIABILITIES AND (b) COVERED LIABILITIES (i) RESULTING FROM EXCLUDED ENVIRONMENTAL DEFECT PROPERTIES UNDER **SECTION 6.5** OR **SECTION 6.7**, AND (ii) SUBJECT TO THE PROVISIONS OF **SECTION 10.1**, **SECTION 12.3**, AND **ARTICLE 11**, RESULTING FROM ANY MISREPRESENTATION, BREACH OF ANY REPRESENTATION OR WARRANTY OR NONFULFILLMENT OF ANY COVENANT OR AGREEMENT ON THE PART OF SELLER INCLUDED IN THIS AGREEMENT OR THE CERTIFICATES DELIVERED UNDER **SECTION 8.1(b)**.

12.3 Indemnification and Defense Procedures

A Person that desires to make a claim for indemnification under this **ARTICLE 12** is herein referred to as an "**Indemnified Person**" and the Party from which indemnification is sought under this **ARTICLE 12** is herein referred to as the "**Indemnifying Party.**" All claims for indemnification under this **ARTICLE 12** shall be asserted and resolved as follows:

(a) If a third party claim is asserted against an Indemnified Person for which the Indemnified Person desires to seek indemnification from an Indemnifying Party (an "Indemnified Claim"), then the Indemnified Person shall promptly transmit to the Indemnifying Party a written notice ("Claim Notice") (i) notifying such Indemnifying Party of such Indemnified Claim and requesting indemnity,

- (ii) setting forth the full name, address for all notices and the authorized representatives of such Indemnified Person with respect to such Indemnified Claim, and (iii) describing in reasonable detail the nature of the Indemnified Claim, including a copy of all papers served with respect to such Indemnified Claim (if any) and the basis of such request for indemnification hereunder. Failure to provide such Claim Notice promptly shall not affect the right of the Indemnified Person to indemnification hereunder except to the extent the Indemnifying Party is prejudiced thereby; provided that the Indemnifying Party shall not be obligated to defend, indemnify or otherwise hold harmless an Indemnified Person with respect to a third party claim until a Claim Notice meeting the foregoing requirements is furnished to the Indemnifying Party by the Indemnified Person. Within thirty (30) days after receipt of any Claim Notice (the "Election Period") from a Party, the Indemnifying Party shall notify the Indemnified Person (A) whether the Indemnifying Party disputes its obligation to indemnify the Indemnified Person with respect to such third party claim and (B) whether the Indemnifying Party desires to defend the Indemnified Person against such third party claim; provided that, if the Indemnifying Party fails to so notify the Indemnified Person during the Election Period, the Indemnifying Party shall be deemed to have elected to dispute such liability and not to defend against such third party claim.
- (b) If the Indemnifying Party notifies the Indemnified Person within the Election Period that the Indemnifying Party (i) does not dispute its liability to indemnify the Indemnified Person and (ii) elects to assume the defense of such Indemnified Person with respect to such third party claim, then the Indemnifying Party shall have the right to defend, at its sole cost and expense, such third party claim by all appropriate proceedings, which proceedings shall be prosecuted diligently by the Indemnifying Party to a final conclusion or settled at the discretion of the Indemnifying Party in accordance with this Section 12.3(b). The Indemnifying Party shall have full control of such defense and proceedings, including any compromise or settlement thereof; provided that the Indemnifying Party shall not enter into any settlement agreement (or settle or compromise any such third party claim in a manner) which provides for or results in any payment by or liability of the Indemnified Person of or for any damages or other amount, any Lien on any property of the Indemnified Person, any finding of responsibility or liability on the part of the Indemnified Person or any sanction or restriction upon the conduct of any business by the Indemnified Person without the Indemnified Person's express written consent, which consent shall not be unreasonably withheld. The Indemnified Person is hereby authorized, at the sole cost and expense of the Indemnifying Party (but only if the Indemnified Person is actually entitled to indemnification hereunder), to file, during the Election Period, any motion, answer or other pleadings which the Indemnified Person shall deem necessary or appropriate to protect its interests or those of the Indemnifying Party and not reasonably expected to be prejudicial to the Indemnifying Party. If requested by the Indemnifying Party, the Indemnified Person agrees, at the sole cost and expense of the Indemnifying Party, to cooperate with the Indemnifying Party and its counsel in contesting any such third party claim which the Indemnifying Party elects to contest, including the making of any related counterclaim or crosscomplaint against any Person (other than a Buyer Indemnified Party, if the Indemnified Person is a Buyer Indemnified Party, or a Seller Indemnified Party, if the Indemnified Person is a Seller Indemnified Party). The Indemnified Person may participate in, but not control, any defense or settlement of any third party claim controlled by the Indemnifying Party pursuant to this Section 12.3(b), and the Indemnified Person shall bear its own costs and expenses with respect to such participation. The prosecution of the defense of a third party claim with reasonable diligence shall include the taking of such action (including the posting of a bond, deposit or other security) as may be necessary to prevent any action to foreclose a Lien against or attachment of the property of the Indemnified Person for payment of such third party claim.

- If the Indemnifying Party (i) fails to notify the Indemnified Person within the Election Period that the Indemnifying Party elects to defend the Indemnified Person pursuant to Section 12.3(b) or (ii) elects to defend the Indemnified Person pursuant to Section 12.3(b) but fails to prosecute the defense of (or to settle) the third party claim with reasonable diligence, then the Indemnified Person shall have the right to defend, at the sole cost and expense of the Indemnifying Party (but only if the Indemnified Person is actually entitled to indemnification hereunder), the third party claim by all appropriate proceedings, which proceedings shall be promptly and vigorously prosecuted by the Indemnified Person to a final conclusion or settled. The Indemnified Person shall have full control of such defense and proceedings; provided, however, that the Indemnified Person may not enter into any compromise or settlement of such third party claim, without the Indemnifying Party's express written consent, which shall not be unreasonably withheld or delayed. The Indemnifying Party may participate in, but not control, any defense or settlement controlled by the Indemnified Person pursuant to this Section 12.3(c), and the Indemnifying Party shall bear its own costs and expenses with respect to such participation.
- (d) If an Indemnified Person desires to seek indemnification from an Indemnifying Party for a claim or other matter which does not involve a third party claim, then the Indemnified Person shall promptly transmit to the Indemnifying Party a written notice describing in reasonable detail the nature of such claim or other matter, the Indemnified Person's reasonable estimate of the amount of damages attributable to such claim or other matter and the basis for the Indemnified Person's entitlement to indemnification. Unless the Indemnifying Party notifies the Party who sent such notice within thirty (30) days from its receipt of such notice that the Indemnifying Party disputes such claim for indemnity, the Indemnifying Party shall be deemed to have accepted and agreed to such claims.

12.4 Seller's General Liability Limitation

- (i) Notwithstanding anything contained to the contrary in this Agreement:
 - (i) Seller shall not be required to indemnify any Person under **Section 12.2(b)(ii)** with respect to a breach of any representation or warranty made by Seller in this Agreement (other than the representations and warranties set forth in **Section 3.1(i)** and the Fundamental Representations) unless the amount which would otherwise be payable by Seller thereunder with respect to the breach or series of related breaches of any such representation and warranty exceeds a fifty thousand dollar (\$50,000) threshold per event or circumstance.
 - (ii) In no event shall Seller's aggregate liability arising out of or relating to **Section 12.2(b)(ii)** with respect to a breach by Seller of any representation or warranty made by Seller in this Agreement (other than the representations and warranties set forth in **Section 3.1(i)** and the Fundamental Representations) exceed an amount equal to [***].
 - (iii) Seller shall not be required to indemnify any Person under **Section 12.2(b)(ii)** with respect to a breach of any representation or warranty made by Seller in this Agreement (other than the representations and warranties set forth in **Section 3.1(i)** and the Fundamental Representations) unless the aggregate amount which would otherwise be payable by Seller thereunder with respect to all breaches of any such representations and warranties exceeds an amount equal to [***] (the "**Deductible Amount**"), and in such event, Seller shall be responsible for only the amount in excess of the Deductible Amount.
- (j) The amount to which a Buyer Indemnified Party may become entitled under <u>Section 12.2</u> shall be net of any actual recovery (whether by way of payment, discount, credit, off-set or otherwise)

received from a third party (including any insurer) less any cost associated with receiving such recovery in respect of a claim.

12.5 Exclusive Remedy

Seller and Buyer acknowledge and agree that from and after the Closing, the indemnification provisions of this **ARTICLE 12** and those obligations contained in the Parent Guaranty are the sole and exclusive remedy of Seller and Buyer for the breach of any representation or warranty or nonfulfillment of any covenant or agreement on the part of Seller or Buyer under this Agreement or confirmed in any certificate delivered pursuant hereto, and Seller does hereby release, acquit and forever discharge all Buyer Indemnified Parties and Buyer does hereby release, acquit, and forever discharge all Seller Indemnified Parties from any such other remedies; *provided, however*, that either Party shall be entitled to pursue all remedies available at Law or in equity (including specific performance and injunctive relief without the necessity of posting bond) for any breach by the other Party of the provisions of **Section 4.2(b)** or **Section 4.2(c)**, as the case may be.

12.6 Determination of Damages Related to Breach

Notwithstanding anything contained herein to the contrary, for purposes of determining the amount of liabilities associated with a breach of any representation or warranty by any Party hereunder (but not for purposes of determining whether a breach has occurred), the representations and warranties of such Party shall, for purposes of this **ARTICLE 12**, be read without giving effect to any materiality, Material Adverse Effect or qualification with a similar meaning in such representation or warranty.

12.7 Recovery

In the event Covered Liabilities suffered by an Indemnified Party are recoverable under more than one provision of this Agreement, such Indemnified Party will only be permitted to recover with respect to any particular Covered Liabilities suffered by it one time as it is the Parties' intent that once any particular Covered Liabilities have been recovered by a particular Indemnified Party under one provision, such Covered Liability no longer exists with respect to such Indemnified Party and, therefore, recovery by such particular Indemnified Party for such same Covered Liabilities under another provision would constitute unintended and prohibited "double" recovery. No Buyer Indemnified Party shall be entitled to any claim for Covered Liabilities under Section 12.2 arising from claims based on the determination of the Final Settlement Statement, as it is the intent that the procedures set forth in Article 2 be the exclusive remedy for such claims.

ARTICLE 13 TERMINATION; REMEDIES

13.1 Termination

- (d) <u>Termination of Agreement</u>. This Agreement and the transactions contemplated hereby may be terminated at any time prior to the Closing:
 - (iii) By the mutual written agreement of Seller and Buyer; or
 - (iv) If the Closing has not occurred by 5:00 P.M., Mountain Time, on October 31, 2014 (the " **Termination Effective Time**"), then (A) by Seller if any condition specified in **Section 8.1** has not (1) been satisfied on or before the Termination Effective Time or (2) been

waived by Seller, or (B) by Buyer if any condition specified in **Section 8.2** has not (1) been satisfied on or before the Termination Effective Time or (2) been waived by Buyer; *provided*, in each case, that the failure to consummate the transactions contemplated hereby on or before such time did not result from the failure by the Party seeking termination of this Agreement to fulfill any undertaking or commitment provided for herein on the part of such Party that is required to be fulfilled on or prior to the Closing.

- (v) By Seller, if on the Scheduled Closing Date (A) Seller is not then in material breach of any provision of this Agreement and (B) Buyer shall have breached its representations, warranties, covenants or agreements contained in this Agreement, which breach results in the failure of a condition set forth in **Section 8.1(a)**, **Section 8.1(b)** or **Section 8.1(c)**, and (1) such breach cannot by its nature be cured prior to the Termination Effective Time, or (2) Buyer fails to initiate action to cure such breach within 7 days following receipt of written notice from Seller; or
- (vi) By Buyer, if on the Scheduled Closing Date (A) Buyer is not then in material breach of any provision of this Agreement or the Deposit Escrow Agreement and (B) Seller shall have breached its representations, warranties covenants or agreements contained in this Agreement, which breach results in the failure of a condition set forth in Section 8.2(a), Section 8.2(b), Section 8.2(c) or Section 8.2(f), and (1) such breach cannot by its nature be cured prior to the Termination Effective Time, or (2) Seller fails to initiate action to cure such breach within 7 days following receipt of written notice from Buyer;
- (e) <u>Effect of Termination</u>. In the event of termination of this Agreement by Seller, on the one hand, or Buyer, on the other hand, written notice thereof shall forthwith be given by the terminating Party to such other Party, and this Agreement shall thereupon terminate; *provided, however*, that **Section 7.3**, **Section 13.2**, **ARTICLE 11** and **ARTICLE 14** shall survive such termination. If this Agreement is terminated as provided herein, all filings, applications and other submissions made to any Governmental Authority shall, to the extent practicable, be withdrawn from the Governmental Authority to which they were made.

13.2 Remedies

- (e) <u>General</u>. Except as set forth in **Section 14.5** or this **Section 13.2**, all fees and expenses incurred in connection with this Agreement and the transactions contemplated hereby shall be paid by the Party or Parties, as applicable, incurring such expenses whether or not the transactions contemplated by this Agreement are consummated.
- (f) <u>Seller's</u> <u>Remedies</u>.
 - (iii) Notwithstanding anything herein provided to the contrary, if this Agreement is validly terminated by Seller pursuant to (i) Section 13.1(a)(ii)(A) due to the failure of the conditions in Section 8.1(a), Section 8.1(b), or Section 8.1(c) to have been satisfied, or Section 13.1(a)(iii), Buyer and Seller shall promptly, and in any event within two (2) Business Days following such termination, deliver written instructions to the Escrow Agent directing the Escrow Agent to disburse the Cash Deposit, together with all interest earned thereon, to Seller.
 - (iv) In the event that Seller shall receive the Cash Deposit pursuant to **Section 13.2(b)(i)** then (A) the receipt of the Cash Deposit shall (i) be deemed to be liquidated damages for any

and all losses or damages suffered or incurred by Seller or any of its Affiliates or any other Person in connection with this Agreement (and the termination hereof), the transactions contemplated hereby (and the abandonment thereof) or any matter forming the basis for such termination and (ii) except as set forth in the proviso below, upon Seller's receipt of the Cash Deposit, be the sole and exclusive remedy of Seller and its Affiliates against the Buyer Indemnified Parties for any loss or damage suffered as a result of the failure of the transactions contemplated by this Agreement to be consummated or for a breach or failure to perform hereunder, under the Commitment Letter or otherwise and (B) except as set forth in the proviso below, none of the Buyer Indemnified Parties shall have any further liability or obligation relating to or arising out of this Agreement or the transactions contemplated hereby; *provided*, *however*, if Buyer Willfully Breaches this Agreement in a manner causing a failure of a closing condition in **Section 8.1**, Seller shall be entitled to retain the Cash Deposit and pursue compensation from Buyer for any and all losses or damages suffered or incurred by Seller or any of its Affiliates in connection with this Agreement (and the termination hereof), the transactions contemplated hereby (and the abandonment thereof) or any matter forming the basis for such termination, provided that the Seller's right to recovery in such circumstance shall be limited to [***], with such amount to be offset by the Cash Deposit. Notwithstanding the foregoing, Buyer shall continue to be bound by its obligations set forth in **ARTICLE 4** after any termination by any Party, and Seller shall be entitled to pursue all remedies available at Law or in equity (including specific performance and injunctive relief without the necessity of posting bond) for any breach by Buyer of such obligations.

- (g) <u>Buyer's Remedies</u>. Notwithstanding anything herein provided to the contrary:
 - (vii) If this Agreement is terminated by any Party pursuant to **Section 13.1(a)**, Buyer shall be entitled to receive the Cash Deposit free and clear of any claims by Seller thereto and Buyer and Seller shall promptly, and in any event within two (2) Business Days following such termination, deliver written instructions to the Escrow Agent directing the Escrow Agent to disburse the Cash Deposit to Buyer; *provided, however*, that if Seller is entitled to receive the Cash Deposit pursuant to **Section 13.2(b)**, then Buyer shall not be entitled to receive the Cash Deposit; and
 - (viii) If Buyer has the right to terminate this Agreement pursuant to (A) Section 13.1(a)(ii)(B) due to the failure of the conditions in Section 8.2(a), Section 8.2(b), Section 8.2(c) or Section 8.2(f) to have been satisfied, or (B) Section 13.1(a)(iv), then Buyer, at its sole option, may either (I) seek to enforce specific performance of this Agreement or (II) pursue compensation in the form of monetary damages provided that there has been a material breach by Seller under this Agreement; provided, however, that any damages payable to Buyer under this Section 13.2(c) will be limited to [***]; provided, further, if Seller breaches Section 7.11 or Willfully Breaches the Agreement in a manner causing a failure of a closing condition in Section 8.2, Buyer's damages for any and all losses or damages suffered or incurred by Buyer or any of its Affiliates in connection with this Agreement (and the termination hereof), the transactions contemplated hereby (and the abandonment thereof) or any matter forming the basis for such termination shall be limited to [***].

ARTICLE 14 MISCELLANEOUS

14.1 Counterparts

This Agreement may be executed in one or more counterparts, all of which shall be considered one and the same agreement, and shall become effective when one or more counterparts have been signed by each of the Parties and delivered to such other Party.

14.2 Notices

Unless otherwise expressly provided in this Agreement, all notices required or permitted hereunder shall be in writing and deemed sufficiently given for all purposes hereof if (a) delivered in person, by courier or by registered or certified United States Mail to the Person to be notified, with receipt obtained, or (b) sent by facsimile, with "answer back" or other "advice of receipt" obtained, in each case to the appropriate address or number as set forth below. Each notice shall be deemed effective on receipt by the addressee as aforesaid; provided that, notice received by facsimile after 5:00 P.M. at the location of the addressee of such notice shall be deemed received on the first Business Day following the date of such electronic receipt.

Notices to Seller shall be addressed as follows:

Baytex Energy USA LLC c/o Baytex Energy Corp. Centennial Place, East Tower 2800, 520 – 3 rd Avenue SW Calgary, Alberta T2P 0R3 Attention: General Counsel Facsimile: (587) 952-3029

With copy (which shall not constitute notice) to:

Baytex Energy USA LLC 600 17th Street, Suite 1600

Denver, CO

80202

Attention: Tim

Morris

Facsimile: (303) 825-

2790

or at such other address or to such other facsimile number and to the attention of such other Person as Seller may designate by written notice to Buyer.

Notices to Buyer shall be addressed to:

SM ENERGY COMPANY 1775 Sherman Street, Suite 1200 Denver, CO 80203 Attention: Herb Vogel

Facsimile: 303.864.2598

With copy (which shall not constitute notice) to:

SM ENERGY COMPANY

1775 Sherman Street, Suite 1200

Denver, CO 80203

Attention: David W. Copeland Facsimile: 303.864.2598

or at such other address or to such other facsimile number and to the attention of such other Person as Buyer may designate by written notice to Seller.

14.3 Governing Law; Venue; Jurisdiction.

- (g) THIS AGREEMENT AND THE TRANSACTIONS CONTEMPLATED HEREBY SHALL BE GOVERNED BY AND INTERPRETED IN ACCORDANCE WITH THE LAWS OF THE STATE OF COLORADO WITHOUT GIVING EFFECT TO PRINCIPLES THEREOF RELATING TO CONFLICTS OF LAW RULES THAT WOULD DIRECT THE APPLICATION OF THE LAWS OF ANOTHER JURISDICTION.
- (h) The Parties agree that the appropriate, exclusive and convenient forum for any disputes between the Parties arising out of this Agreement or the transactions contemplated hereby shall be in any state or federal court in Denver, Colorado, and each of the Parties irrevocably submits to the jurisdiction of such courts solely in respect of any legal proceeding arising out of or related to this Agreement. The Parties further agree that the Parties shall not bring suit with respect to any disputes arising out of this Agreement or the transactions contemplated hereby in any court or jurisdiction other than the above-specified courts; *provided, however*, that the foregoing shall not limit the rights of the Parties to obtain execution of judgment in any other jurisdiction. The Parties further agree, to the extent permitted by Law, that a final and unappealable judgment against a Party in any Action contemplated above shall be conclusive and may be enforced in any other jurisdiction within or outside the United States by suit on the judgment, a certified copy of which shall be conclusive evidence of the fact and amount of such judgment. Except to the extent that a different determination or finding is mandated due to the applicable law being that of a different jurisdiction, the Parties agree that all judicial determinations or findings by a state or federal court in Denver, Colorado, with respect to any matter under this Agreement shall be binding.
- (i) To the extent that a Party has or hereafter may acquire any immunity from jurisdiction of any court or from any legal process (whether through service or notice, attachment prior to judgment, attachment in aid of execution, execution or otherwise) with respect to itself or its property, each Party hereby irrevocably (i) waives such immunity in respect of its obligations with respect to this Agreement and (ii) submits to the personal jurisdiction of any court described in **Section 14.3(b)**.
- (j) EACH PARTY AGREES THAT IT HEREBY IRREVOCABLY WAIVES THE RIGHT TO TRIAL BY JURY IN ANY ACTION TO ENFORCE OR INTERPRET THE PROVISIONS OF THIS AGREEMENT.

14.4 Entire Agreement

This Agreement, the Appendix, Exhibits and Schedules hereto, the Confidentiality Agreement and the Deposit Escrow Agreement contain the entire agreement among the Parties with respect to the subject matter hereof and there are no agreements, understandings, representations or warranties among

the Parties other than those set forth or referred to herein. The headings herein are for convenience only and shall have no significance in the interpretation hereof.

14.5 Expenses

Buyer shall be responsible for all recording, filing or registration fees relating to the filing, recording or registration of any instruments transferring title to the Membership Interests to Buyer from Seller. All other costs and expenses incurred by each Party in connection with all things required to be done by it hereunder, including attorneys' fees, accountant fees and the expense of title examination, shall be borne by the Party incurring same. Any recordation fees incurred in connection with the transfer of the Subject Assets to the Company pursuant to Section 7.10 shall be borne 100% by Buyer.

14.6 Successors and Assigns

This Agreement shall be binding upon and inure to the benefit of the Parties and their respective successors and permitted assigns; provided, however, the respective rights and obligations of the Parties shall not be assignable or delegable by any Party without the express written consent of each non-assigning or non-delegating Party. Any purported assignment or delegation without such consent is void.

14.7 Amendments and Waivers

This Agreement may not be modified or amended except by an instrument or instruments in writing signed by the Party against whom enforcement of any such modification or amendment is sought. Any Party may, only by an instrument in writing, waive compliance by another Party with any term or provision of this Agreement on the part of such other Party. The waiver by any Party of a breach of any term or provision of this Agreement shall not be construed as a waiver of any subsequent breach.

14.8 Attorneys' Fees

The substantially prevailing Party or Parties in any legal proceeding brought under or to enforce this Agreement shall be additionally entitled to recover court costs and reasonable attorneys' fees from the non-substantially prevailing Party or Parties.

14.9 Severability

If any term or other provision of this Agreement is invalid, illegal or incapable of being enforced by any rule of Law or public policy, all other conditions and provisions of this Agreement shall nevertheless remain in full force and effect so long as the economic or legal substance of the transactions contemplated hereby is not affected in any adverse manner to any Party. Upon such determination that any term or other provision is invalid, illegal or incapable of being enforced, the Parties shall negotiate in good faith to modify this Agreement so as to effect the original intent of the Parties as closely as possible in an acceptable manner to the end that the transactions contemplated hereby are fulfilled to the extent possible.

14.10 Appendix, Schedules and Exhibits

The Appendix and all Schedules and Exhibits hereto which are referred to herein are hereby made a part of this Agreement and incorporated herein by such reference.

14.11 Interpretation

It is expressly agreed that this Agreement shall not be construed against any Party, and no consideration shall be given or presumption made, on the basis of who drafted this Agreement or any particular provision hereof or who supplied the form of Agreement. Each Party agrees that this Agreement has been purposefully drawn and correctly reflects its understanding of the transactions that this Agreement contemplates. In construing this Agreement:

- (a) examples shall not be construed to limit, expressly or by implication, the matter they illustrate;
- (b) the word "includes" and its derivatives means "includes, but is not limited to" and corresponding derivative expressions;
- (c) a defined term has its defined meaning throughout this Agreement and each Appendix, Exhibit and Schedule to this Agreement, regardless of whether it appears before or after the place where it is defined;
- (d) each Exhibit and Schedule to this Agreement is a part of this Agreement, but if there is any conflict or inconsistency between the main body of this Agreement (including **Appendix A** which shall be considered part of the main body of this Agreement) and any Exhibit or Schedule, the provisions of the main body of this Agreement shall prevail;
- (e) the term "cost" includes expense and the term "expense" includes cost; and
- (f) all references to "dollars" or "\$" shall be deemed references to United States dollars.

14.12 Agreement for the Parties' Benefit Only

Except to the extent provided in **ARTICLE 12**, this Agreement is for the sole benefit of Buyer and Seller and their respective successors and assigns as permitted herein and no other Person shall be entitled to enforce this Agreement, rely on any representation, warranty, covenant or agreement contained herein, receive any rights hereunder or be a third party beneficiary of this Agreement. Any Indemnified Person which is a third Person shall be indemnified and held harmless under the terms of this Agreement only to the extent that a Party expressly elects to exercise such right of indemnity and hold harmless on behalf of such third Person Indemnified Person pursuant to **ARTICLE 12**; and no Party shall have any direct liability or obligation to any third Person or be liable to any third Person for any election or non-election or any act or failure to act under or in regard to any term of this Agreement. Any claim for indemnity or hold harmless hereunder on behalf of an Indemnified Person must be made and administered by Buyer or Seller, as the case may be.

14.13 No Recordation

Without limiting any Party's right to file suit to enforce its rights under this Agreement, the Parties expressly covenant and agree not to record or place of record this Agreement or any copy or memorandum hereof; *provided*, *however*, that any Party shall have the right to file all or any part of this Agreement (including any document executed by any of the Parties in connection herewith) with the SEC in connection with any filing that a Party in good faith believes is required by Law or any national securities exchange.

14.14 Time of Essence

Time is of the essence in this Agreement. If the date specified in this Agreement for giving notice or taking action is not a Business Day (or if the period during which any notice is required to be given or any action taken expires on a date which is not a Business Day), then the date for giving such notice or taking such action (and the expiration date of such period during which notice is required is given or action taken) shall be the next day which is a Business Day.

[SIGNATURE PAGE FOLLOWS]

IN WITNESS WHEREOF,	this Agreement has been sig	gned by or on behalf o	f each of the Parties of	on the day first above	written to be
effective as of the Effective Time					

	SELLER
	BAYTEX ENERGY USA LLC
	By: Name: Timothy R. Morris Title: Vice President
	BUYER
	SM ENERGY COMPANY
	By: Name: Javan D. Ottoson Title: President and Chief Operating Officer
Signatur Purchase	re Page to Agreement

APPENDIX A

Attached to and made part of that certain Purchase Agreement, dated July 29, 2014, by and among Baytex Energy USA LLC, as Seller, SM Energy Company, as Buyer

DEFINITIONS

- "Action" shall mean any action, suit or other proceeding by or before any Governmental Authority or any arbitration proceeding before any arbitral tribunal.
- "Adjusted Purchase Price" shall be as defined in Section 2.2.
- "Adjustment Period" shall be as defined in Section 2.5(c).
- "Affiliate" shall mean, with respect to any Person, any other Person that directly or indirectly (through one or more intermediaries or otherwise) Controls, is Controlled by, or is under common Control with the first Person. No Person shall be deemed an Affiliate of any Person by reason of the exercise or existence of rights, interests or remedies under this Agreement.
- "Agreement" shall be as defined in the Preamble.
- "Allocated Value" shall be as defined in Section 5.2(b).
- "Allocation Schedule" shall be as defined in Section 7.9(f).
- "Allocation Amount" shall be as defined in Section 7.9(f).
- "Asset and Liability Transfer" shall be as defined in Section 7.10(a).
- "Asset Taxes" shall mean all Property Taxes and Severance Taxes.
- "Assumed Liabilities" shall mean (i) all Covered Liabilities arising out of or attributable to (a) the ownership, use, construction, maintenance or operation of the Subject Assets after the Effective Time, (b) the obligation to plug and abandon any well included in the Subject Assets (whether arising prior to or after the Effective Time) and (c) the obligation to properly pay and disburse the funds comprising the Suspense Funds to the proper Person, irrespective of whether such Suspense Funds are attributable to oil, gas or other hydrocarbons produced from the Subject Interests before or after the Effective Time, and (ii) all Environmental Liabilities, known or unknown, arising out of or attributable to the ownership, use, construction, maintenance or operation of the Subject Assets at any time (whether arising prior to or after the Effective Time).
- "Assumption Agreement" shall be as defined in Section 7.11(c).
- "Bakken Drillsite Spacing Unit" shall be as defined in the definition of "Subject Assets" set forth in this Appendix A.
- "Bakken Pool" means that interval from fifty (50) feet above the upper Bakken shale being the top of the Bakken Pool, being the stratigraphic equivalent of 8,569 feet (as measured from the Kelly Bushing), to one

hundred (100) feet below the top of the Three Forks formation, measured at a depth of 8,823 feet (as measured from the Kelly Bushing), being the base of the Bakken Pool, as measured depth in the Dual Laterolog MSFL, Gearhart Logging, Run 1, logged October 3, 1985 in the No. 1 Knudsvig 14-1 well, located 660 feet south and 660 feet west, Section 1, Township 161 North, Range 100 West, Divide County, North Dakota, API #33023003560000.

"Business" means the ownership and operation of the Subject Assets.

"Business Day" shall mean any day that is not a Saturday, Sunday or legal holiday recognized by the State of Colorado, the Province of Alberta or the United States of America.

"Buyer Indemnified Parties" shall mean Buyer, Buyer's Affiliates and each of Buyer's and its Affiliate's respective past, present and future partners, directors, officers, employees, consultants, agents and other representatives.

"Buyer" shall be as defined in the Preamble.

"Cash Deposit" shall be as defined in Section 2.3.

"Casualty Loss" shall be as defined in Section 7.7.

"Claim Notice" shall be as defined in Section 12.3(a).

"Closing" shall mean the consummation of the transactions contemplated by ARTICLE 9.

"Closing Amount" shall be as defined in Section 2.4.

"Closing Date" shall be as defined in Section 9.1.

"Closing Time" shall be as defined in Section 2.5(c).

"Code" shall mean the Internal Revenue Code of 1986, as amended.

"Company" shall be as defined in the Recitals.

"Confidentiality Agreement" shall mean that certain Confidentiality Agreement dated May 30, 2014, by and between Seller and Buyer.

"Contracts" shall be as defined in the definition of "Subject Assets" set forth in this Appendix A.

"Control" shall mean the power to direct or cause the direction of the management and policies of a Person, whether through the ability to exercise voting power, by contract or otherwise; and "Controlled by" and "Controlling" have the meaning correlative thereto.

"Covered Liabilities" shall mean any and all debts, losses, liabilities, duties, claims (including those arising out of any demand, assessment, settlement, judgment or compromise relating to any actual or threatened Action), Taxes, fees, penalties, costs and expenses (including any attorneys' fees and any and all expenses whatsoever incurred in investigating, preparing or defending any Action), matured or unmatured, absolute or contingent, accrued or unaccrued, liquidated or unliquidated, known or unknown, including any of the

foregoing arising under, out of or in connection with any Action, any order or consent decree of any Governmental Authority, any award of any arbitrator, or any Law, contract, commitment or undertaking.

"Deductible Amount" shall be as defined in Section 12.4(a)(iii).

"Defect Notice Date" shall be as defined in Section 5.2(a).

"Defensible Title" shall mean such title of the Seller to a Property Subdivision that:

- (a) with respect to each Well and Bakken Drillsite Spacing Unit described on <u>Part II(A) of Exhibit A</u>, entitles the Company to receive not less than the applicable Net Revenue Interest specified for such Property Subdivision in <u>Part II(A) of Exhibit A</u>;
- (b) with respect to each Well and Bakken Drillsite Spacing Unit described on <u>Part II(A) of Exhibit A</u>, obligates the Company to bear the Working Interest of such Property Subdivision in an amount not greater than the Working Interest specified for such Property Subdivision in <u>Part II(A) of Exhibit A</u>; and
- (c) with respect to each Property Subdivision, is free and clear of all Liens, except for Permitted Encumbrances.

"Deferred Adjustment Claim" shall be as defined in Section 5.5(a).

"Deferred Environmental Adjustment Claim" shall be as defined in Section 6.8(a).

"Deferred Matters Date" shall be as defined in Section 5.5(a).

"Deposit Escrow Agreement" shall be as defined in Section 2.3.

"Effective Time" shall mean 7:00 A.M. at the location of the Subject Assets on July 1, 2014.

"Election Period" shall be as defined in Section 12.3(a).

"Environmental Arbitrator" shall be as defined in Section 6.8(b).

"Environmental Defect" shall be as defined in Section 6.1.

"Environmental Defect Adjustment" shall be as defined in Section 6.5(a).

"Environmental Defect Amount" shall be as defined in Section 6.1.

"Environmental Defect Deductible" shall be as defined in Section 6.6.

"Environmental Defect Notice" shall be as defined in Section 6.2.

"Environmental Defect Properties" shall be as defined in Section 6.1.

"Environmental Laws" shall mean all Laws pertaining to pollution, remediation of contamination, protection of the environment and natural resources, and restoration of environmental quality, including the following federal statutes and the regulations promulgated thereunder: (a) the Comprehensive Environmental Response, Compensation and Liability Act of 1980; (b) the Emergency Planning and Community Right-to-

Know Act; (c) the Resource Conservation and Recovery Act; (d) the Clean Air Act; (e) the Clean Water Act; (f) the Safe Drinking Water Act; and (g) the Toxic Substances Control Act; and any state or local laws and regulations similar thereto, as each of the foregoing has been amended and is in effect on the Defect Notice Date.

"Environmental Liabilities" means all liabilities, obligations, responsibilities, remedial actions, losses, damages, punitive damages, consequential damages, treble damages, costs and expenses (including all reasonable fees, disbursements and expenses of counsel, experts and consultants and costs of investigation and feasibility studies), fines, penalties, sanctions, supplemental environmental projects in lieu of any fines, penalties or sanctions, and interest incurred as a result of any claim or demand by any Person or in response to any violation of or liability under Environmental Law, whether known or unknown, accrued or contingent, whether based in contract, tort, implied or express warranty, strict liability, criminal or civil statute, to the extent based upon, related to, or arising under or pursuant to any Environmental Law, permit, order or agreement with any Governmental Authority or other Person, which relates to any environmental, health or safety condition, violation of Environmental Law or a release or threatened release of, or exposure to, Hazardous Materials and/or NORM, including all obligations and liabilities, known or unknown, with respect to the Subject Assets, relating to on- or off-site ground water, surface water, sediment, or soil in accordance with applicable agreements and Laws, including any obligations to pay for or perform the assessment, remediation, removal or disposal of NORM, asbestos, mercury, polychlorinated biphenyls, drilling fluids and chemicals, produced waters and hydrocarbons, hazardous substances, or other liabilities arising under or relating to Environmental Laws or environmental obligations in applicable agreements with respect to the Subject Assets.

"Environmental Permits" shall be as defined in Section 3.1(1)(i)(B).

"Equipment" shall be as defined in the definition of "Subject Assets" set forth in this Appendix A.

"Escrow Account" shall be as defined in Section 2.3.

"Escrow Agent" shall be as defined in Section 2.3.

"Excluded Environmental Adjustment" shall be as defined in Section 6.7.

"Final Settlement Date" shall be as defined in Section 2.6.

"Final Settlement Statement" shall be as defined in Section 2.6.

"Flat Reps" shall be as defined in Section 8.2(a).

"Fundamental Representations" shall be as defined in Section 10.1.

"General Conveyance" shall be as defined in Section 7.11(c).

"Governmental Authority" shall mean the United States of America and Canada, and any state, province, county, municipality or other governmental subdivision thereof, and any court or any governmental department, commission, board, bureau, agency or other instrumentality thereof.

"Hazardous Materials" shall mean petroleum substances (including crude oil and any components, fractions, by-products or derivatives thereof), exploration and production wastes, any substance defined or regulated as a hazardous substance, toxic substance, hazardous waste, hazardous material, radioactive

material, radioactive waste, pollutant, or contaminant by any Environmental Law, including asbestos, polychlorinated biphenyls, and naturally occurring radioactive materials, and any other chemical, pollutant, contaminant, substance or waste that is regulated or for which liability or standards of care are imposed under any Environmental Law.

"Hedge Agreement" means any and all transactions, agreements or documents now existing or hereafter entered into, which provide for an interest rate, credit or commodity swap, cap, floor, collar or any combination of, or option with respect to, these or similar transactions.

"Hydrocarbon Interests" shall mean (a) mineral servitudes and leases affecting, relating to or covering any oil, gas and other hydrocarbons in place and the leasehold interests and estates in the nature of working or operating interests under such leases, as well as overriding royalties, net profits interests, production payments, carried interests, rights of recoupment and other interests in, under or relating to such leases, (b) fee interests in oil, gas or other hydrocarbons in place, (c) royalty interests in oil, gas or other hydrocarbons in place, (d) any other interest in oil, gas or other hydrocarbons in place, (e) any economic or contractual rights, options or interests in and to any of the foregoing, including any sublease, farmout or farmin agreement or production payment affecting any interest or estate in oil, gas or other hydrocarbons in place, and (f) any and all rights and interests attributable or allocable thereto by virtue of any pooling, unitization, communitization, production sharing or similar agreement, order or declaration.

"Imbalance" shall mean any imbalance at the wellhead between the amount of oil, gas or other hydrocarbons produced from a well and allocable to the interests of the Company therein and the production from the relevant well to which the Company is entitled, together with any appurtenant rights and obligations concerning future in kind and/or cash balancing at the wellhead.

"Income Taxes" shall mean all Taxes based upon, measured by, or calculated with respect to (a) gross or net income or gross or net receipts or profits (including franchise Tax and any capital gains, alternative minimum Taxes, net worth and any Taxes on items of Tax preference, but not including sales, use, goods and services, real or personal property transfer or other similar Taxes), (b) multiple bases (including corporate franchise, doing business or occupation Taxes) if one or more of the bases upon which such Tax may be based upon, measured by, or calculated with respect to, is described in clause (a) above, or (c) withholding Taxes measured with reference to or as a substitute for any Tax described in clauses (a) or (b) above, and (d) and any penalties, additions to Tax, and interest levied or assessed with respect to a Tax described in (a), (b), or (c).

"Indemnified Claim" shall be as defined in Section 12.3(a).

"Indemnified Person" shall be as defined in Section 12.3.

"Indemnifying Party" shall be as defined in Section 12.3.

"Independent Accounting Firm" shall be as defined in Section 2.6.

"Intended Tax Treatment" shall be as defined in Section 7.9(a).

"Knowledge" shall mean the actual knowledge (excluding any imputed or implied knowledge of any kind) of any fact, circumstance or condition by Dan Anderson, Tim Morris, Randy Verret, and Wayne King.

"Lands" shall be as defined in the definition of "Subject Assets" set forth in this Appendix A.

"Law" shall mean any applicable statute, law (including the common law), ordinance, regulation, rule, ruling, order, restriction, requirement, writ, injunction, decree or other official act of or by any Governmental Authority.

"Leases" shall be as defined in the definition of "Subject Assets" set forth in this Appendix A.

"Lien" shall mean any mortgage, pledge, lien, security interest, or other encumbrance.

"Material Adverse Effect" shall mean any result, consequence, condition or matter which (a) materially adversely affects the Subject Assets, taken as a whole, or the ownership, operations, results of operations or the value of the Subject Assets, taken as a whole (after taking into account, in each case, any insurance, condemnation award, indemnity and other recoveries payable in respect thereof) or (b) materially impairs, prevents or delays Seller's ability to perform its obligations under this Agreement or to consummate the transactions contemplated by this Agreement; provided that the following shall not be deemed to constitute a Material Adverse Effect: any effect resulting from (i) any change in economic, industry, financial, market, political, general or regional conditions, (ii) any change in Law, regulatory policy or accounting rules, (iii) entering into this Agreement or the announcement of the transactions contemplated by this Agreement, (iv) any matter, contract or agreement set forth on an exhibit or schedule hereto as of the date hereof or (v) any change arising in connection with any natural disasters, hostilities, acts of war, sabotage or terrorism.

"Material Contract" shall mean, to the extent binding upon the Subject Assets or the Company from and after Closing, any contract or agreement which is one or more of the following types:

- (a) Any contract or agreement with any Affiliate of Seller or the Company;
- (b) Any contract or agreement for the sale, purchase, exchange, processing, marketing, transportation, or other disposition of oil, gas or other hydrocarbons which is not cancelable without penalty on sixty (60) days or less prior written notice;
- (c) Any contract or agreement that creates any area of mutual interest or similar provision with respect to the Subject Interests or any other material Subject Asset or contains any restrictions on the ability of the Company or its assigns to compete with any other Person;
- (d) Any contract or agreement that is binding upon Seller or the Company to sell, lease, farmout or otherwise dispose of any interest in any of the Subject Interests or any other material Subject Asset after the date hereof, other than conventional rights of reassignment arising in connection with the Company's surrender or release of any of the Subject Assets;
- (e) Any contract or agreement for the lease or rental to the Company of compressors which is not cancellable without penalty on sixty (60) days or less prior written notice;
- (f) Any drilling contract, joint operating agreement, exploration agreement, fracturing agreement, development agreement, participation agreement or similar agreement to which the Company is a party;
- (g) Any contract or agreement that is a seismic or other geophysical acquisition agreement or license to which the Company is a party;

- (h) Any contract or agreement that contains any rights allowing a third party to participate in any sales or purchases of any of the Subject Interests or any other material Subject Asset that are triggered by or applicable to the transactions contemplated by this Agreement;
- (i) Any contract or agreement that is an indenture, mortgage, loan, credit or sale-leaseback, guaranty of any obligation, bonds, letters of credit or similar financial instrument; and
- (j) Except for (i) Subject Interests, (ii) operating, unitization, pooling and communitization agreements and (iii) contracts and agreements of the nature described in clauses (a) through (i) above, any contract or agreement that (A) could reasonably be expected to result in aggregate revenues to the Company of more than Two Hundred and Fifty Thousand and No/100 Dollars (\$250,000.00) during the current or any subsequent calendar year or (B) obligate the Company to make payments of more than Two Hundred and Fifty Thousand and No/100 Dollars (\$250,000.00) during the current or any subsequent calendar year.

"Membership Interest" shall mean the rights of Seller as a Member under the Limited Liability Company Agreement of the Company to be entered into by the Seller prior to the Closing.

"Net Revenue Interest" shall mean an interest (expressed as a percentage or decimal fraction) in and to all oil and gas produced and saved from or attributable to a Property Subdivision, after giving effect to all royalties, overriding royalties, production payments, net profits interests and other burdens.

"NORM" shall mean naturally occurring radioactive material.

"Organizational Documents" means, the Certificate of Formation and the Operating Agreement of the Company attached hereto as Exhibit G.

"Parent Guaranty" means the guaranty of Seller's indemnity obligations hereunder, in the form attached hereto as Exhibit F, to be entered into by Baytex Energy Corp. and Buyer at the Closing.

"Party" and "Parties" shall be as defined in the Preamble.

"Permits" shall be as defined in Section 3.1(e).

"Permitted Encumbrances" shall mean any of the following matters:

- (a) All agreements, instruments, documents, Liens, and other matters which are described in **Schedule A-1**;
- (b) Any (i) undetermined or inchoate Liens constituting or securing the payment of expenses which were incurred incidental to maintenance, development, production or operation of the Subject Assets or for the purpose of developing, producing or processing oil, gas or other hydrocarbons therefrom or therein and (ii) materialman's, mechanics', repairman's, employees', contractors', operators' or other similar Liens for liquidated amounts arising in the ordinary course of business incidental to construction, maintenance, development, production or operation of the Subject Assets or the production or processing of oil, gas or other hydrocarbons therefrom, that are not delinquent and that will be paid in the ordinary course of business or, if delinquent, that are being contested in good faith;
- (c) Any Liens for Taxes not yet delinquent;

- (d) Any Liens or security interests created by Law or reserved in oil, gas and/or mineral leases for royalty, bonus or rental or for compliance with the terms of the Subject Interests;
- (e) Any easements, rights-of-way, servitudes, Permits, licenses, surface leases and other rights with respect to operations to the extent such matters do not interfere in any material respect with the operation of the portion of the Subject Assets burdened thereby;
- (f) All Liens, contracts, agreements, instruments, obligations, defects, irregularities and other matters affecting any Subject Asset which individually or in the aggregate are not such as to interfere materially with the operation, value or use of such Subject Asset;
- (g) Any encumbrance, title defect or other matter (whether or not constituting a Title Defect) cured by Seller or waived or deemed waived by Buyer pursuant to **ARTICLE 5**;
- (h) All Laws and all rights reserved to or vested in any Governmental Authority (i) to control or regulate any Subject Asset in any manner; (ii) by the terms of any right, power, franchise, grant, license or permit, or by any provision of Law, to terminate such right, power, franchise, grant, license or permit or to purchase, condemn, expropriate or recapture or to designate a purchaser of any of the Subject Assets; (iii) to use any Subject Asset in a manner which does not materially impair the use or value of such property for the purposes for which it is currently owned and operated; or (iv) to enforce any obligations or duties affecting the Subject Assets to any Governmental Authority with respect to any franchise, grant, license or permit;
- (i) The terms and conditions of the Subject Interests, Surface Contracts, Permits, and Contracts;
- (j) Other than such rights that have already been triggered, rights of reassignment requiring notice and/or the reassignment (or granting an opportunity to receive a reassignment) of a leasehold interest to the holders of such reassignment rights prior to surrendering or releasing such leasehold interest; and
- (k) All consents and approvals of or filings with applicable Governmental Authorities in connection with assignments of the Subject Interests or other material Subject Assets.

"Person" shall mean any Governmental Authority or any individual, firm, partnership, corporation, limited liability company, association, joint venture, trust, unincorporated organization or other entity or organization.

"Preference Right" shall mean any right or agreement that enables or may enable any Person to purchase or acquire any Subject Asset or any interest therein or portion thereof as a result of or in connection with (a) the sale, assignment, encumbrance or other transfer of any Subject Asset or any interest therein or portion thereof or (b) the execution or delivery of this Agreement or the consummation or performance of the terms and conditions contemplated by this Agreement.

"Preliminary Settlement Statement" shall be as defined in Section 2.5(a).

"Property Expenses" shall be as defined in Section 2.5(c).

"Property Subdivision" shall mean each Bakken Drillsite Spacing Unit or Well described or referenced in Part II(A) of Exhibit A.

"Property Taxes" shall mean all ad valorem, real property, personal property, and all other Taxes and similar obligations, and any penalties, additions to Tax, and interest levied or assessed thereon, assessed against the Subject Assets or based upon or measured by the ownership of the Subject Assets, but not including Income Taxes, Severance Taxes, or Transfer Taxes.

"Purchase Price" shall be as defined in Section 2.2.

"Records" shall be as defined in the definition of "Subject Assets" set forth in this Appendix A.

"Release" shall mean any depositing, spilling, leaking, pumping, pouring, placing, emitting, discarding, abandoning, emptying, discharging, migrating, injecting, escaping, leaching, dumping or disposing into the environment.

"Remediation" or "Remediate" shall mean investigation, assessment, characterization, delineation, monitoring, sampling, analysis, removal action, remedial action, response action, corrective action, mitigation, treatment or cleanup of Hazardous Materials or other similar actions as required by any applicable Environmental Laws with respect to soil, land surface, groundwater, sediment, surface water, the ambient environment or subsurface strata or otherwise for the protection of human health and the environment or to restore a property or operation to compliance with Environmental Laws.

"Retained Liabilities" means all Seller's Covered Liabilities arising out of or attributable to the ownership, use, construction, maintenance or operation of any assets other than the Subject Assets.

"SEC" shall mean the United States Securities and Exchange Commission or any successor to such agency.

"Securities Act" shall mean the Securities Act of 1933, as amended.

"Seller" shall be as defined in the Preamble.

"Seller Affiliated Group" means the affiliated group, as defined in Section 1504(a) of the Code, of which Seller is a member.

"Seller Indemnified Parties" shall mean Seller, Seller's Affiliates and each of Seller's and its Affiliate's respective past, present and future shareholders, directors, officers, employees, consultants, agents and other representatives.

"Seller Marks" shall be as defined in Section 7.8.

"Seller Title Credit" shall be as defined in Section 5.4(a).

"Seller Title Credit Amount" shall be as defined in Section 5.4(a).

"Seller Title Credit Notice" shall be as defined in Section 5.2(a).

"Settled Asset Taxes" shall be as defined in Section 7.9(c).

"Settled Property Taxes" shall be as defined in Section 7.9(b)(i).

"Settled Severance Taxes" shall be as defined in Section 7.9(c).

"Severance Amount" means, with respect to the Business Employees, the amounts set forth on a schedule provided to Buyer by Seller on the Execution Date

"Severance Taxes" shall mean all extraction, production, gross production, excise, net proceeds, severance, windfall profit, and all other similar Taxes, and any penalties, additions to such Taxes, and interest levied or assessed thereon, with respect to the Subject Assets that are based upon or measured by the production of hydrocarbons or the receipt of proceeds therefrom, but not including Property Taxes, Income Taxes, and Transfer Taxes.

"Subject Assets" shall mean all of the Company's right, title, and interest in and to the following assets and properties:

- (a) all Hydrocarbon Interests, other properties and interests described in <u>Part I of Exhibit A</u> (the "Leases") and the lands covered by, pooled or unitized with or subject to the Leases (the "Lands");
- (b) all oil, gas, water or injection wells located on the Leases and Lands, including those described in <u>Part IV of Exhibit A</u> (the "Wells");
- the Bakken Pool for those areas described on <u>Part II(A) of Exhibit A</u> as the "Bakken Drillsite Spacing Units". The Company's working interest as to the Bakken Pool for each Bakken Drillsite Spacing Unit are set forth under the column labeled Working Interest and the Company's Net Revenue interest as to the Bakken Pool are set forth under the column labeled Net Revenue Interest on <u>Part II(A) of Exhibit A</u>. The Working Interest and Net Revenue Interest for a Bakken Drillsite Spacing Unit are as of the Effective Time and shall not apply to Wells identified on <u>Part IV of Exhibit A</u>. The Bakken Drillsite Spacing Unit may or may not have been approved by the North Dakota Industrial Commission; however, the size and shape of the unit is materially consistent with such spacing as has been established and approved in the area;
- (d) all other Hydrocarbon Interests of the Company in, to or under, derived or subject to the Leases, Lands or Wells, even though such interests of the Company may be incorrectly described or referred to in, or a description thereof may be omitted from **Part I of Exhibit A** (such interests, together with the Leases, the Lands, the Wells, and the Bakken Drillsite Spacing Units, the "**Subject Interests**");
- (e) all surface leases, rights of way, easements and other rights with respect to the use and occupancy of the surface of and the subsurface depths under the Lands and other lands used in connection with the exploration, development or operation of the Subject Interests or the transportation, marketing or disposition of oil, gas or other hydrocarbons, water or other substance produced therefrom, including those described in Part III of Exhibit A (the "Surface Contracts");
- (f) all personal property used or held for use in connection with the exploration, development or operation of the Subject Interests or the transportation, marketing or disposition of hydrocarbons, water or other substance produced therefrom, including all equipment, machinery, vehicles (whether leased or owned) described in **Part V of Exhibit A**, materials, wellhead equipment, pumping units, flowlines, tanks, buildings, injection facilities, water disposal facilities, processing and separating facilities, compression facilities, gathering systems, casing, rods, tanks, boilers, tubing, pumps, motors, machinery, monitoring equipment, pipe, tubular goods, equipment inventory, fixtures and other tangible personal property and improvements and specifically including those described in **Part V of Exhibit A** (the "**Equipment**");

- (g) all other real and personal property located upon the Lands or lands covered by the Surface Contracts and used in connection with the exploration, development or operation of the Subject Interests;
- (h) all sales contracts, purchase contracts, operating agreements, exploration agreements, development agreements, balancing agreements, farmout agreements, drilling contracts, other service agreements, transportation, marketing, processing, treatment or gathering agreements, equipment leases and other contracts, agreements and instruments, in each case, to the extent related to the exploration, development or operation of the Subject Interests or the transportation, marketing or disposition of oil, gas or other hydrocarbons, water or other substance produced therefrom, including, without limitation, the Material Contracts (the "Contracts");
- (i) all Permits and Environmental Permits with respect to the exploration, development or operation of the Subject Interests or the transportation, marketing or disposition of oil, gas or other hydrocarbons, water or other substance produced therefrom;
- (j) except to the extent transfer thereof may not be made without violating confidentiality obligations or waiving any attorney/client privilege, originals or, if an original is not in Seller's possession, all digital or hard copies of any and all of Seller's books, records and files, including all Surface Contract and Contract files, lease files, land files, wells files, well logs and other well data, maps, division order files, abstracts, title files, title opinions, production files, drilling reports, files related to any Asset Tax, technical, engineering and maintenance files, operations, environmental, safety, production records, and other similar information, in each case, (i) relating to the Subject Interests, Equipment, Surface Contracts or Contracts, or the ownership, use, maintenance or operation thereof, or the other properties or interests described in this definition of Subject Assets, and (ii) in Seller's possession or control (the "Records");
- (k) all (i) oil, gas and other hydrocarbons produced from and to the extent attributable to the Subject Interests with respect to all periods subsequent to the Effective Time and (ii) proceeds from or of such oil, gas and other hydrocarbons;
- (l) the Suspense Funds; and
- (m) all seismic, geological, geophysical, engineering, and other data and interpretations, files and records (in whatever form) related to the Subject Interests.

"Subject Interests" shall be as defined in the definition of "Subject Assets" set forth in this Appendix A.

"Surface Contracts" shall be as defined in the definition of "Subject Assets" set forth in this Appendix A.

"Suspense Funds" shall mean the amounts set forth on <u>Schedule 3.1(n)</u> together with all other funds held in suspense by or on behalf of the Company from and after the date set forth in <u>Schedule 3.1(n)</u> with respect to the Subject Assets.

"Tax" or "Taxes" shall mean (i) any federal, state or local income, gross receipts, ad valorem, sales and use, goods and services, payroll, unemployment, employment, social security (or similar), disability, occupation, property (whether real or personal), severance, value added, transfer, registration, license, stamp, franchise, profits, capital, excise, withholding, premium, federal or provincial capital taxes, occupation, alternative, add-on minimum or other taxes, levies or other like assessments, customs, duties, imposts, charges, unclaimed property and escheat obligations, surcharges or fees imposed by or on behalf of any Governmental Authority, including any interest, penalty or addition attributable thereto, and (ii) any liability in respect of any items

described in (i) that arises by reason of a contract, assumption, transferee or successor liability, operation of Law (including by reason of participation in a consolidated, combined or unitary Tax Return) or otherwise.

"Tax Controversy" shall be as defined in Section 7.9(g)(iv).

"Tax Return" shall mean any return, report, declaration, or similar statement or form required to be filed with a Taxing Authority with respect to any Tax (including any attached schedules and related or supporting information), including any information return, claim for refund, amended return or declaration of estimated Tax, and including any amendment thereof.

"Taxing Authority" means, with respect to any Tax, the Governmental Authority that imposes such Tax, and the agency (if any) charged with the collection of such Tax.

"Termination Effective Time" shall be as defined in Section 13.1(a)(ii).

"Title Arbitrator" shall be as defined in Section 5.5(b).

"Title Defect" shall be as defined in Section 5.3.

"Title Defect Adjustment" shall be as defined in Section 5.2(d).

"Title Defect Amount" shall be as defined in Section 5.2(d).

"Title Defect Deductible" shall be as defined in Section 5.7.

"Title Defect Property" shall be as defined in Section 5.2(c)(i).

"Transfer Requirement" shall mean any consent, approval, authorization or permit of, or filing with or notification to, any Person which is required to be obtained, made or complied with for or in connection with any sale, assignment or transfer of any Subject Asset or any interest therein other than those customarily obtained from or made or complied with any Governmental Authority following the closing in transactions of this nature.

"Transfer Tax" shall mean any and all transfer Taxes (excluding Income Taxes, Severance Taxes and Property Taxes), including sales, use, excise, goods and services, stock, conveyance, gross receipts, registration, business and occupation, securities transactions, real estate, land transfer, stamp, documentary, notarial, filing, recording, Permit, license, authorization and similar Taxes, fees, duties, levies, customs, tariffs, imposts, assessments, obligations and charges.

"Treasury Regulations" shall mean the regulations promulgated by the United States Department of the Treasury pursuant to and in respect of provisions of the Code. All references herein to sections of the Treasury Regulations shall include any corresponding provision or provisions of succeeding, similar, substitute, proposed or final Treasury Regulations.

"Wells" shall be as defined in the definition of "Subject Assets" set forth in this Appendix A.

"Willful Breach" means a material breach that is a consequence of an intentional act or omission undertaken by the breaching Party with the actual knowledge that the taking of, or failure to take, such act would, or would be reasonably expected to, cause a material breach of such covenant or agreement.

"Working Interest" means with respect to any Property Subdivision, the interest in and to such Property Subdivision that is burdened with the obligation to bear and pay costs and expenses of maintenance, development and operations on or in connection with such Property Subdivision, but without regard to the effect of any royalties, overriding royalties, production payments, net profits interests and other similar burdens upon, measured by, or payable out of production therefrom.

CERTIFICATION

I, Anthony J. Best, certify that:

- I have reviewed this quarterly report on Form 10-Q of SM Energy Company;
- 2. Based on my knowledge, this report does not contain any untrue statement of a material fact or omit to state a material fact necessary to make the statements made, in light of the circumstances under which such statements were made, not misleading with respect to the period covered by this report;
- 3. Based on my knowledge, the financial statements, and other financial information included in this report, fairly present in all material respects the financial condition, results of operations and cash flows of the registrant as of, and for, the periods presented in this report;
- 4. The registrant's other certifying officer and I are responsible for establishing and maintaining disclosure controls and procedures (as defined in Exchange Act Rules 13a-15(e) and 15d-15(e)) and internal control over financial reporting (as defined in Exchange Act Rules 13a-15(f) and 15d-15(f)) for the registrant and have:
 - (a) Designed such disclosure controls and procedures, or caused such disclosure controls and procedures to be designed under our supervision, to ensure that material information relating to the registrant, including its consolidated subsidiaries, is made known to us by others within those entities, particularly during the period in which this report is being prepared;
 - (b) Designed such internal control over financial reporting, or caused such internal control over financial reporting to be designed under our supervision, to provide reasonable assurance regarding the reliability of financial reporting and the preparation of financial statements for external purposes in accordance with generally accepted accounting principles;
 - (c) Evaluated the effectiveness of the registrant's disclosure controls and procedures and presented in this report our conclusions about the effectiveness of the disclosure controls and procedures, as of the end of the period covered by this report based on such evaluation; and
 - (d) Disclosed in this report any change in the registrant's internal control over financial reporting that occurred during the registrant's most recent fiscal quarter (the registrant's fourth fiscal quarter in the case of an annual report) that has materially affected, or is reasonably likely to materially affect, the registrant's internal control over financial reporting; and
- 5. The registrant's other certifying officer and I have disclosed, based on our most recent evaluation of internal control over financial reporting, to the registrant's auditors and the audit committee of the registrant's board of directors (or persons performing the equivalent functions):
 - (a) All significant deficiencies and material weaknesses in the design or operation of internal control over financial reporting which are reasonably likely to adversely affect the registrant's ability to record, process, summarize and report financial information; and
 - (b) Any fraud, whether or not material, that involves management or other employees who have a significant role in the registrant's internal control over financial reporting.

Date: October 29, 2014

/s/ ANTHONY J. BEST Anthony J. Best Chief Executive Officer

CERTIFICATION

I, A. Wade Pursell, certify that:

- I have reviewed this quarterly report on Form 10-Q of SM Energy Company;
- 2. Based on my knowledge, this report does not contain any untrue statement of a material fact or omit to state a material fact necessary to make the statements made, in light of the circumstances under which such statements were made, not misleading with respect to the period covered by this report;
- 3. Based on my knowledge, the financial statements, and other financial information included in this report, fairly present in all material respects the financial condition, results of operations and cash flows of the registrant as of, and for, the periods presented in this report;
- 4. The registrant's other certifying officer and I are responsible for establishing and maintaining disclosure controls and procedures (as defined in Exchange Act Rules 13a-15(e) and 15d-15(e)) and internal control over financial reporting (as defined in Exchange Act Rules 13a-15(f) and 15d-15(f)) for the registrant and have:
 - (a) Designed such disclosure controls and procedures, or caused such disclosure controls and procedures to be designed under our supervision, to ensure that material information relating to the registrant, including its consolidated subsidiaries, is made known to us by others within those entities, particularly during the period in which this report is being prepared;
 - (b) Designed such internal control over financial reporting, or caused such internal control over financial reporting to be designed under our supervision, to provide reasonable assurance regarding the reliability of financial reporting and the preparation of financial statements for external purposes in accordance with generally accepted accounting principles;
 - (c) Evaluated the effectiveness of the registrant's disclosure controls and procedures and presented in this report our conclusions about the effectiveness of the disclosure controls and procedures, as of the end of the period covered by this report based on such evaluation; and
 - (d) Disclosed in this report any change in the registrant's internal control over financial reporting that occurred during the registrant's most recent fiscal quarter (the registrant's fourth fiscal quarter in the case of an annual report) that has materially affected, or is reasonably likely to materially affect, the registrant's internal control over financial reporting; and
- 5. The registrant's other certifying officer and I have disclosed, based on our most recent evaluation of internal control over financial reporting, to the registrant's auditors and the audit committee of the registrant's board of directors (or persons performing the equivalent functions):
 - (a) All significant deficiencies and material weaknesses in the design or operation of internal control over financial reporting which are reasonably likely to adversely affect the registrant's ability to record, process, summarize and report financial information; and
 - (b) Any fraud, whether or not material, that involves management or other employees who have a significant role in the registrant's internal control over financial reporting.

Date: October 29, 2014

/s/ A. WADE PURSELL

A. Wade Pursell

Executive Vice President and Chief Financial Officer

CERTIFICATION

PURSUANT TO

18 U.S.C. SECTION 1350,

AS ADOPTED PURSUANT TO

SECTION 906 OF THE SARBANES-OXLEY ACT OF 2002

In connection with the Quarterly Report on Form 10-Q of SM Energy Company (the "Company") for the quarterly period ended September 30, 2014, as filed with the Securities and Exchange Commission on the date hereof (the "Report"), Anthony J. Best, as Chief Executive Officer of the Company, and A. Wade Pursell, as Executive Vice President and Chief Financial Officer of the Company, each hereby certifies, pursuant to and solely for the purpose of 18 U.S.C. § 1350, as adopted pursuant to § 906 of the Sarbanes-Oxley Act of 2002, to the best of his knowledge and belief, that:

- (1) The Report fully complies with the requirements of section 13(a) or 15(d) of the Securities Exchange Act of 1934 (15 U.S.C. 78m or 78o(d)); and
- (2) The information contained in the Report fairly presents, in all material respects, the financial condition and results of operations of the Company.

/s/ ANTHONY J. BEST

Anthony J. Best Chief Executive Officer October 29, 2014

/s/ A. WADE PURSELL

A. Wade Pursell Executive Vice President and Chief Financial Officer October 29, 2014